

Provided by the author(s) and University of Galway in accordance with publisher policies. Please cite the published version when available.

Title	Assessing and training temporal relational responding in children
Author(s)	Neufeld, Jacob
Publication Date	2023-12-19
Publisher	NUI Galway
Item record	http://hdl.handle.net/10379/18005

Downloaded 2024-04-29T00:29:27Z

Some rights reserved. For more information, please see the item record link above.





Ollscoil na Gaillimие University of Galway

## Assessing and Training Temporal Relational Responding in Children

Jacob Neufeld

Dissertation submitted in partial fulfilment of the requirements for the Degree of Doctor of Philosophy in Applied Behaviour Analysis Supervisor: Dr. Ian Stewart School of Psychology University of Galway July 2023 This research was supported by the University of Galway Doctoral Research Scholarship Scheme.

## **Table of Contents**

Declaration Regarding the Work	iv
Abstract	v
Dedication	viii
Acknowledgments	ix
Publications and Conference Presentations	xi
List of Figures	xii
List of Tables	xiii
List of Abbreviations	xiv
Chapter 1. Introduction and Background	1
Philosophical Assumptions	2
Relational Frame Theory	6
Chapter 2. Time-Related Skills in Humans and Relational Frame Theory: A Review and Conceptus Analysis	al 13
Past Behavioral Research on Responses to Time in Humans	14
Temporal Relations: An Overview and Present RFT Research	17
Research on Temporal Skill Sets and RFT Application	26
Chapter 3. Assessing Temporal Relational Responding in Children	53
Method	59
Results	65
Discussion	71
Chapter 4. Developing Procedures for Training Temporal Relational Responding in Children	79
Method	81
Results	84
Discussion	88
Chapter 5. Training Temporal Relational Framing in Young Children	90
Method	93
Results	99
Discussion	103
Chapter 6. Teaching Nonarbitrary Temporal Relational Responding in Adolescents with Autism	108
Method	112
Results	116
Discussion	119
Chapter 7. General Discussion	123
Overview of Thesis	124

Overall Contributions and Implications	
Limitation and Future Research Directions	
Conclusion	
References	137
	166
Appendices	
Appendices	
Appendices	

## **Declaration Regarding the Work**

I, the **Candidate**, certify that the Thesis is all my own work and that I have not obtained a degree in this University or elsewhere on the basis of any of this work.

This thesis is the result of my own investigations, except where otherwise stated.

This thesis is the result of my own investigations, except where otherwise stated. The following chapter includes work conducted in collaboration with other students:

- Chapter 6 (Study 4) includes work conducted in collaboration with Deirdre Barry, ABA MSc student.
  - o Co-designed assessment and training protocol, assisted with analysis and write-up

Signed:	2-1-2-	Date:	21/07/2023

#### Abstract

Time is a fundamentally important dimension of human experience and responding adaptively in terms of this dimension is critical to human personal and societal functioning. However, there is an important distinction to be made between responding to time as a physical or nonarbitrary dimension of existence and time as an abstract concept. It is the latter that is critical to the type of self-knowledge and societal organization that is unique to human life. Relational frame theory (RFT) sees relational framing, and temporal relational framing in particular, as key to this uniquely human level of responding. Temporal relational framing is a form of derived relational framing that involves responding under the control of contextual cues such as the words "before" and "after" to stimuli in terms of their directional displacement along a temporal dimension (e.g., if taught that A is after B; then derive that B is before A).

Important skills such as sequencing events, planning, time understanding, or talking about the past or future typically involve temporal relations. Accordingly, relational responding in the presence of temporal relational cues such as "before" and "after" is an important building block to more complex repertoires from an RFT perspective. Existing psychological research on temporal relations is mostly composed of descriptive studies and limited to investigations with age-level performance. Moreover, although previous RFT studies have taught several other varieties of relational responding, little work has yet been done in the case of temporal relational frames. The current thesis aimed to extend research in this area by conceptualizing how time as an abstract concept can be approached from an RFT perspective, and by developing RFT-based assessment and training procedures for temporal relational framing in children.

In support of these overall objectives, a narrative review and conceptual analysis (Chapter 2) was first done, which had three primary aims. The first aim was to summarize the existing psychological research into time-related skill sets in humans given the vast areas of investigation within this domain. The second aim was to explain temporal relational framing and the present state of relevant research into this skill. The third aim was to conceptually explore how the RFT approach to human temporal responding might amplify and extend the existing research base, in particular with respect to the acquisition and training of key aspects of temporal responding.

Study 1 (Chapter 3) was an empirical starting point based on one area of need identified in Chapter 2. The aim of Study 1 was to assess performance on a test of temporal relational responding in young children at increasing levels of complexity. Twenty-five typically developing children between 3-8 years were assessed on tasks of nonarbitrary (i.e., based on physical events) and arbitrary (i.e., based on contextual cues only) temporal relations. Nonarbitrary testing involved responding to observed sequences of stimuli and answering "before" and "after" questions about the order. Arbitrary testing was conducted using an adapted version of a relational evaluation procedure (REP) format which involved vocally responding to "before" and "after" questions about a presented temporal relational network across many exemplars (Kirsten & Stewart, 2022). Additionally, a transformation

v

of stimulus function was assessed on trials that involved performing motor actions in a particular sequence based on the presented relational network. Results showed a correlation between overall performance across temporal relational responding tasks and age. Performance on nonarbitrary "before" and "after" trials improved similarly with age whereas with arbitrary relations, participants performed much more poorly on "after" trials than on "before" trials and some interesting cohort specific patterns were also seen.

The data from Study 1 provided insight into how children of various ages may respond on temporal relational responding tasks; however, it remained unclear whether a behavioral training approach could be used improve performance. Study 2 (Chapter 4) aimed to develop a training protocol for teaching derived temporal relations in young children. Previous research has shown that multiple exemplar training (MET) can be used to train relational framing in children, though no work has been done specifically with temporal frames. Informed by the results of Study 1, this second study trained two typically developing 6-year-olds in derived temporal relational responding using a MET procedure in a nonconcurrent multiple baseline design. Study 2 used a similar REP format as in Study 1 in which various relational networks were shown and participants vocally answered "before" and "after" questions about the presented relations. Training involved reinforcement and error correction procedures for correct and incorrect responding, respectively. Following training, both participants reached criterion levels and passed generalization tests for both mutual and combinatorial relations with untrained stimuli and maintained performance four-weeks after training was finished.

Study 3 (Chapter 5) aimed to replicate and extend Study 2. A similar procedure was used as in the previous study but now with the inclusion of a protocol for training and testing of sequenced responses to allow for assessment of transformation of stimulus function. Three typically developing 5-year-olds were taught using MET in a combined multiple probe design across participants and responses. Training involved reinforcement procedures for correct responses and error correction procedures for incorrect responses. Following training, all participants reached criterion levels and passed generalization tests for mutual and combinatorial entailment, and transformation of function with untrained stimuli and maintained high overall performance four weeks following training.

In both Studies 2 and 3, participants were specifically taught to relationally respond to arbitrary temporal relations. Despite the importance of arbitrary relations, some individuals may present with more significant deficits in which beginning intervention at a nonarbitrary (i.e., physically based) level is needed prior to progressing to more abstract relations. Study 4 (Chapter 6) aimed to assess existing temporal relational responding skills in three adolescents with ASD and to evaluate the efficacy of an RFT-based multiple exemplar training (MET) procedure to teach nonarbitrary temporal relations (responding to sequential changes in the environment), which constitute an important foundation for derived temporal relations. Participants were taught using multiple exemplars of 2-image stimulus sequences and answered "before" and "after" questions regarding the observed order. Reinforcement and error correction procedures were applied for correct

vi

and incorrect responding, respectively. Results indicated that all participants achieved mastery for nonarbitrary temporal relations following MET and that the skill generalized across novel stimuli.

The present thesis contributed to the literature base by providing an initial data set in terms of age-level responding on temporal relational responding tasks, creating a training protocol that can be used to teach temporal frames, and by extending research on temporal frames to new demographics. Additionally, several other time-related skills were identified as potential areas for further investigation from an RFT approach. In Chapter 7, the final chapter of this thesis, applied and theoretical implications of the data obtained, limitations of the studies, and potential future directions for research in this area are discussed.

Dedication

To Theo.

You are such an amazing joy.

I look forward to supporting you with your own dreams and adventures.

## Acknowledgments

My sincere gratitude to:

Dr. Ian Stewart. You have sincerely taught me so much. Your expertise and incredible supervision have made a world of difference for me throughout this PhD experience. I am so grateful for your guidance, feedback, and encouragement. I valued all our conversations and your insights that you so graciously shared. Thank you for making this work possible.

My Graduate Research Committee: Dr. Anne O'Connor, Dr. Helena Lydon, Dr. Mark Elliott. You were such wonderful supports throughout the past 4 years. I am grateful for each of your unique suggestions and perspectives that helped guide this thesis.

Dr. Aoife McTiernan and Dr. Ciara Gunning. Thank you for the variety of opportunities that you have given me throughout the PhD. I have learned so much and am so thankful for your leadership.

All the participants and their families. You all are so gracious. Thank you for your interest, willingness to participate (especially during a pandemic!), and for making this research possible.

Kristin. Your friendship, support, and experience were instrumental for me throughout this journey. Thank you for your assistance with recruitment and for bringing a slice of home away from home.

Rachel. I am so grateful for getting to do this PhD alongside you. You inspire me and it was such an honor to go through this journey together. A special thanks for saving me with some data collection!

Dearbhaile. You really helped me transition into this program. I am so appreciative of your guidance and support each step of the way and for even letting me join in on your own work.

Shauna, Aoife, Esther. You brought so much life to this program! I wish we all had more time together. I am so grateful for your friendship and the community that you brought.

Dr. Elle Kirsten. Thank you for all your time and insight with our conversations and for your foundational work that helped make this research possible. Your enthusiasm in this area is inspiring and I'm excited to see what's next.

Deirdre. It was such a pleasure to work on a project together! Thank you for your kindness, collaboration, and assistance.

Eoin and Maria. Thank you for your help with collecting IOA and fidelity data. Your flexibility with the ever-moving schedule was a lifesaver.

Thiago, Elisa, Aine. Thank you for your friendship and for helping me maintain my sanity!

Mike. It still shocks me how your friendship knows no borders. Thank you for continuing to jump on a plane and for always being a call away.

Mom and Dad. Thank you for your support, for always believing in me, and for helping me get to a place where I could embark on this journey.

Maddi. Words cannot express how important you were to me throughout this process. Thank you for your constancy—your encouragement when things were low, your celebrations when things were high, and for always being there for me no matter what. This thesis would be impossible without you, and I am forever grateful to you. I love you!

### **Publications and Conference Presentations**

### **Publications**

- Neufeld, J. & Stewart, I. (2023). A behavioral approach to the human understanding of time: Relational frame theory and temporal relational framing. *The Psychological Record*, 73(2), 301-332. <u>http://doi.org/10.1007/s40732-022-00529-7</u>
- Neufeld, J., Stewart, I., & McElwee, J. (2023). Assessing temporal relational responding in children. *The Psychological Record*, (73)2, 163-182. <u>https://doi.org/10.1007/s40732-023-00534-4</u>
- Neufeld, J., Stewart, I., & McElwee, J. (2023). Training temporal relational framing in young children. *Journal of Contextual Behavioral Science*, 28, 81-89. <u>https://doi.org/10.1016/j.jcbs.2023.03.013</u>

### **Manuscripts Accepted for Publication**

Barry, D., Neufeld, J., & Stewart, I. (2023). Teaching nonarbitrary temporal relational responding in adolescents with autism. *The Analysis of Verbal Behavior*.

## **Conference Presentations**

- Neufeld, J. & Stewart, I. (2023, April 21-22). Training Temporal Relational Framing in Children [Conference paper] Psychological Society of Ireland Division of Behaviour Analysis 2022, Athlone, Ireland
- Neufeld, J. & Stewart, I. (2022, September 1-3). Assessing & Training Temporal Relational Responding in Young Children [Conference poster]. 11th International Conference of the Association for Behavior Analysis International, Dublin, Ireland
- Neufeld, J. & Stewart, I. (2022, September 1-3). A Relational Frame Theory Approach to the Human Understanding of Time: Past, Present, and Future Research [Conference paper]. 11th International Conference of the Association for Behavior Analysis International, Dublin, Ireland
- Neufeld, J. & Stewart, I. (2022, June 15-18). Training Temporal Relational Responding in Young Children [Conference symposium]. 10th Conference of the European Association for Behaviour Analysis, Tampere, Finland
- Neufeld, J. & Stewart, I. (2022, April). Developing Procedures for Testing and Training Temporal Relational Responding in Children [Conference symposium] Psychological Society of Ireland Division of Behaviour Analysis 2022, Galway, Ireland
- Neufeld, J. & Stewart, I. (2020, April 3-4). Research Plan for Assessing and Training Temporal Relational Responding in Children with and without Development Delay [Conference presentation] Psychological Society of Ireland Division of Behaviour Analysis 2020, Galway, Ireland

# List of Figures

Figure 2.1 Example Relational Network Involving Quantification	20
Figure 2.2 Example Relations Involved in a Rule	23
Figure 2.3 Hypothetical Temporal Bisection and Temporal Generalization Graphs	35
Figure 2.4 Example Visual Display of Comparing Relational Networks	45
Figure 3.1 Example Stimulus Presentations	62
Figure 3.2 Example Stimulus Presentation for Transformation of Function Trials	65
Figure 3.3 Total Scores on the Temporal Relations Assessment (TRA) vs. Age	66
Figure 3.4 Average Percentage Correct per Age Cohort on Nonarbitrary Relational	70
Responding (NARR) Trials	
Figure 3.5 Average Percentage Correct per Age Group on Mutual Entailment and	71
Combinatorial Entailment Trials	
Figure 3.6 Average Percentage Correct per Age Group on Transformation of Function Trials	71
Figure 3.7 Comparison of Relational Statements Using "Before" and "After" Contextual Cues	74
Figure 4.1 Percent Correct across Both Skills in Each Participant	85
Figure 4.2 Percent Correct on "Before" Trials and "After" Trials in Each Participant across	87
All Phases	
Figure 5.1 Example Relational Statements	95
Figure 5.2 Example Stimuli Arrangement for Transformation of Function Trials	95
Figure 5.3 Example Flow of Presented Stimuli During Color-Action Pre-Training	97
Figure 5.4 Combined Multiple Probe Design Across Participants and Responses	101
Figure 5.5 Participant Responses During Baseline and Testing Sessions	102
Figure 6.1 Percentage Correct on Nonarbitrary Temporal Relations Trials across All Phases	118
for All Participants	

## List of Tables

Table 1.1 Summary of World Views	3
Table 1.2 Levels and Dimensions in the HDML Framework	10
Table 2.1 Example Arrangements of Sentences Describing a Temporal Sequence	29
Table 2.2 Example Relational Patterns with Time Units	41
Table 2.3 Example Planning and Time Management Relational Responses	52
Table 3.1 Nonarbitrary Trial Types	62
Table 3.2 Mutual and Combinatorial Entailment Trial Types	64
Table 3.3 Age Cohort Mean Scores on the Pre-assessment	66
Table 3.4 Spearman's Rho Correlations for Age and Temporal Relations Assessment (TRA)	66
Table 3.5 Participants' Scores on the Temporal Relations Assessment (TRA)	68
Table 4.1 Multiple Exemplar Training (MET) Stimulus Sets	82
Table 4.2 Mutual and Combinatorial Entailment Trial Types	82
Table 5.1 Trial Types for Each Relational Response	96

## List of Abbreviations

А	After		
AARR	Arbitrarily applicable relational responding		
ABA	Applied behavior analysis		
В	Before		
CE	Combinatorial entailment		
CFTQ	Child's Future Thinking Questionnaire		
$C_{\text{func}}$	Functional contextual cue		
CR	Conditioned response		
C <sub>rel</sub>	Relational contextual cue		
CS	Conditioned stimulus		
DRL	Differential reinforcement of low rate		
DRR	Derived relational responding		
EBI	Equivalence-based instruction		
FI	Fixed interval		
HDML	Hyperdimensional multilevel		
IOA	Interobserver agreement		
ME	Mutual entailment		
MET	Multiple exemplar training		
MTS	Match-to-sample		
NA-2	Nonarbitrary-2 stimuli		
NA-3	Nonarbitrary-3 stimuli		
NARR	Nonarbitrary relational responding		
PPVT-4	Peabody Picture Vocabulary Test—Fourth Edition		
REP	Relational evaluation procedure		
ROE-M	Relating, orienting, evoking, and motivating		
RFT	Relational frame theory		
SET	Scalar expectancy theory		
TONI-3	Test of Nonverbal Intelligence—Third Edition		
ToF	Transformation of function		
ToF-ME	Transformation of function via mutual entailment		
ToF-CE	Transformation of function via combinatorial entailment		
TRA	Temporal relations assessment		
UCR	Unconditioned response		
UCS	Unconditioned stimulus		
WAIS-III	Weschler Adult Intelligence Scale—Third Edition		
WCST	Wisconsin Card Sorting Test		

Chapter 1. Introduction and Background

Portions of this chapter have been accepted for publication:

Neufeld, J., & Stewart, I. (2022). A behavioral approach to the human understanding of time: Relational frame theory and temporal relational framing. *The Psychological Record*. https://doi.org/10.1007/s40732-022-00529-7 Responding to how events are temporally related to each other is a key skill set that underlies many complex behavioral repertoires such as telling time, sequencing events, planning, setting goals, problem solving, time-management, and perspective taking. In socioverbal communities, this kind of behavior is typically evoked by words such as, "before/after," "now/then," "earlier/later," etc., which signal a temporal relation between events (e.g., March is before April, lunch is sooner than recess). Relational frame theory (RFT; Hayes, Barnes-Holmes, et al., 2001) sees this type of responding as verbal and more specifically as a category of relational framing referred to as temporal relational framing (Hayes, Fox, et al., 2001). The purpose of this thesis is to investigate this concept by reviewing existing relevant research, carrying out a set of studies relative to the needs identified in the literature, and exploring potential future directions in this area.

Over the course of this research program, this project sought to contribute to existing knowledge by a) identifying specific temporal skill sets in humans of which RFT might offer analysis; b) extending the current research on temporal relations to new populations (neurotypical children and autistic adolescents); c) creating an assessment procedure that can be used to determine skill level with temporal relations; and d) establishing effective training methods for teaching temporal relational responding when this skill is still emerging.

The chapters of this thesis describe the research journey aligned to these objectives. In the present chapter, an overview of the philosophical and theoretical frameworks underpinning this thesis is first provided. Chapter 2 includes a narrative review of the literature on temporal skills from various psychological approaches. In conjunction with this review, a conceptual analysis is provided by laying out how RFT might be applied in terms of research and practice of temporal skill sets. Chapters 3-6 then focus on one area of application identified in Chapter 2: assessing and training temporal responding with "before" and "after." Chapter 3 describes the implementation of an assessment procedure for multiple levels of temporal relational responding across different ages. Chapter 4 discusses developing and piloting a training protocol to examine the efficacy of RFT-based procedures for training skills that had not yet been acquired based on findings in Chapter 3. Chapter 5 describes how the training protocol was revised and expanded for replication with slightly younger participants using a stronger experimental design. Chapter 6 details the application and adaptation of the assessment and training procedure for temporal relational responding for autistic participants. Lastly, Chapter 7 summarizes the key findings of this research program, the theoretical and applied implications, limitations, and directions for future work.

### **Philosophical Assumptions**

An overview of the philosophical assumptions of this approach is first made as the philosophy and scientific goal presented here may differ from other psychological approaches that investigate temporal skill sets. While the phenomenon of interest overlaps, there are distinctions between how the relevant events are studied and what the central goal of the analysis is. Making known these philosophical differences is important when communicating between disciplines as our different truth

criteria give rise to different analyses and interpretations as well as provide a useful context for the evaluation of a scientific work (Hayes et al., 1999).

Pepper (1942) presented a selection of world views that constitute systems by which analyses of the world, or a particular phenomenon of interest are understood and interpreted. Although Pepper's work was originally used to discuss philosophical systems, his framework is also relevant to behavioral science as pre-analytic assumptions may differ across scientific perspectives concerning the nature of reality, what should be analysed, or even how science should be carried out. These pre-analytic assumptions are influential to theory, methodology, and scientific interpretation (Levin et al., 2016). Some researchers have found applying Pepper's work to the science of behavior to be useful as the underlying philosophical assumptions of different psychological perspectives share similarities to the world views he described (Hayes et al., 1988)<sup>1</sup>. World views were not considered to be better or worse than each another, but rather were seen to encompass different views and objectives.

#### Table 1.1

#### Summary of World Views

World view	Root metaphor	Truth criterion	Characteristics	Question asked (Dilworth- Anderson et al., 2005)
Mechanism	Machine	Correspondence	Analytic / Integrative	"How does it work?
Formism	Similarity	Correspondence	Analytic / Dispersive	"What is it like?"
Organicism	Developing organism	Coherence	Synthetic / Integrative	"How does it develop?"
Contextualism	Act-in-context	Successful working	Synthetic / Dispersive	"How is it happening?"

The four world views Pepper described are: mechanism, formism, organicism, and contextualism (Table 1.1). Each world view stems from a root metaphor for describing the nature of phenomena (ontological assumptions) and has its own truth criterion for determining the validity of analyses of phenomena (epistemological assumptions). Formism and mechanism are analytic (reductionistic) while organicism and contextualism are synthetic (holistic). Mechanism and organicism are integrative (facts are placed into a determinate structure) while formism and contextualism are dispersive (facts are interpreted as they come). Each of these world views is explained below. For illustration, the example of "planning"—a behavioral repertoire relevant to the

<sup>&</sup>lt;sup>1</sup> Some authors have offered critiques and discussed potential limitations with applying Pepper's framework to behavioral science (see for example, Capaldi & Proctor, 1994; Dixon & Belisle, 2021). This framework is included in the current thesis as it provides a reasonably informative framework for discussing comparisons between goals and priorities within different psychological approaches when investigating similar phenomena.

concept of time—is used to show how each world view may approach and analyze this phenomenon slightly differently.

*Mechanism* uses the machine as a root metaphor and a truth criterion of correspondence. A machine, such as a computer, is made of separate parts that are related together and then when acted upon by a force, the machine functions in a particular way. The chief goal of the scientist is to explain how the machine works. A constructed model of a phenomenon is tested and considered true to the extent it corresponds with how the world actually is. When examining behavior, this world view may be recognized in various cognitive approaches (e.g., a cognitive model of planning might be formed and tested). For example, a machine (the organism) is seen to contain discrete, interrelated parts (e.g., attention, working memory, executive function, domain knowledge) while a force/causal agent (e.g., a problem scenario is presented) propels some type of effect or output to occur (e.g., a plan is formed). The model of the phenomenon is considered true to the extent it accurately predicts behavior.

*Formism* uses similarity as its root metaphor and correspondence as a truth criterion. Similarity here refers to the recurrence of recognizable forms (Hayes et al., 1988); that is, a certain phenomenon of interest is examined and categorized in terms of its similarity or dissimilarity to other phenomena. The goal of the scientist is to describe the essence of what the phenomenon is like. With behavior, the study of personality traits may be an example (e.g., examining which people tend to plan more). An explanation as to why a person tends to plan for events and activities may be that they are considered to have more of a conscientiousness trait (form). If a person tends to be a "planner," the scientist may predict that they would score similarly on measures of conscientiousness as others who tend to plan (recurrence of the recognizable form). The category is true to the extent that it corresponds with the observed phenomena.

*Organicism* uses the root metaphor of the developing organism (i.e., a living system) and a truth criterion of coherence. The process of organic development involves the unfolding of a phenomenon over time, such as a seed growing into a plant. The scientist seeks to explain and predict the orderly, sequential progression of the event or system. With behavior, developmental stage models in psychology may align with this world view (e.g., Piaget's stages). A proposed model of stages is analyzed based on how well it predicts an orderly reality. For example, a model might seek to explain how the ability of planning unfolds in development and at what ages certain planning skills are observed. If contradictions exist (e.g., a particular planning skill does not reveal a developmental trend), then the "organic whole" has not yet been realized. As facts about the presumed orderly process are obtained, an analysis is considered true to the extent that all the facts cohere with one another, leading to a full picture of the organic whole.

*Contextualism* uses the root metaphor of the ongoing act-in-context and has a truth criterion of successful working. Events are considered inseparable from their context (hence the hyphenation with "act-in-context") and therefore a scientist seeks to understand an event in relation to its context. For instance, the unit of the discriminated operant within behavior analysis is an example as each

component (antecedent-behavior-consequence) is defined and analyzed by its relation to the other components. Acts-in-context can be integrated or divided in an infinite number of ways, so truth is understood only in a relative sense (Gifford & Hayes, 1999). The behavior of making a plan can be analyzed in the context of going to the store, going to the store as part of hosting a party, and so on.

In contextualism, the purpose of the act is of primary importance. Any act has a functional purpose in its situational and historical context. A "correct" behavior in one context may not be as workable in another. Planning for the future (e.g., an upcoming vacation) may be more desirable in one context (e.g., to get the cheapest airfare) and less desirable in another context (e.g., when being spontaneous is a priority). In this example, a contextual perspective analyses planning by attending to the historical and situational contextual features in which it occurs. Since all behavior is situated in context, this also means that the scientist doing the analyzing is itself an act-in-context. Therefore, a theory or analysis of behavior is true or not insofar as it "works" to achieve the pre-established goal. Accordingly, an analysis of behavior is useful to the extent it aids the scientist in achieving their explicit goal.

One variety of contextualism, *functional contextualism*, has an explicit goal of prediction and influence of behavior. The extent to which this goal has been achieved can be evaluated across the dimensions of precision, scope, and depth. Precision refers to the degree in which a phenomenon can be explained in a limited number of ways. Scope describes the degree to which an analysis can be applied to a range of phenomena. Depth refers to the consistency between an analysis made and findings from other areas or levels of analysis (Biglan & Hayes, 2016). More broadly, functional contextualism is a variant of the larger philosophical tradition of pragmatism.

Functional contextualism is advantageous for the study of human behavior due to its unique emphasis on influence (Biglan & Hayes, 1996). The other world views also seek to predict behavior through theoretical accounts, but it is not necessarily derivative that a given model can be used to influence behavior; especially if a model appeals to explanations only involving variables that cannot be manipulated (e.g., an internal timing mechanism, psychological traits, age). The additional focus on influence insists on the contextualist to include variables that can be directly changed (Hayes & Brownstein, 1986). The prediction of behavior can indeed provide informative findings but nevertheless, a functional contextual approach argues that the analysis is incomplete if stopped there. Biglan and Hayes (1996) argued that in order to change a behavior, one must target variables that can in fact be altered. Ultimately, to achieve a scientific goal of both prediction and influence of a behavior, an analysis must include adjusting the context in which that behavior occurs. Changing environmental variables that are sensitive to the historical and situational contexts of individual learners specifically provide practical approaches to lead to socially significant behavior change—a central aim of applied behavior analysis (ABA; Baer et al., 1968). While behavior analysis has been interpreted both mechanistically as well as contextualistically, researchers within the field of relational frame theory (RFT), which provides the theoretical background to the present thesis, see

behavior analysis and hence also RFT as forms of functional contextualism. From this point of view the explicit goal of behavior analysis is prediction-and-influence of behavior in general while the explicit goal of RFT is prediction-and-influence of human language and cognition.

## **Relational Frame Theory**

Relational frame theory (RFT) is a contextual behavior analytic account of language and cognition (Hayes, Barnes-Holmes, et al., 2001) and is the theoretical foundation of this thesis. The following section provides a brief overview of RFT in terms of its background, key concepts, and empirical research.

### **Background to RFT**

Historically, RFT emerged from limitations in Skinner's conceptual approach to verbal behavior and empirical findings on rule-governed behavior and stimulus equivalence. Skinner defined verbal behavior as behavior reinforced through the mediation of a listener trained by the verbal community to do so (Skinner, 1957a). Challenges to this definition were raised in two primary ways. First, the definition was too broad as it would include the behavior of any organism that is reinforced by a listener that is trained to do so. Second, it was non-functional as the behavior of the relevant organism (the speaker) relies on the history of a different organism (the listener).

Empirical findings on rules also uncovered important considerations relevant to language. Research on fixed interval (FI) reinforcement schedules revealed that humans with verbal abilities appear to respond differently than infants and other organisms (Bentall et al., 1985; Pouthas, 1985). While rules presented by experimenters appeared to influence performance, so did the production of self-generated rules (Lowe et al., 1983; Rosenfarb et al., 1992; Vaughan, 1985). Studies also demonstrated sensitivity effects as a behavior that is controlled by a rule can continue to occur even when the direct contingencies have been changed (Hayes et al., 1986). A clear behavioral account of language itself appeared relevant in order make sense of these findings and clarify conceptually what verbal stimuli are, but such an account was lacking at the time.<sup>2</sup> The research on stimulus equivalence however, provided new insight into how language could be approached from a behavioral perspective.

Sidman (1971) provided the first empirical demonstration of stimulus equivalence in a seminal study that aimed to teach a participant with a cognitive impairment to read. The participant could already match spoken words to pictures ( $A \rightarrow B$ ) and was taught to match text to spoken words ( $C \rightarrow A$ ). What incidentally resulted from this training though, was the subsequent emergence of

<sup>&</sup>lt;sup>2</sup> Skinner (1969) defined a rule as a verbal stimulus that specifies a contingency and suggested that a rule functions as a discriminative stimulus in relation to the listener. A few issues are evident. Since Skinner's definition of verbal behavior itself was too broad and not functional, it was unclear what would be considered a "verbal stimulus" within a rule. Secondly, rules cannot simply be discriminative stimuli, as stimuli become discriminative when they were present when a response has been differentially reinforced in the past. This issue is clearly evidenced when novel rules control responding, as these rules have not been contacted previously. Additionally, Skinner's account of verbal behavior considered the speaker but deemphasized the listener. This is problematic in relation to rule-governed behavior, as the listener is the one who understands and follows the rule. As such, there was not a way to define, in a technical sense, what is meant by "specifies" (Hayes & Hayes, 1989).

completely untaught responses of matching pictures to text (B $\rightarrow$ C) and orally naming the printed words (A $\rightarrow$ C). This experiment demonstrated that sets of stimuli without formal similarity can be derived as equivalent without having to directly train every relation. These relations that emerged are defined as *derived* stimulus relations as they were never directly taught. Subsequent research showed stimulus equivalence and language are closely linked. Stimulus equivalence has been evidenced in humans at young ages while absent in non-humans (Sidman et al., 1982), has been correlated with verbal abilities (Devany et al., 1986), and has been found to show similar brain activation patterns as verbal fluency tasks (Dickins et al., 2001). Further evidence has shown that a psychological function of one stimulus that is in a derived equivalent relation to other stimuli may transfer to those stimuli without additional training. This transfer of function effect has been exhibited with both respondent and operant functions (de Rose et al., 1988; Dougher et al., 1994) and has been shown to occur in stimulus equivalence training approaches to teach language skills, such as making requests (Rehfeldt & Root, 2005).

RFT emerged as a conceptual account that specifically proposed in operant terms where stimulus equivalence came from and why it was related to language. It explained that stimulus equivalence and other (non-equivalent) forms of derived relational responding could be considered as learned patterns of generalized relational responding and that in fact such responding was the key operant that underlay language and cognition. Additionally, RFT provided an account for the impact of rules or language more broadly, arguing that the difference in responding between verbal and nonverbal organisms is underscored by the ability to derive relations.

## **Overview of RFT**

Relational responding refers to behavior in which an organism responds to one stimulus in terms of its relation to another stimulus. According to RFT, this behavior can be divided into two varieties: nonarbitrary relational responding (NARR) and arbitrarily applicable relational responding (AARR) (see e.g., Stewart & McElwee, 2009).

## Nonarbitrary and Arbitrary Relations

NARR is a type of relational responding in which an organism responds to one stimulus in terms of its physical or formal relation to another stimulus (Hughes & Barnes-Holmes, 2016). Selecting a stimulus physically the same as another (AKA identity matching) is an example of NARR. The second variety of relational responding, AARR, is a more abstract type of relational responding that is based not on the physical relation between stimuli but instead on contextual cues that specify the relation between the stimuli. For example, imagine I tell someone unfamiliar with US currency that a dime is worth more than a nickel and they subsequently produce the untrained relational response that a nickel is worth less than a dime. In this case I have used the cue "more than" to specify for this person a relation of comparison between two stimuli without any physical coins being present and they have produced an untrained relational response in accordance with the cued relational pattern (i.e., they have "derived" the untrained relation). Such relations are "arbitrarily

applicable" in the sense that they can be applied to any stimuli irrespective of their actual physical properties (Blackledge, 2003). While NARR has been demonstrated both with humans and non-humans (Giurfa et al., 2001; Harmon et al., 1982), evidence for non-humans showing AARR is weak (Dugdale & Lowe, 2000; Galizio & Bruce, 2018; Hayes, 1989; Lionello-DeNolf, 2009; Sidman et al., 1982) or controversial at best (Dymond, 2014; Hughes & Barnes-Holmes, 2014; McIlvane, 2014; Zentall et al., 2014). Even in the studies showing them to pass such tests, the demonstration is relatively limited (Schusterman & Kastak, 1993; Urcuioli, 2008). Meanwhile, humans readily pass such tests from a very early age and go on to show multiple complex patterns of derived relations (Kirsten & Stewart, 2022; Lipkens et al., 1993). Due to this, it is this latter skill (AARR) that RFT considers as particularly relevant to complex human behavior.

## Patterns and Properties of AARR / Relational Framing

RFT suggests that humans can show a variety of patterns of AARR including, for instance, coordination (a cup is like a mug), distinction (lemons are different than limes), opposition (north is opposite to south), comparison (winter is colder than summer), spatial (the toy is under the couch), temporal (morning is before night), and hierarchical (a cat is a type of animal). These different patterns of AARR are referred to as "relational frames" (Hayes, Fox, et al., 2001) based on the analogy of a picture frame in which the content can differ while the frame or outline remains the same, with different types of relations (e.g., same, different, opposite, etc.) constituting different ways in which stimuli (as the content) can be framed. All frames involve three core properties: mutual entailment, combinatorial entailment, and transformation of stimulus function. Mutual entailment, the reversibility of relations, is demonstrated, for instance, if a child is taught that coin A is worth less than coin B and then derives without further training that coin B is more than coin A. Combinatorial entailment is shown when relations are combined with other relations to produce novel untaught relations. For example, if a child is taught that coin A is worth less than coin B and that coin B is less than coin C, then they might derive that A is less than C and that C is more than A. Transformation of stimulus function refers to the empirical phenomenon wherein a particular function of a stimulus (e.g., discriminative, reinforcing, motivative, perceptual, eliciting, etc.) is changed or transformed based on a derived relation with another stimulus (Dymond & Rehfeldt, 2000). For example, imagine the child in the previous example has direct experience with coin A such that it has become an appetitive stimulus but no direct experience with coin C. After deriving that coin C is worth more than coin A, upon subsequently being given the choice of coin A or coin C the child is likely to pick coin C rather than coin A despite their previous history, based purely on the derived comparative relation. This property of AARR is particularly important from an RFT view as it demonstrates the manner in which language can impact behavior and it has been empirically shown with a variety of functions and relational patterns (Stewart & Roche, 2013).

## **Contextual Control and Acquisition of Relational Framing**

Derived relational responses are under the control of environmental stimuli, referred to as contextual cues. RFT research has distinguished between two types of contextual cues: relational cues (Crel), which come to control the type of relation derived (e.g., coffee is better than tea; a house contains a kitchen), and functional cues (Cfunc), which come to control the stimulus function to be transformed in accordance with the relation (e.g., imagine the *smell* of coffee; *picture* your kitchen). RFT suggests that contextually controlled AARR is an operant that is established through multiple exemplar training (MET), wherein a variety of stimuli are used as relata but the specific contextual cues remain the same.<sup>3</sup> Through a history of differential reinforcement, a child's relational responding comes under the control of such cues (Hayes, Fox, et al., 2001). For instance, Barnes-Holmes and colleagues (2004) used MET to train comparison relations in 4- and 5-year-old children. Sets of two or three novel coins were presented and the children were told a relation between the coins such as, "if this coin (points to coin A) buys more/less sweets than this coin (points to coin B), which would you take to buy as many sweets as possible?" Following differential reinforcement for accurate responding, participants demonstrated derived comparison framing in accordance with "more than/less than" cues and passed tests of generalization with novel stimuli. Several studies have shown that different patterns of AARR can be established and/or strengthened in various populations through MET, such as coordination (Rehfeldt & Root, 2005), distinction (Dunne et al., 2014), comparison (Berens & Hayes, 2007), opposition (Y. Barnes-Holmes et al., 2004), hierarchy (Ming et al., 2018), deixis (Gilroy et al., 2015), and analogy (Carpentier et al., 2002).

### **Recent Theoretical Advances in RFT**

Barnes-Holmes and colleagues (2017; 2020) provided an updated model of the dynamics of relational behavior. The hyperdimensional multilevel (HDML) framework suggests that AARR can be experimentally analyzed along different levels and dimensions (Table 1.2)<sup>4</sup>. The HDML includes five levels of relational responding. *Mutually entailing* refers to the bidirectional relation between two stimuli (e.g., if A less than B, then B more than A). *Relational framing* involves the combining of

<sup>&</sup>lt;sup>3</sup> RFT suggests that AARR is operant behavior that is strengthened by a history of differential reinforcement in the presence of specific contextual cues. Operant behaviors can be seen to have four specific properties: they develop over time, are flexible/changeable, can be brought under antecedent control, as well as consequential control (Hayes, 1994; Hughes & Barnes-Holmes, 2016). Multiple studies have provided evidence of gradual acquisition of AARR in a variety of types of relations across childhood (Kirsten & Stewart, 2021; McHugh et al., 2004; Mulhern et al., 2017) while other studies have demonstrated that when certain relational framing repertoires are absent, they can be established through training (Y. Barnes-Holmes et al., 2004). The operant characteristic, flexibility (i.e., the behavior can be altered even after being learned), has also been shown with AARR. For example, O'Connor et al. (2011) taught participants to alternate between correct and incorrect responses in the presence of contextual cues. The different patterns of AARR that can be established is evidence for antecedent control as a wide variety of contextual cues can be established for responding. Consequential control of AARR has also been shown. For instance, Healy et al. (2000) manipulated the use of accurate and inaccurate feedback for AARR and demonstrated that both mutually and combinatorially entailed relational responses are sensitive to consequential control.

<sup>&</sup>lt;sup>4</sup> The HDML framework also includes a proposed new conceptual unit of analysis referred to as the ROE-M (relating, orienting, evoking, and motivating). In this approach, instances of AARR involve the combination of relating stimuli in various ways (Crel properties), as well as orienting toward stimuli and the evoking of responses under appetitive or aversive control (Cfunc properties) within a context that includes dynamic motivating variables. The ROE-M is analyzed as a singular unit as relating, orienting, evoking, and motivating all influence one another. See Harte and Barnes-Holmes (2021) for an overview.

multiple mutually entailed relations (e.g., if A less than B, B less than C, then A is less than C and C is more than A). *Relational networking* involves the combining of different relational patterns (e.g., if A less than B, B less than C, C same as D, D same as E, then deriving that D and E are larger than A and B). *Relating relations* involves relating one derived relation to another derived relation (e.g., if A1 is less than B1, and A2 is less than B2, then it can be derived that the relation between A1-B1 is the same as the relation between A2-B2). *Relating relational networks* involves the relating of one relational network to another relational network (e.g., if A1 less than B1 and B1 same as C1, and A2 less than B2 and B2 same as C2, then it can be derived that relational networks 1 and 2 are the same).

## Table 1.2

Levels	Dimensions			
	Coherence	Complexity	Derivation	Flexibility
Mutually Entailing	Analytic Unit 1	Analytic Unit 2		
Relational Framing				
Relating Relations				
Relating Relational Networks				Analytic Unit 20

Levels and Dimensions in the HDML Framework

Relational responding can also be examined along four dimensions. *Coherence* refers to the extent to which a relational response is consistent with one's learning history. If a child has previously learned that "Monday is before Tuesday" and is later asked if Tuesday is after Monday, responding "yes" would be considered coherent with their learning history in terms of relating these stimuli. *Complexity* refers to how "detailed" the relational response is (e.g., number of stimuli or relations, types of relations). For instance, "Monday is before Tuesday" entails two relations ("before" and "after") which is more complex than a difference relation such as "Monday is different than Tuesday," which entails one relation ("different than"). *Derivation* refers to the degree to which a relational response has been practiced. The first instance of deriving "Tuesday is after Monday" would be considered high in derivation; as that response continues to be emitted, it progressively acquires a history of direct reinforcement, and the derivation becomes lower over time. *Flexibility* refers to the extent to which a relational response can be modified by the current context. If a child is now told, "Monday is after Tuesday," the faster they derive the relation in this new context, the more flexible their relational response is.

## **Key Research Findings on AARR**

Evidence from multiple studies conducted over the past number of decades (see for example Hughes & Barnes-Holmes, 2016; Stewart & Roche, 2013) has been interpreted by RFT researchers as indicating that AARR is the critical repertoire that underpins symbolic language and cognition. For example, the development of AARR has been evidenced in humans as young as 17 months and parallels language development (Lipkens et al., 1993), and is correlated with verbal abilities (Barnes et al., 1990; Devany et al., 1986; Moran et al., 2014; Moran et al., 2015). Research involving neurophysiological methods has shown similar brain activation patterns between AARR and responding on verbal/semantic tasks including fluency (Dickins et al., 2001) and lexical decision making (Barnes-Holmes et al., 2005). Studies have also shown strong correlations between AARR and performance on standardized tests of cognitive abilities in adults (Colbert, Dobutowitsch, Roche, & Brophy, 2017; Colbert, Malone, Barrett, & Roche, 2020; O'Hora et al., 2005; O'Hora et al., 2008; O'Toole & Barnes-Holmes, 2009) and in children with and without disabilities (Belisle et al., 2018; Dixon, Belisle, & Stanley, 2018; Kirsten & Stewart, 2022). Other studies with school-age children have shown that training in relational skills may lead to improvements in performance on standardized tests of cognitive abilities (Amd & Roche, 2018; Cassidy et al., 2016; Cassidy et al., 2011; Colbert, Tyndall, Roche, & Cassidy, 2018; Hayes & Stewart, 2016; McLoughlin, Tyndall, Pereira, et al., 2020). Most recently, a randomized control trial evaluated the effects of training relational skills with 73 school-aged autistic children receiving ABA services (Dixon et al., 2023). Overall, the data showed that the group that received relational training made greater skill gains and scored higher on standardized intelligence measures compared to the treatment as usual group.

Over the past several decades, research inspired by RFT has continued to grow and the theory has accumulated strong empirical support (Dymond et al., 2010; M. O'Connor et al., 2017). RFT research has opened up a new behavioral approach to investigating human language and complex behavior (Stewart, 2016) including rules (Harte et al., 2020), analogies and metaphors (Foody et al., 2014), and perspective taking (Barnes-Holmes, McHugh, et al., 2004). Furthermore, evidence for the practical applied importance of RFT has been very prominent in this expanding program of work. **Summary** 

RFT is a contemporary behavior analytic approach to language and cognition that fits with the philosophy of functional contextualism, which specifies a scientific goal of both prediction and influence of behavior. While RFT emerged from a behavior analytic tradition, limitations with Skinner's analysis of verbal behavior, findings from research on rule-governed behavior, and the discovery of stimulus equivalence were specific milestones that paved the way for RFT. This approach emphasizes that the key to language and cognition is relational framing, the ability to derive relations between stimuli, irrespective of their physical properties. Relational framing is evidenced by the specific properties of mutual entailment, combinatorial entailment, and transformation of stimulus function and these properties can be observed across a variety of relational patterns that are under contextual control. Different levels of complexity of AARR can also be demonstrated, and instances

of relating can be analyzed along different response dimensions like coherence, complexity, derivation, and flexibility. AARR is a generalized operant that develops over time, is changeable, and comes under the control of both antecedents and consequences. Relational responding repertoires emerge across development in naturalistic language contexts through exposure to sufficient multiple exemplars, while experimental research has shown that arranging systematic training conditions can improve these repertoires when underdeveloped or absent.

Chapter 2. Time-Related Skills in Humans and Relational Frame Theory: A Review and Conceptual Analysis

Portions of this chapter have been accepted for publication:

Neufeld, J., & Stewart, I. (2022). A behavioral approach to the human understanding of time: Relational frame theory and temporal relational framing. *The Psychological Record*. https://doi.org/10.1007/s40732-022-00529-7

Time is a fundamentally important dimension of human experience and responding adaptively in terms of this dimension is critical to human personal and societal functioning (Brislin & Kim, 2003; Grondin, 2008). We organize ourselves and our activities, both short and long term, in time. This has been true throughout human existence but it has become particularly important in the post-industrial era in which organization and coordination of human activity has become critical to societal functioning (Thompson, 1967; Whipp et al., 2002). All organisms are sensitive to time at some level as all organisms respond to changes in their environment (Block et al., 1999; Brannon et al., 2007; Gibbon, 1977). From habituation and sensitization, to classical and operant conditioning, time is a common factor across all learning processes. However, there is an important distinction to be made between responding to time as a nonarbitrary dimension of existence (i.e., to the experienced sequence of ongoing physical changes in the environment) and responding to time as an abstract concept (i.e., time as described in symbolic language such as, "You'll have to wait a few minutes" or "I have a lot of time on my hands") (Hayes, 1992). It is the latter that is critical to the type of selfknowledge and societal organization that is unique to human life. The purpose of this chapter is to explain this concept and discuss relevant research both in terms of what has been done as well as potential future directions, especially in terms of the acquisition and training of this repertoire.

#### Past Behavioral Research on Responses to Time in Humans

How humans experience time has been a subject of interest in philosophical, natural science, and social science disciplines (Arstila & Lloyd, 2014; Einstein, 1905; Evans, 2013; Gale, 2016; Grondin, 2010; Levin & Zakay, 1989; McTaggart, 1908; Piaget, 1971). Within behavior analysis, Skinner (1938) notably dedicated a chapter on the temporal discrimination of the stimulus in *The Behavior of Organisms* (see Lejeune et al., 2006 for a discussion). He elsewhere indicated that operant behaviors, like those seen in reaction-time experiments, can be affected by adding temporal properties to a contingency, such as by reinforcing responses that are made as quickly as possible or by only reinforcing responses that are delayed for a period of time (Skinner, 1953). Within behavior analytic work, rate could be considered a key dependent variable. However, measurement of rate is usually used as a means of assessing some other variable, such as reinforcement, rather than there being a focus on temporal responding itself. Nonetheless, some behavior analytic methods have been used to empirically explore the human response to time as an aspect of the environment. **Respondent Conditioning and Temporal Responding** 

Historically, both respondent and operant approaches have been used to study human responding to time across different ages. Early behavioral research examined temporally conditioned responses in infants with respondent conditioning procedures. This approach involved the presentation of a stimulus on a particular temporal interval followed by trials that omit the stimulus to test if the infant's conditioned response continues to occur around the time the stimulus would have been delivered (Colombo & Richman, 2002). Such a response is said to provide an indication of the infant's sensitivity to duration. For example, Brackbill and Fitzgerald (1972) temporally conditioned a

pupillary reflex in 1-month-old infants. In their approach, changes in lighting were administered (e.g., 4s light on or 4s light off; UCS) at a constant 20s interval. They observed that these lighting changes elicited an unconditioned pupillary reflex (UCR) by the infants. Subsequently, they ran test trials in which the light changes (UCS) were not presented and found that the pupillary reflexes continued to occur, thus showing sensitivity to the passage of time as a conditioned stimulus (CS) by the production of conditioned pupillary response (CR). Researchers have also examined whether other responses such as heart rate deceleration might be temporally conditioned, but while some positive evidence has been provided, results have not been as clear or definitive as for the pupillary reflex (Colombo & Richman, 2002; Stamps, 1977).

## **Operant Conditioning and Temporal Responding**

Operant procedures have also been used to evaluate sensitivity to different duration values with regard to various types of reinforcement schedules. In particular, studies with Fixed Interval (FI) schedules of reinforcement have shown notable differences in responding in humans compared to nonhumans. Whereas nonhuman organisms have been shown to demonstrate a "scalloped" response pattern (a pause after reinforcement followed by a gradual increase in response rate) on FI schedules (Ferster & Skinner, 1957), adult humans do not tend to produce this pattern. Instead, studies show that adult humans produce either high-rate or low-rate responding during the interval, and studies also show that this performance can be substantially influenced by instructions (Baron et al., 1969; Leander et al., 1968; Lippman & Meyer, 1967; Matthews et al., 1977; Weiner, 1964).

Other behavior analytic research with adults has investigated the effect of verbal behavior on temporal responding more specifically. One such example is from a study by Lowe et al. (1978). In one condition, participants could press a panel that illuminated a binary clock that only signaled the availability of reinforcement (the panel lit up green when reinforcement was available, and white when it was unavailable). In the second condition, participants could press a panel that produced a digital clock that showed the time since last reinforcement in minutes and seconds. The authors found that performance in the digital clock condition was similar to a conventional scalloped response pattern on FI schedules; however, performance in the binary clock condition did not show similar sensitivity to the temporal parameter of the reinforcement schedule. One possible reason for the difference as suggested by Lowe and colleagues, may be the presence of counting in the former condition. When presented with a questionnaire about the experimental task, all participants accurately indicated that the delivery of points was contingent on the passage of time. However, the binary clock participants also stated that they counted to themselves until they estimated the duration to be finished whereas the digital clock group did not report any use of counting.

Young children's sensitivity to the temporal properties of contingencies have been evaluated using similar methods to those used with adults. These types of studies have typically involved tasks in which the delivery of a reinforcing consequence (e.g., cartoon presentation) is made contingent on the emission of a response (e.g., button pressing) in accordance with the respective reinforcement

schedule. Research using Differential Reinforcement of Low-Rate (DRL) schedules has shown it to be difficult for young children to withhold their response during the interval. While training can decrease the response rate in young children, their behavior does not necessarily match the set duration interval (Pouthas, 1985). Research with FI schedules has revealed that children produce greater response variability compared to ratio schedules, and that they may either wait for the interval to elapse before responding or show high rates of button pressing during the interval (Bentall et al., 1985; Zeiler & Kelley, 1969). Bentall and colleagues (1985) in fact noted that the expected FI response pattern observed in non-humans is frequently observed in infants but that as children begin to acquire verbal behavior, their response patterns on such schedules begin to change. Particularly, by the age of 5 years, children demonstrated either high-rate or low-rate response patterns, which is similar to what is seen in adults. Darcheville et al. (1993) extended this work to even younger participants. They found that infants between 3-5 months demonstrate sensitivity to various duration values (10s-80s). These 3- to 5-month-olds all showed post-reinforcement pauses characteristic of FI schedules in basic animal research, whereas 9- to 23-month-olds produced greater variability in response patterns. Furthermore, these response patterns (i.e., waiting for the interval to elapse, high rates of response) have been observed to persist across training sessions if children did not receive verbal instructions with regard to the interval (Droit et al., 1990).

A number of additional studies suggest an effect of verbal abilities on responses to time in children. In one study that incorporated different verbal rules as instructions for button pressing (e.g., "press harder," and "press longer"), performance differed across age. Whereas 3-year olds pressed the button with increased force as well as duration when told to "press harder," 5<sup>1/2</sup>-year olds pressed with increased force but not for longer durations and were more accurate with temporal instructions ("press longer," Droit-Volet, 1998). There is support that providing training using temporal instructions (e.g., to attend to the interval) can improve performance with both FI schedules (Bentall & Lowe, 1987; Droit et al., 1990) and DRL schedules (Pouthas & Jacquet, 1987). Lejeune et al. (1992) showed that when temporal verbal instructions are provided to children, better performance on the timing tasks is observed compared to same-aged children that did not receive the verbal instructions. Pouthas and colleagues (1993) argue that this is evidence that timing performance can come to be controlled by acquired "abstract representations of time" in addition to control by direct contingencies. Additionally, they suggest that providing temporal "information" is not strictly limited to issuing instructions vocally. Better performance has also been observed when children were given a supplemental visual stimulus in the form of a series of small light bulbs that changed from green to red, which indicated the passage of time (Droit et al., 1991). The authors posit that the children in this study abstracted temporal information from this external clock as they verbalized self-formulated rules such as, "I must wait until the [lights] turn red." (Pouthas et al., 1993, p. 211).

While there appears to be evidence that verbal behavior can alter human responses to time, there has as yet been insufficient research examining exactly how it does so from a behavioral

perspective. Learning to abstract rules that influence temporal responses appears relevant to language skills; however, it is not clear in a technical sense what this means and how this occurs. Beyond discriminating the durations of stimulus events, humans acquire further complex behaviors with time-related concepts. Critically, humans learn to organize events based on their temporal order and verbally construct sequences of past or hypothetical future events. Skinner (1957b) in fact suggested that verbal responses which describe a speaker's own past or future behavior are at least in part controlled by private stimuli and advocated for further analysis with respect to the variables that control these types of responses. Behavior analytic work has thus far insufficiently engaged on these more complex time-related repertoires in humans. In order to evaluate behaviors of this sort and other skills relevant to temporal understanding that seem to require verbal behavior, we suggest that an adequate functional analytic account of language and cognition is critical, and that relational frame theory (RFT) meets this criterion.

#### **Temporal Relations: An Overview and Present RFT Research**

RFT proposes that non-humans and humans experience time differently due to the fact that humans can respond to time verbally, that is, as described via symbolic language (Hayes, 1992). In RFT terms, this means engaging in AARR or relational framing with respect to time (Hughes & Barnes-Holmes, 2016) and RFT sees temporal relational framing as central in this respect. **Nonarbitrary and Arbitrary Temporal Relations** 

As with other relational frames, temporal relational responding has both nonarbitrary and arbitrarily applicable varieties. Hayes (1992, 1993) elaborated on this point when discussing a distinction between non-verbal and verbal time. In a nonarbitrary sense, time can be understood as a measure of change. This change is unidirectional in the sense that we always move from a "now" to a "new now." Non-verbal organisms experience this change through their direct contact with sequences of events in their environments. For example, a pigeon that has learned that in the presence of a particular stimulus order (e.g., a light that flashes green, then red, then blue) and not others that pecking a key results in obtaining food, has learned a temporal sequence through its direct experience. Different species including humans can learn through experience and respond to change in this nonverbal or experiential manner, although discriminating temporal sequences of this sort can take substantially more learning trials for non-humans (for a review, see Ghirlanda et al., 2017). As another example, a child may demonstrate nonarbitrary temporal relational responding in a context of being asked to repeat a set of actions shown by a model in a similar or different sequence to that shown by the model. The child is relationally responding in accordance with their actual experience of sequential change in this instance.

Learning to engage in arbitrarily applicable temporal relational responding (i.e., temporal relational framing), and hence to respond to time verbally, however, is unique to humans. This opens up vastly more extensive and transformative possibilities with respect to temporal responding (Hayes, 1992). One possibility is that this derived relational behavior allows humans to not just experience

time unidirectionally like non-verbal organisms, but to also verbally construct time as a bidirectional dimension in which events can be ordered (Hughes & Barnes-Holmes, 2016). For instance, a person may temporally frame events that have not been experienced from the past (e.g., my grandparents' lives before I was born) or the future (e.g., what happens after I die), and respond to these events as though they belong to a particular sequence. Temporal relational framing is also significant as it allows individuals to frame verbally constructed past or future events and bring functions of these events to bear in the present without needing to contact them directly. For example, a person may temporally relate events such as, "You get lung cancer after you start smoking," or "Save money before you retire to avoid poverty in old age." Such patterns of framing may transform the functions of the present context and support producing alternative, perhaps more beneficial patterns of behavior. For example, the functions of smoking may be transformed such that it acquires functions of lung cancer which may make smoking less appetitive. In the second example, the functions of saving may be transformed such that it takes on functions of poverty avoidance which may make saving more appetitive. Though the degree to which responding is altered in the present is ultimately dictated by the context, temporal framing allows for the possibility to respond to a verbally-constructed future without having to experience direct contingencies (e.g., involving aversive scenarios such as lung cancer or poverty) (Bach & Moran, 2008).

Whereas a nonarbitrary temporal relational response is primarily controlled by experienced change from one event to another, arbitrary applicable temporal relational responding is primarily under the control of a contextual cue that specifies the relation. In an English-speaking socioverbal community, common stimuli likely to be used as contextual cues for temporal relational framing include words such as, "before/after," "first/last," "earlier/later," etc. An RFT account argues however that these stimuli are not unique in and of themselves but rather that any stimulus can function as a cue for temporal framing depending on the reinforcement history of the relational response. For example, someone tapping a wristwatch or even just their wrist in a particular context might function as a cue that they wish an observer to speed up what they are doing. In this particular case RFT would suggest that this gesture participates in a relational network in which the gesture means something akin to, "Pay attention to the time" which might be further elaborated as, "It is now (or soon will be) AFTER the time at which you should be finished and there are aversive consequences for not finishing on time." The person would perform the task faster to the extent that their behavior was under the control of relevant contextual cues affecting transformation of those particular functions. **Properties of Temporal Relational Framing** 

Temporal relational framing exhibits the same three properties as relational framing more generally. An example of mutual entailment would be being taught that June is before July and then deriving without further training that July is after June. Mutually entailed relations can be combined to allow combinatorial type derivations. For example, if a learner is taught that June is before July, and July is before August, then they can derive that June is before August and that August is after

June. Additionally, the stimulus functions of an event can be transformed based on the derivation of temporal relations between that event and others. For example, imagine that an employee is told that upcoming meeting events A and B, which are separated from each other by a number of weeks, both have to be prepared for and that event A happens before event B. Based on this temporal relation, the employee may then prioritize their preparations by allocating more of their immediate time and attention to event A than to event B. As a second example imagine that we are told that something eventful happened at a party and want to find out what happened from potential witnesses. If we learn that Person A turned up at the party at some time after Person B, then we might be more interested in talking to Person B than to Person A.

It should be noted that arbitrary temporal relations do not pertain to the properties of the stimuli or the specific events that are being related. In other words, the relations involved typically do not refer to properties of the events being related but instead to properties of the context of those events. For example, deriving that A is after B only details the relative temporal order of stimuli A and B, not any of their attributes, such as color or size. While the features of temporally related events may remain unknown in some circumstances, it is possible that the ordinal functions of those events can be transformed through a temporal relation. For instance, given a relational context that A is after B and B is after C, a new response can be made about which event happens first (event C), or last (event A) perhaps in contrast to the way an individual might have ordered those events prior to deriving this relation.

## **Temporal Relations and Quantification**

Although temporal relations in accordance with "before" and "after" cues do not detail the physical properties of the relata, events themselves do have the nonarbitrary temporal property of extension in time. Arbitrary words used for temporal extent (e.g., minute, hour) can be related in frames of coordination with various stimulus events within a given relational network (Fig. 2.1). As such, more precise relations can be derived (Hayes, Fox, et al., 2001). Quantification can be applied to both order of events and duration of events. With ordering in time, quantification allows greater precision with respect to when events happen (e.g., if A is five minutes before B and B is five minutes before C, then the precise relations that A is at least ten minutes before C and C is at least ten minutes after A can be derived). With duration, quantification allows greater precision in terms of specifying extension in time (e.g., if A is ten minutes longer than B and B is five minutes longer than C, then we might derive that A is fifteen minutes longer than C and that C is fifteen minutes shorter than A). This verbal specification of when stimulus events begin and end allows even greater precision with respect to the temporal organization or classification of events. If a person frames both the order and duration of events, then they can make better predictions with respect to the timing of those events and their subsequent behavior in the context of those events may be more successful (see example derived relations in Fig. 1).

#### Figure 2.1

Example Relational Network Involving Quantification



## **Temporal and Other Relations**

## **Comparison Relations**

Temporal relations have occasionally been conceptualized as a particular subset of comparison relations (Dixon & Stanley, 2020). Comparison relations involve relating stimuli along a specified quantitative or qualitative dimension, which could include a temporal dimension. Temporal and other comparison relations share the feature of non-symmetry. If A is "bigger than" or "before" B, a non-symmetrical relation B is "less than" or "after" A is entailed. Temporal and comparison relations also share the feature of specificity when quantification is added to the relational context. If A is "twice the size" or "two minutes before" B, a more precise relation can be derived (Hayes, Fox, et al., 2001). However, comparison and temporal frames might be distinguished from one another in terms of a corresponding transformation of function (O'Hora & Maglieri, 2006). Transformation of function in accordance with a comparison relation may frequently involve changes to the physical response (e.g., pressing a button faster to earn a larger prize). With temporal relations, a transformation of function involves the occurrence or non-occurrence of a response (only pressing the button after the instructor says "go," and not before).

## Spatial Relations

Temporal relations can be likened to spatial relations, which may also be considered a subset of comparison relations. A relationship between space and time has often been appreciated across disciplines, such as in physics (Einstein, 1905), philosophy (Grünbaum, 1974) and psychology (Piaget, 1971). Within RFT, spatial and temporal framing are skills that support a person's ability to advantageously navigate their environment in space and time. Temporal relations between events are often be represented spatially due to the abstract nature of time, often corresponding with a language's writing direction in natural contexts. In (English speaking) socioverbal communities, certain
Chapter 2

contextual cues may even be used in the case of both spatial and temporal frames. For example, "before" might indicate a stimulus closer in space, or an event that occurred earlier in time. Conceptually, it is possible a more general frame is learned earlier in acquisition prior to finer discriminations between temporal and spatial features of the context. Nonetheless, empirical work is needed on this point regarding the acquisition of both frames. Temporal and spatial frames may also be considered component relations involved in deictic framing, which RFT suggests is the basis of perspective taking (McHugh et al., 2004).

## **Deictic Relations**

Temporal relations can also be either deictic or non-deictic. Deictic relations (I-You, Here-There, Now-Then) depend on the perspective of the person doing the relating (see e.g., McHugh et al., 2004; McHugh & Stewart, 2012). Deictic temporal relations involve relating events in accordance with contextual cues such as now-then or today-yesterday-tomorrow. A person relates events occurring at other points in time, whether past or future, relative to their current temporal perspective (e.g., Christmas was not that long ago; my birthday is X days from now). The derived response in these deictic temporal frames would change depending on the perspective. RFT-based research on deictic relations more broadly has revealed developmental patterns, and provided evidence that deictic relational frames seem relevant to perspective taking skills, and that they can be trained (Gilroy et al., 2015; McHugh et al., 2004; McHugh et al., 2009; Montoya-Rodríguez et al., 2017). Non-deictic temporal relations in contrast to their deictic counterparts do not depend on the perspective of the speaker. Events can be temporally related as before or after one another independent of the present moment perspective (e.g., university is after high school; Toy Story 1 was released before Toy Story 2). Indeed, a similar distinction between deictic and non-deictic temporal responding has been noted outside of the RFT literature (McCormack & Hoerl, 1999) and some theorists have suggested that learning to adopt temporal perspectives other than the here-and-now is an important aspect of a mature understanding of time (Cromer, 1971; McCormack & Hoerl, 2017; Weist, 1989).

# **Causal Relations**

Another issue concerning temporal relations is how they relate to causal or conditional relations. Causal/conditional relations are similar to temporal relations in that both specify temporal ordering of events. However, whereas a temporal relation describes events as simply occurring in succession, a causal or conditional relation additionally specifies a contingent relationship between events (e.g., if A, then B; A causes B, B is an effect of A) (Biglan & Barnes-Holmes, 2015; Carmelo Visdómine-Lozano, 2015; Hayes, Fox, et al., 2001). Hayes, Fox, et al. (2001) suggest that causality does not have a physical dimension, as "causes" of events may be based on any number of characteristics such as contiguity, manipulability, beliefs, etc. Furthermore, in the socioverbal community, contextual cues typically used for each of these frames appear to overlap somewhat flexibly, at least within the English language. For example, "*First* A, *then* B" may refer to a simple sequence of events, or refer to a behavioral contingency, or even a cause-effect relationship. It is

Chapter 2

necessary to consider the functional properties of the verbal relation on the listener in these circumstances rather than only the topography of the contextual cues. For the purposes of the present chapter because of the functional overlap between the two varieties of relations, causal/conditional relations will be treated as a special subcategory of temporal relations and hence will include the former in many of the analyses and examples. However, additional research is needed to explore to what degree temporal and conditional or causal relations should be considered functionally similar to or distinct from each other (Hughes & Barnes-Holmes, 2016).

### **Acquisition of Temporal Frames**

With respect to the acquisition of temporal relational framing, this repertoire is learned similar to other frames—through a history of multiple exemplar training (MET). Some authors have suggested that while temporal relations are similar to comparative and spatial relations (i.e., all involve relating events along a specified dimension), temporal frames are acquired later than either comparative or spatial relations, possibly due to the more abstract nature of ordering events along a temporal dimension (Hayes, Fox, et al., 2001). Some support in this respect can be found in research on deictic framing. McHugh and colleagues (2004) showed that spatial (here/there) relations appeared before temporal (now/then) relations, while other studies in this area point also toward temporal relations being more difficult to acquire than spatial (Gore et al., 2010; Weil et al., 2011). Deictic frame training protocols have generally recommended that temporal relations be trained after both interpersonal (I/you) and spatial (here/there) relations (McHugh et al., 2009). More recent RFT research examining the development of relational framing in children between 3-7 years across multiple frames and at different levels of complexity indicated that temporal frames appeared to be the last to emerge (Kirsten & Stewart, 2022). More research is needed however to determine the optimal point for introducing temporal relations in a relational frame training sequence.

## **Temporal Relations and Rules**

Temporal relations are also considered important in rules and rule-following. Skinner (1969) introduced the term rule-governed behavior to describe behavior that differs from contingency-shaped behavior. Whereas contingency-shaped behavior occurs due to a history of direct contact with contingencies, rule-governed behavior has not contacted the contingencies and is instead under control of "rules;" which he defined as verbal stimuli that *specify* contingencies. From an RFT view, verbal stimuli (e.g., rules) acquire their specifying properties via derived stimulus relations.

Rule-governed behavior very frequently but not invariably involves responding in accordance with derived relational networks with both equivalence and temporal relations (Harte et al., 2020)<sup>5</sup>. Equivalence relations are established between the words in the rule and the objects or events they refer to. The temporal contextual cues specify the order in which events or responses are to occur

<sup>&</sup>lt;sup>5</sup> Temporal relations may often be involved in rule-following, but this may not always be the case. Some simple rules might not specify a temporal relation (e.g., "do this" does not include a temporal contextual cue). Temporal relations are however arguably involved in any rules that specify a contingency since such rules tend to involve cues such as "if" and "then."

and/or the possible contingent relation between them (Figure 2.2). Derived relational responding in accordance with the rule may transform the function of events and actions within a particular context, and thereby introduce a new source of influence on behavior (Hayes & Hayes, 1989). For example, consider a parent introducing a rule to their young child within the context of a new game, "If you collect three tokens before the other player does, then you win the game." In this example, the child must engage in a number of relational framing responses for this statement to be understood and function as a rule, including frames of equivalence (between the words and the actual events or objects), and temporal frames (contextual cues "if" and "then" specify a behavioral contingency and the contextual cue "before" specifies a temporal order of events).

## Figure 2.2

Example Relations Involved in a Rule



Figure adapted from Tarbox, Campbell, et al. (2020)

If the child has a repertoire involving these frames, this can provide a source of behavioral regulation for the child as they can detect whether their behavior and actual events in the environment coordinate with the events described in the rule. There is already relevant empirical evidence on rule-following thus conceptualized. Some research with children with disabilities has shown that basic rule-following repertoires can be established and generalized to novel rules by teaching causal ("if/then") relations between antecedents and behaviors (Tarbox et al., 2011) and between behaviors and consequences (Wymer et al., 2016). Additional factors influence whether or not a rule is actually followed but without such a relational framing repertoire, the rule could be neither understood nor followed (Stewart & Roche, 2013; Tarbox, Campbell, et al., 2020). These studies further support the key role of temporal/causal framing in rule-governed behavior, but much additional research is

needed. For example, the children involved in these studies had already acquired the key framing repertoires needed and thus the job of the researchers was essentially to teach exemplars involving the combination of these frames. However, the history and training needed to establish key frames such as temporal/causal frames themselves in very young children and children with disabilities needs to be explored in detail as well as the potential range of behaviors that the acquisition of such frames facilitates.

## **RFT-Based Empirical Studies on Temporal Relations**

While research on temporal relational framing is as yet limited, there have been a number of studies conducted. Temporal frames were first examined in studies investigating rule-based (or instructional) control. O'Hora et al. (2004) trained and tested 12 undergraduates on a computer-based relational responding protocol that involved "before/after" and "same/different" relations across two experiments. In the case of the temporal relations, two arbitrary symbols were trained as contextual cues for "before" and "after" by reinforcing the choice of a "relational statement" involving one or the other cue as a description of a previously seen nonarbitrary sequence of shapes. Once responding in accordance with contextual cues for "before/after" and "same/different" relations was acquired, participants were then given a test for instructional control via a task in which they had to press computer keys in particular sequences that were specified by arbitrary relational networks involving all four cues. Sequenced responding was observed to be controlled by the relational cues in 8 of 12 participants as they met performance criteria on the tests for instructional control. These performances thereby demonstrated transformation of an ordering function in accordance with the temporal contextual cues. In the second experiment, each of the participants who reached criterion on a test for instructional control additionally showed generalization of responding to novel stimulus sets. Accordingly, these results provided preliminary evidence both that derived temporal relational responding could be induced in a laboratory setting and that it could contribute to a laboratory-based model of rule-governed behavior.

Two subsequent studies expanded on these findings by using the O'Hora (2004) relational testing and training protocol to correlate derived relational responding performance with scores on a standardized measure of cognitive abilities. O'Hora et al. (2005) trained and tested seventy-five undergraduate students on a protocol similar to the one employed in O'Hora (2004). Once mastery criteria were reached, participants completed the vocabulary, arithmetic, and digit-symbol coding subtests of the Weschler Adult Intelligence Scale (WAIS-III). Results showed that mastery performance on the relational responding task was a significant predictor for performance on both vocabulary and arithmetic subtests. A follow-up study by O'Hora et al. (2008) examined whether performance on a similar temporal relations task predicted performance on all four indices of the WAIS-III in eighty-one undergraduate students. Results showed significant correlations between percentage of correct trials on the temporal relations task and scores on the Verbal Comprehension,

and Perceptual Organization indices. Significant correlations were also found between relational responding and each subtest of the Verbal Comprehension index.

A number of more recent studies have investigated accuracy and speed of responding to "before" and "after" contextual cues in the context of nonarbitrary ordering tasks. While this work was centered ultimately on a nonarbitrary ordering task, the findings are likely relevant also with respect to temporal framing. Hyland et al. (2012) delivered a training/testing procedure with "before" and "after" cues to university students. In Experiment 1 participants observed a sequence of two physical shapes (e.g., a square followed by a circle) on a computer screen and were then shown an array of four physical shapes (square, circle, triangle, and cross) on both the left and right side of the screen. Following this, either the word "before" or "after" was presented in the middle of the screen between the two arrays of shapes and participants were required to select the shape on both the left and right of the cue word that would correspond to the order in which the two original shapes had appeared. For example, if a circle appeared following a square and the word "before" was then shown between the two shape arrays, then the participant would have to select the square on the left and the circle on the right. Results showed that participants responded significantly faster on "before" cue trials than on "after" cue trials and a second experiment replicated this result using abstract as opposed to familiar stimuli. A subsequent study by Hyland et al. (2014) further examined speed and accuracy with temporal contextual cues using a slightly different methodology. On each trial participants were shown a relational statement involving physical shapes with either the cue "before" or "after" between them (e.g., circle BEFORE triangle). After viewing the statement, participants were given an array of physical shapes and asked to select the shapes in the order specified by the relational statement (e.g., choosing a circle and then a triangle). Results revealed no difference in accuracy between "before" and "after" trials, but response speeds were again significantly faster for "before" compared to "after" trials.

McGreal et al. (2016) extended these findings by comparing performance on a temporal relations task between younger (M = 19 years) and older (M = 61 years) participants. A go/no-go task was used in which each participant first observed a sequence of images on a computer screen (A...B) and was then presented with a temporal relational statement involving the two stimuli involved (A BEFORE B; B BEFORE A; A AFTER B; B AFTER A). If the relational statement accurately described the sequence shown, participants had to provide a key-pressing response whereas if the statement did not describe the sequence accurately, they were to make no response. Results showed that the older participants demonstrated slower response speeds compared to the younger participants. Across both groups however, participants showed less accurate and slower responses to statements involving the "after" cue (e.g., A AFTER B) than to statements involving the "before" cue (e.g., A BEFORE B).

Overall, these findings suggest that when asked to engage in nonarbitrary temporal relational responding (i.e., identifying the temporal order of events) under the control of temporal relational

cues, respondents are faster to respond given a "before" cue than an "after" cue. A subsequent paper by Brassil and colleagues (2019) offered a possible explanation for this finding. Even in arbitrary relational statements, a nonarbitrary temporal relation is in effect. Consider an arbitrary relational statement, "A before B." As an individual either hears that statement spoken or reads the statement, they are responding to a nonarbitrary relation. In other words, the arbitrary relata in the statement itself (i.e., "A" and "B") are being experienced sequentially in time. It is possible that subsequent correspondence or non-correspondence between the nonarbitrary order of actual events and the order of the names for those events as they appear in the relational statement (i.e., the order in which the names are heard or read) may influence response latency. For example, if an arbitrary relational statement "B after A" is an accurate description of an event sequence (i.e., A happens and then B happens) there is non-correspondence between the nonarbitrary relationship between the spoken or read stimuli B and A in the statement and the nonarbitrary relationship between the physical events described by that statement, which may affect response latency. Brassil et al. (2019) demonstrated that this increased response latency with reversed relational statements can occur not only with the temporal cues, "before/after," but also with cues for other relations. More specifically they showed that it could occur for the comparative cues "bigger/smaller," supporting previous findings that patterns of relational responding share similar characteristics (Steele & Hayes, 1991). Nevertheless, the fact that this effect is caused by nonarbitrary temporal relations suggests the possibility that it might impinge on arbitrary temporal relational responding, particularly when individuals are first learning this repertoire. For example, it has been shown in previous research that nonarbitrary relational responding can supersede arbitrarily applicable relational responding on tasks involving competing sources of stimulus control (Kenny, Barnes-Holmes, et al., 2014). Competition between nonarbitrary and arbitrary temporal relational stimulus control, especially in a developmental context, should be a focus of future research.

Despite the growing research attention devoted to derived relational responding more generally, research on temporal relations has presently only been conducted with typically developing adults with already strong verbal repertoires. Questions remain on which types of relational skills and other prerequisites are needed to establish temporal relational responding, and which types of training methods are effective with children and neurodiverse populations. Accordingly, a key focus of future RFT research should be to examine the acquisition of temporal relational framing in young children and to investigate in addition how this repertoire can facilitate responding to time as an abstract verbal concept.

## **Research on Temporal Skill Sets and RFT Application**

Derived (arbitrarily applicable) relations, and particularly temporal relations, allow the derivation of rules about sequences of events and this changes how humans respond to their environment (O'Hora et al., 2004; 2014; Stewart & Roche, 2013). These acquired relational skills are critical to human functioning at both individual and societal levels. As such, it behooves both basic

and applied researchers to provide a functional understanding of how these repertoires are acquired and how they can be trained when absent or deficient and RFT is well-equipped to provide this. In the following section, we offer a range of important repertoires in humans that require the capacity for abstract temporal responding and hence in RFT terms require arbitrarily applicable temporal relational responding. These repertoires include sequencing, timing, use of conventional time systems, selfcontrol, planning, and time management. This list of repertoires is not intended to be exhaustive but rather to draw the reader's attention to key areas of investigation in which derived relations may be relevant to a human understanding of time. We argue that an RFT paradigm offers both theoretical and applied implications for future research into the human understanding of time in these areas. For each area, we will first provide a brief overview of relevant literature and will then offer recommendations that exemplify how to possibly apply RFT to the topic.

### **Temporal Sequencing Skills**

#### **Summary**

Temporal sequencing or event ordering refers to the ability to order events according to a temporal dimension. This skill is important to organizing events, scheduling, and sharing about the past or future. Most research in this area has stemmed from cognitive developmental or psycholinguistic frameworks (Zhang & Hudson, 2018). Research with children has evidenced improvements in temporal sequencing as language and cognitive abilities improve throughout development (McCormack, 2015; McCormack & Hanley, 2011). A variety of methods have been used to evaluate this skill such as picture sequencing tasks, answering questions about the temporal event order, and acting out sequences in a specific order (Amidon & Carey, 1972; McColgan & McCormack, 2008; Moore et al., 2014; Pyykkönen & Järvikivi, 2012). Age-related performance in children with these tasks appears sensitive to particular aspects of the context that are emphasized (Bauer & Mandler, 1989; Clark, 1971; Friedman, 1990). The research presented below is organized by age-related performance with temporal ordering tasks that have focused on different aspects of the context.

**Ordering Events Forward and Backward.** In sequencing tasks, events can be ordered going forward (e.g., starting with the first event, and sequencing the events that come after) or backward (e.g., starting with the last event, and sequencing the events that come before). Research has shown forward ordering to be easier for young children than backward ordering. Fivush and Mandler (1985) assessed performance in 4-6-year-olds on picture sequencing with familiar events (e.g., going to McDonald's) and unfamiliar events (e.g., going parachute jumping) and found that overall performance improved with age. Across ages and materials, children performed best with familiar events in forward order followed by unfamiliar events in forward order, familiar events in backward order. By 5 years, children may sequence pictures of daily events forward and backward, and judge forward order from multiple points of reference (i.e., ordering events when beginning at different points in the sequence) (Friedman, 1990). Other research

has shown that it was not until mid-adolescence that children accurately judged the backward order of days and months from multiple reference points (Friedman, 1986). More recently, Moore et al. (2014) observed that in children aged 5-10 years, backward ordering daily events was more difficult than forward ordering and relative ordering (a sample picture is given and selecting which picture comes before or next in the sequence).

**Ordering Invariant and Variant Events.** The relative connection between events as they occur in the environment (i.e., whether one event tends to naturally lead to another event or not) may also affect performance on temporal sequencing tasks. Invariant events refer to events that always occur in this order in the real world (e.g., grocery shopping prior to leaving with grocery bags). Variant events are arbitrary sequences that can occur in any order (playing with toys before or after eating a snack). Carni and French (1984) assessed preschoolers' comprehension of temporal words by using pictures depicting a story of either invariant or variant sequences and asking them questions about what happened before or after particular events. Results indicated that 3-year-olds were able to answer "before" and "after" questions with only invariant sequences while 4-year-olds could answer both types of questions accurately. Some research has also examined recall of temporal order in even younger children. Bauer and Mandler (1989) found that 1- to 2-year-olds performed better with causal sequences on immediate and deferred imitation tasks (i.e., reproduce action-sequences in the correct order) compared to arbitrary sequences.

Ordering Past and Future Events. Other research has investigated temporal order in terms of past or future remoteness from the present. Sequencing future events has been observed to be more challenging for children than past events. In a study on children's reporting of past events, Friedman (1991) presented two unrelated novel events, with the second event taking place six weeks after the first. Children were subsequently shown cards about the events one week later and asked questions about them. Four-year-olds were able to discriminate which event was more or less recent but could not specify their temporal location with conventional time units (e.g., estimating the month or approximate day of occurrence). Between 6-8 years, children showed increased accuracy with reporting on when the past events occurred with reference to calendar time. This finding with past events has been extended in other situations and with varying approaches, such as using a spatial timeline (Friedman et al., 1995; Friedman & Kemp, 1998). Friedman (2000, 2002) later showed that children's discrimination of the relative temporal distances of future events also improves with age. In the Friedman (2000) study, children between 4-10 years were tested on future event ordering by answering questions about how far into the future certain events would occur or by pointing to parts of a spatial timeline. In contrast to performance with past events, 4-year-olds failed to accurately discriminate future events as coming "soon" or "in a long time." Between 5-7 years, children could discriminate between events in the near or more distant future, and between 7-10 years, children gradually improved in using conventional time units to reference events (e.g., estimating how many months from now).

McCormack and Hanley (2011) extended these findings by directly comparing performance on a past/future order task in 4-5-year-olds. Children were shown two pictures of familiar past or future holidays and asked questions about their temporal order relative to the present (e.g., "Which happened a short/long time ago?" or "Which will happen a short/long time from now?"). Both age groups performed well with "past" questions but only 5-year-olds answered "future" questions with a similar level of accuracy. Performance on the past and future reasoning tasks did not correlate with each other, however performance on both tasks correlated with comprehension of the terms "before" and "after."

**Order-of Mention and Event Order.** Performance with temporal order may also be influenced by the position of temporal terms like "before" and "after" in sentences. A description or instruction about an event sequence can be stated in multiple ways (Table 2.1).

#### Table 2.1

*Example Arrangements of Sentences Describing a Temporal Sequence* 

Event order	Order of mention		
Ate dinnerwatched TV	<ol> <li>"She ate dinner before she watched TV."</li> <li>"After she ate dinner, she watched TV."</li> <li>"Before she watched TV, she ate dinner."</li> <li>"She watched TV after she ate dinner."</li> </ol>		

*Note.* Event order refers to the actual temporal sequence of events as they occur. Order of mention refers to how that series of events is verbally described.

In sentences 1 and 2, event order and the order-of-mention match, whereas in sentences 3 and 4 the event order and order-of-mention do not match. Experimental methods have compared performance in children across each sentence type by either showing a modeled sequence of events and asking participants questions about the event order, or by presenting a temporal instruction and having participants perform sequenced actions. Multiple studies have found that young children perform better with matching sentences (sentences 1 and 2 in Table 2.1) compared to non-matching sentences (sentences 3 and 4 in Table 2.1) (Blything et al., 2015; Clark, 1971; Keller-Cohen, 1987; McCormack & Hanley, 2011; Trosborg, 1982). Errors in comprehension of these terms have been demonstrated up to early adolescence depending on their sentence placement. Pyykkönen and Järvikivi (2012) assessed 8-12-year-olds using a written questionnaire that included these four sentence varieties. Participants were asked to answer comprehension questions about the order of the events referenced in the question. Results showed that the children's accuracy was affected by the position of "before" and "after" within the sentences. While performance generally increased with age, even the 12-year-olds did not perform to the levels of accuracy that an adult comparison group demonstrated.

**Ordering Events across Languages.** The way in which children order events may also depend on their learning histories in their respective socioverbal communities. Some research has

investigated this by examining how time is represented across different languages. Boroditsky (2001) suggested that since "time" is not observable, it is most often specified in languages using spatial metaphors (e.g., whether time moves forward/backward, left/right, up/down, past us, or through us). Since these terms are used to represent the unobservable aspect of time, there may then be variability across languages in how time is expressed. Boroditsky (2001) showed that English-speakers often used horizontally oriented metaphors to talk about time (e.g., the past is *behind* us, the future is *ahead* of us), while Mandarin speakers additionally used vertically oriented metaphors for representing time (events move from *up* to *down*). Temporal order also tends to be spatially represented in accordance with writing direction. Whereas the order of events is conventionally arranged left-to-right in languages such as English, languages that read right-to-left (e.g., Arabic) tend to spatially order events right-to-left (Fuhrman & Boroditsky, 2010). Moreover, Boroditsky and Gaby (2010) observed that an Australian aboriginal community ordered pictures east-to-west instead of a relative order like left-to-right or toward/away from the body on a sequencing task. In other words, the direction they arranged the pictures changed depending on which direction they were seated so that the pictures consistently progressed from east-to-west.

Further research by Hendricks and Boroditsky (2017) examined the effects of learning new relational metaphors on representations of temporal event order. Undergraduate English-speaking students were trained in new ways to talk about time. One group was trained that earlier events were "down" (e.g., Thursday is *lower than* Friday) whereas another group were trained that earlier events were "up" (e.g., breakfast is *above* dinner). Both groups subsequently completed a space-time implicit-association task. Results showed that with sufficient exposure and response training with a new representation of time, participant responses were quicker with trials on the implicit-association test that were congruent with the representation of time that they learned during training. The authors noted that differences between the two groups were similar to those observed in cross-cultural studies of time representation. These findings show how language and social context may influence the way in which a person uses language to represent time.

## **RFT** Extension

A challenge raised in the cognitive developmental literature is that it is difficult to fully examine the role of language acquisition in the development of time concepts in children with the methods commonly used. As language is necessary to describe the more abstract nature of time, methodologies have necessarily relied on tasks involving language (McCormack & Hoerl, 2017; Zhang & Hudson, 2018). Nelson (1996a, 2007a, 2007b) has argued that children learn how to temporally organize events and reference themselves or others in the past and future through verbal interactions (e.g., sharing of past and future experiences, direct instruction of conventional time systems). However, cognitive developmental and psycholinguistic research has largely been descriptive and has deemphasized environmental conditions that give rise to the acquisition of these skills in children.

Mainstream literature has indicated that as children age, the way in which they orient themselves in time tends to shift and become increasingly abstract (McCormack & Hoerl, 2017). From an RFT perspective, we suggest that this happens based on the gradual acquisition of temporal relational responding. In the younger years, children temporally relate events at the nonarbitrary level (i.e., the change from one event to another). Across a wide variety of exemplars, contextual cues used with these nonarbitrary events come to control arbitrarily applicable temporal relational responding. The acquisition of AARR allows for a person to increasingly experience time in a more verbal or abstract sense. As such, a child's ability to temporally order events might be approached by examining temporal relational responding at both nonarbitrary and arbitrary levels.

This critically important RFT-based distinction between nonarbitrary and arbitrarily applicable temporal relations can be used to analyze and understand findings from cognitive developmental research. Studies in which children observe a sequence of actions and respond to their temporal order (e.g., Blything et al., 2015; McCormack & Hanley, 2011) are focused on temporal NARR as the relational response is controlled by the nonarbitrary feature of sequential change. Such tasks assess a simpler, more concrete level of responding than tasks targeting temporal AARR. With temporal AARR, no sequence of actual physical events need be involved as the relational response is instead primarily under the control of particular (temporal) contextual cues. For example, Pyykkönen and Järvikivi (2012) used a written questionnaire to assess "before" and "after" such that children had to temporally order events based on the presence of the temporal cue in the sentence. The researchers reported challenges in children even up to the age of 12 years. From an RFT perspective, this finding makes sense given the more abstract nature of arbitrarily applicable relations, as well as the fact that time is a relatively verbally constructed aspect of experience (see overview of temporal relations section).

This can make sense of other findings, such as data indicating that children's understanding of relations between future events is more difficult than with past events (Friedman, 2000, 2002; Friedman & Kemp, 1998). Relating future events involves increased arbitrariness given the fact that past events can be directly experienced whereas future ones are always necessarily hypothetical. Indeed, McCormack and Hanley (2011) provided a similar, yet cognitively oriented reason for their results showing that future reasoning was more difficult for children than reasoning about past events. The authors suggest that while children need to understand before-after sequences about past events, they do not need to construct the sequence and can recall the temporal order of events. The same approach for future events does not work however, and they must construct novel temporal sequences. Bringing an RFT approach to such findings might allow novel insights and exploration with a more practical educational orientation.

Future RFT research can take multiple directions to address present limitations in the literature on children's understanding of temporal order. One direction includes examining age-related performance on increasingly difficult tasks of temporal relational responding (e.g., from nonarbitrary

to arbitrary relations, from mutual to combinatorial entailment). Changes in children's temporal understanding have been observed between the ages of 3-5 years and some accounts have proposed that a critical shift in how children think about time occurs within this age range (Atance & O'Neill, 2005; Busby Grant & Suddendorf, 2010, 2011; McCormack & Hanley, 2011; McCormack & Hoerl, 1999; Perner 2001; Weist, 1989). RFT research could investigate whether the acquisition of temporal or other patterns of relational framing corresponds with improvements on temporal sequencing tasks during these ages. Similarly, researchers might consider age-related performance with temporal deictic relations ("now/then") and compare this to performance on tasks involving the relative distance of past/future events (e.g., Friedman, 2000). Exploring age-related performance with temporal relational responding at increasing levels of complexity throughout childhood may offer new insights into whether derived relational skills are related to performance on other measures of temporal understanding.

A second direction is to establish effective training methods to improve temporal relational responding when absent or deficient. While current research has revealed age-related and context-specific performance differences on a variety of tasks of temporal understanding (McCormack & Hanley, 2011; Moore et al., 2014), there is currently minimal research evaluating how these skills can be trained. RFT suggests that temporal frames can be trained using MET, as has been successfully accomplished with various other relational patterns (Berens & Hayes, 2007; Mulhern et al., 2018). In one version of MET, a specific relational response might be trained using a particular stimulus set and once established, a novel stimulus set might be used to test for generalization. If the participant fails the generalization probe, they can then be trained to mastery on the failed test set before being tested again for generalization on another novel set. This testing-training approach is repeated until sufficient exemplars have been used to establish the generalized relational response.

When conducting MET, nonarbitrary temporal relations may first be trained if not already present in the learner's repertoire by presenting multiple exemplars of stimulus sequences (e.g., stimulus A presented first followed by stimulus B). The stimuli are rotated across trials and the learner is asked which stimulus came before or after the other one. During training, correct responses are reinforced and incorrect responses receive error correction until the goal criterion is met. Once nonarbitrary temporal relational responding is demonstrated, the next phase is to transfer contextual control to arbitrary stimuli. To establish arbitrarily applicable temporal relations, a learner may be presented and trained with two or more stimuli simultaneously with a contextual cue indicating a temporal relation between stimuli (e.g., shown an image that reads A is before B, B is before C) while being tested for derivation (e.g., "Is B after A?", "Is C before A?", etc.). At this level, an instructor may consider not presenting the physical stimuli sequentially. This would introduce a nonarbitrary temporal relation which may interfere with establishing appropriate stimulus control of the relational response under the "before" and "after" arbitrary contextual cues. Testing for a transformation of function in accordance with a temporal relation might be done in a variety of ways. For example, a

game context might be employed in which various action functions are first differentially trained to a set of stimuli (e.g., blue card-raise hand, red card-clap, yellow card-wave). When presented with the stimuli interspersed with temporal contextual cues (e.g., yellow after red after blue), transformation of function is shown if the child is able to perform the actions in the order specified (i.e., for this example a correct response occurs if the learner first raises their hand, then claps, then waves).

Future RFT research could also investigate whether differences in the types of stimuli used as relata support or hinder the acquisition of deriving temporal relations. A few studies with equivalence relations have shown that using familiar stimuli may be more efficient and lead to higher likelihood of derived responses occurring compared to unfamiliar stimuli (Fields et al., 2012; Ming et al., 2015; J. O'Connor et al., 2017). Considering that a temporal dimension is inherently more abstract than other stimulus dimensions (e.g., physically perceptible features like color or size), this may be particularly relevant to temporal frames. If using familiar stimuli, an additional consideration might be whether derived temporal relations with "before/after" cues are more easily demonstrated with invariant events (e.g., go to bed, fall asleep) compared to variant events (e.g., play the piano, watch TV). Sequencing invariant events appears to be easier than variant events in early childhood (Carni & French, 1984). It is possible that this finding may be relevant to training temporal relations. Comparing rates of acquisition of AARR with differing stimuli may help point towards potentially more efficient training regimens. Examining potential effects of using invariant versus variant stimulus events may help not just in facilitation of training, but also in examining the distinct trajectory of causal framing as a possible subset of temporal framing.

Along these lines, future research with temporal frames may be directed toward how this repertoire can be supported by, combined with, or augment other relational skills. For example, do derived non-deictic temporal relations precede or support the acquisition of deictic temporal relations, or vice versa? In a series of studies, Kent et al. (2017) directly compared relational training sequences for the frames of coordination, distinction, opposition, and comparison, wherein the order of training for opposition and comparison frames was manipulated. Results provided preliminary evidence that training the relation of opposition before comparison resulted in faster acquisition of comparison relations. Similar research might extend this work by including temporal and deictic relations. Further, given the similarity between temporal and comparison relations, perhaps training the latter to a high strength might boost learning of the former.

Finally, research questions with temporal contextual cues could be explored in light of the findings from other psychological approaches. For instance, some research has found better performance with the term "before" compared to "after" in children (Table 1; Blything et al., 2015; Clark, 1971). RFT research might investigate whether children show differences in performance with one particular contextual cue (e.g., "before") compared to another (e.g., "after") on relational responding tasks. Research might examine whether training temporal AARR is best facilitated by training one cue, the other, or both concurrently. Research on temporal relations with adults has

shown faster responses with "before" cues compared to "after" but no work has yet been done with children (Hyland et al., 2012, 2014; McGreal et al., 2016). Further, linguistic studies have shown that the placement of "before" or "after" in a sentence can influence comprehension and subsequent demonstration of sequenced actions (Keller-Cohen, 1987; Trosborg, 1982). An RFT account might suggest that one interpretation of this finding is that a nonarbitrary temporal relation (i.e., the order in which each event is heard spoken aloud) is interfering with the arbitrarily applicable relation (i.e., how the events are to be related based on the contextual cue). This could be examined in the context of rule-governed behavior such as: how does the placement of a contextual cue in a sentence influence rule-understanding and in-turn, rule-following? Tasks that require participants to act-out sequences of actions given a novel sentence with "before" or "after" (e.g., Clark, 1971) share similarity to demonstrations of rule-based control and a transformation of ordering function. RFT research could examine how the arrangement of temporal contextual cues in relational networks might influence rule-following behavior.

## **Human Timing Skills**

#### **Summary**

A second key area of investigation is interval timing, which refers to the ability to accurately discriminate the duration of stimulus events. Discriminating how long events last is important to predicting the timing of events, such as engaging in a behavior at the "right time" or waiting during periods of delay (Casassus et al., 2019). Research on timing has often examined seconds to minutes intervals, though it may also extended to shorter and longer durations (Droit-Volet & Meck, 2007).

Historically, resistance to control by simple interval schedules of reinforcement as children age (e.g., see Bentall et al., 1985; Droit et al., 1990) presented a challenge with using conditioning procedures to assess timing in children and alternative paradigms were subsequently used to explore this phenomenon (Droit-Volet, 2011). At present, the predominant approach to timing within psychology is Scalar Expectancy Theory (SET). SET is based on an information-processing model that suggests that organisms have an internal clock system that is used for making temporal discriminations. The scalar property of the theory is represented as an organism's estimate of an interval's duration is respective to the interval's actual length. That is to say, durations that are longer (e.g., 100s) are likelier to have more variability of response than shorter durations (e.g., 10s; Church, 1984; McCormack, 2015). Duration discrimination has been found to fit Weber's Law and multiple studies have demonstrated that this scalar property of timing is observed in both non-humans and humans (Brannon et al., 2007; Gibbon, 1977; Wearden & Lejeune, 2008). Although SET has been the predominant model in duration judgment research, various alternative models of timing have also been presented not involving an internal clock, such as the behavioral theory of timing and learning to

time model (Dragoi et al., 2003; Killeen & Fetterman, 1988; Machado, 1997; Staddon & Higa, 1999; Wackermann & Ehm, 2006)<sup>6</sup>.

Research suggests that young children, including infants, may accurately discriminate the duration of events. Multiple studies with infants have demonstrated that duration discrimination is ratio dependent rather than being based on absolute durations and that discrimination improves with age. Using a visual habituation procedure, vanMarle and Wynn (2006) showed that 6-month-old infants can discriminate between durations that differ at a 1:2 ratio (e.g., 0.5s vs 1s) but not at a 2:3 ratio (e.g., 0.67s vs 1s). Brannon et al. (2007) replicated this finding and additionally found that 10-month-olds could discriminate at a 2:3 ratio but not at a 3:4 ratio.

#### Figure 2.3

Hypothetical Temporal Bisection and Temporal Generalization Graphs



*Note.* The left-side graph displays hypothetical data on a temporal bisection task. The right-side graph displays hypothetical data on a temporal generalization task.

With children, two major research methods have often been used to examine timing skills: the temporal bisection task and temporal generalization task. In a temporal bisection task, participants are given two sample durations—a "short" duration and a "long" duration (e.g., a 500ms sound and a 1.0s sound). Participants are asked to categorize subsequently presented comparison durations that are either equal to one of the two samples or are in between them in terms of duration. A bisection curve is then constructed by graphing the proportion of "long" responses relative to the sample duration, with steeper bisection curves indicating finer temporal discriminations (Figure 2.3). In a temporal generalization task, participants are given a sample duration (e.g., 500ms) followed by comparison durations (e.g., 200ms, 800ms, etc.). Participants are then asked to judge (by responding yes or no) whether the comparison duration matched the sample duration or not. Generalization gradients are

<sup>&</sup>lt;sup>6</sup> An analysis of these different models is beyond the scope of this chapter. For this interested reader, see Allman et al. (2014) for a review on the internal clock model. For arguments against an internal clock, see Eckard and Lattal (2019). See Machado et al. (2009) for a comparison between SET and the learning to time model.

constructed which represent the proportion of "yes" responses, with steeper generalization gradients again representing finer temporal discriminations (Figure 2.3). Research utilizing these approaches with children have shown developmental differences across both tasks. On the temporal bisection task, 8-year-olds have shown higher temporal sensitivity compared to 3- and 5-year-olds (Droit-Volet & Wearden, 2001). Similarly, on the temporal generalization task, 8-year-olds have produced finer temporal discriminations compared to 3- and 5-year-olds (Droit-Volet, Clement, & Wearden, 2001; McCormack et al., 1999).

A few other methods have also been used to assess children's duration judgments. These tasks include verbal estimation (e.g., "tell me how many seconds that was"), production (e.g., "Say 'start' and then 'stop' when 1 minute has passed"), and reproduction (e.g., a 10s sound clip is played, and the child is told to "press this button for the same amount of time"). In a meta-analysis by Block and colleagues (1999), age-related differences were found between children (7-12 years), adolescents, and adults with these methods. Results showed that children made larger verbal estimations, comparable productions, and shorter reproductions compared to adolescents and adults. Additionally, the interindividual variability in duration estimations decreased with age regardless of the type of duration task.

Language and verbal understanding typically play a central role in studies of duration estimation, predominantly in the form of instructions. A distinction has been made between implicit (i.e., prospective) and explicit (i.e., retrospective) timing tasks. Wherein the former involves a person being unaware of processing the time during the interval, the latter requires a person to attend to and be aware that they need to time an event as they provide an overt estimate of the duration (Coull & Nobre, 2008; Droit-Volet, 2013; Zelaznik et al., 2002). Some research has found differing developmental patterns with implicit and explicit timing by comparing performance in adults and children of 5 and 8 years. Whereas implicit timing emerges early and may not vary with age, explicit timing emerges around the age of 3 years and does improve with age alongside the development of language and cognitive skills (Droit-Volet & Coull, 2016).

It has also been noted that temporal instructions given by adults are important to children's temporal responding before they begin to follow their own rules about time (Droit-Volet, 2013). In a relevant study, Pouthas and colleagues (1990) tested 18 children aged 4.5, 7, and 11 years on a timing task in which reinforcement was contingent on a button press duration (5s). Participants were also asked questions about the contingency. Accurate responses by 11-years-old closely matched their verbalized strategies (e.g., counting the interval), while accurate responses of the two younger age groups were not coherent with their verbalizations (i.e., descriptions of the temporal aspects of the contingency were mostly absent). Droit-Volet (2013) highlighted a possible shift towards rule-control when referencing a u-shaped developmental trajectory seen with temporally-regulated behavior in experiments using FI schedules of reinforcement. Infants show the expected pattern with FI schedules, more erratic patterns are seen in young children, and then there is a re-emergence of a

temporally-regulated response pattern in 7-8-year-olds (Bentall & Lowe, 1987; Bentall et al., 1985). By the age of 7-8 years however children's behavior in experimental tasks is due to rules about the interval, not the direct contingency.

Research has also shown that duration estimates can be distorted depending on contextual variables (Droit-Volet & Gil, 2009). For instance, when attention is diverted away from the duration of an event, intervals may be judged shorter while longer estimations of intervals may be made when specifically attending to the passage of time (Block et al., 2010; Coull et al., 2004; Zakay & Block, 1996). Other research in adults and children has shown that different "emotional stimuli" (e.g., pictures of facial expressions depicting high/low arousal or positive/negative valence) can influence estimates of duration (Angrilli et al., 1997; Droit-Volet & Meck, 2007; Gable & Poole, 2012). Gil et al. (2007) showed that on a temporal bisection task with pictures of angry and neutral faces, 3-, 5- and 8-years-olds estimated the duration of angry faces to be longer than that of neutral faces. In another study with university students, Fayolle et al. (2015) found that when a stimulus was paired with a mild shock during a temporal bisection task, participants judged the durations to be longer than trials with no shock. More accurate duration estimates have also been observed when participants are made aware of attention- or emotion-related time distortion (Droit-Volet, Lamotte, et al., 2015; Lamotte et al., 2012).

## **RFT** Extension

Some studies suggest that rules are relevant to children's timing responses (e.g., Droit-Volet, 1998; Pouthas et al., 1990). RFT sees relational framing as central to rule-following and thus also to timing tasks that involve experimenter delivered verbal instructions or self-verbalized strategies. Attending to the temporal features of the timing task has been highlighted as important to performance (Block et al., 2010; Lamotte et al., 2012). In typical timing tasks, providing verbal instructions plays a critical role in bringing a child's behavior under the control of the temporal features of the task. Droit-Volet (2013) stated that as children become able to understand instructions used in experimental timing tasks around the age of 3 years, the methods used shift towards those also used with adults. RFT might help provide further precision around specifically which relational framing skills are necessary to enable children to respond on such tasks as well as how performance on such tasks might be improved.

As one example, consider a novel instruction used within a verbal production task of timing abilities such as, "Say start and then stop when one minute has passed" (Block et al., 1999). To accurately follow this instruction, the child must derive a temporal relation under control of the contextual cue "then" within the instruction (i.e., saying "start" comes before saying "stop"). This temporal relation is important to the sequence or order in which the child's vocal behavior occurs in this example. However, this temporal relation (saying "start" then "stop") is also made more specific by speaking of the stimulus event as having an end and adding quantification (one minute) to the relational context. A child with a longer and more extensive history of reinforcement with respect to

equivalence relations between duration words (e.g., 1 minute, 10 minutes, 1 hour, etc.) and the corresponding amounts of time passed would likely perform better on this task. As a child learns to verbally relate duration words with passages of time, a gradual improvement of explicit timing abilities would be expected as this may coincide with more precise rule-control.

Learning to arbitrarily relate words as equivalent to quantities (involved in learning to count) and applying this to the temporal dimension of stimulus events allows for more precise relations to be derived. This may be especially relevant to explicit or verbal timing. For instance, utilizing counting strategies to keep track of the passage of time becomes advantageous as our temporal estimates are not always reliable (Droit-Volet, 2011). Research shows an increasing shift as children age towards using counting or quantification strategies (Espinosa-Fernandez et al., 2004; Levin & Wilkening, 1989; Wilkening et al., 1987). RFT suggests that as a child's AARR repertoire improves, there is an increased opportunity to engage in verbal strategies to time events and to act more effectively in one's environment. Rules to withhold behavior ("Wait until X amount of time has passed,") or persist in behavior ("Keep going for X amount of time") may be more functionally useful when arbitrary quantitative relations are accurately applied to the rule.

Research could examine at what ages certain relational frames are typically acquired and how this impacts timing behavior. At present, RFT research on temporal relations has been limited to investigations of derived relational responding with "before/after" and "now/then" cues or their functional equivalents. Deriving relations in accordance with "before" and "after" cues allows an individual to order events or sequence their own behavior (O'Hora et al., 2004); however this only enables a person to respond appropriately to event order, not to event duration. Applying quantification to durations of events can provide further specification of derived relations, which may be advantageous to timing behavior. As yet however, no RFT work has examined temporal relations that have been further specified with the quantification of durations and what effect this has on timing behavior. One direction for research might be to investigate whether training in equivalence relations between quantities and durations in children alters observed performance on timing tasks.

Another potential focus of RFT research relevant to timing might be to examine contextual conditions that influence duration estimation and passage of time judgments. Duration estimations can differ depending on the context, such as with emotion-stimuli (e.g., Angrilli et al., 1997). RFT would suggest that the direct and relational learning histories of a person are relevant, as these would establish particular functions of such stimuli, emotional or otherwise. Research could investigate how the functional properties of stimuli influence how quickly or slowly time seems to pass and importantly, whether transforming the existing stimulus functions through derived relations changes duration estimations. Additionally, some recent work has begun exploring how duration estimations are affected during mindfulness activities (Droit-Volet, Fanget, et al., 2015; Kramer et al., 2013; Wittmann & Schmidt, 2014). Future RFT research could examine such phenomena in more functional behavioral terms. For example, through temporal frames and AARR more generally, verbally

constructed past or future events could be imagined which might affect attendance to present moment contingencies. It has been stated that mindfulness activities aim to strengthen observing responses toward nonarbitrary features of the current environment while reducing the dominance of AARR about past or future events (Belisle & Dixon, 2021; Tarbox, Szabo, et al., 2020). It might be useful to use this functional conception to examine the effect of relevant experimental analogs of such processes on the subjective experience of time.

## Acquisition of the Clock and Calendar Systems

### Summary

Another key repertoire important to an understanding of time as an abstract verbal concept is using the clock and calendar systems. These conventional systems are used pervasively in societal organization and are advantageous tools to adaptive functioning and independence. An adult-like use of these systems requires an understanding of both temporal order and temporal duration. Other skills important for appreciation and use of these conventional systems include familiarity with repeating sequences and temporal perspective.

Mastery of the clock and of calendar systems are acquired gradually during childhood (McCormack & Hoerl, 2017; 2020; Thornton & Vukelich, 1988). Friedman and Laycock (1989) showed that children are able to tell time on digital clocks and to tell the whole hour on analog clocks by 6-7 years, to tell half-hour time on analog clocks by 7-8 years, and to tell the time to the minute by 9 years, albeit errors were still observed in some children up 10-11 years. The researchers also showed that the skill of transforming the time on a clock (e.g., correctly answering the question, "What time would it say in 30 min?") tended to emerge by the upper-elementary school years. With other time units, children 6-8 years can order the days and seasons but have difficulty with the cyclical aspect of time, such as that fall-winter-spring-summer and spring-summer-fall-winter are both correct sequences (Friedman, 1977, 1982). Between 8-10 years, ordering months, sequencing from different starting points, and referencing the calendar when describing the nearness of past/future events improves (Friedman, 2000; Friedman et al., 1995).

Tartas (2001) tested 200 children between 4-10 years using an interview task involving "when" questions (e.g., "When do you go for break at school?" "When does Christmas happen?" etc.) and a card ordering task about daily, weekly, and annual events. Results showed that as children age, there was a gradual shift from answering with relative temporal locations (e.g., "First we go to the toilet and then we go for break") to more absolute locations in their responses (e.g., "We go for break in the morning," or "We go for a break at 10:00 am"). By 5-6 years children used days of the week to order events, between 6-8 years they began to use hours to do so, and by 8-10 years they began to use months of the year.

Recently, Tillman and Barner (2015) found that understanding of the relative order of duration words compared to each other (e.g., "day" > "hour" > "minute") emerges and gradually improves between the ages of 4-6 years. However, an understanding the absolute durations of these

Chapter 2

words and their proportion to each other (e.g., how much longer an hour is than a minute) did not occur until later in childhood. The 6-7-year-olds who had learned the formal definitions of the duration words (e.g. that 1 hour = 60 minutes) performed better than those who did not on a task to represent their durations on a timeline. Tillman et al. (2017) observed a similar trajectory with the acquisition of deictic terms, such as "yesterday" and "tomorrow" using a timeline-based task. Between the ages of 4-6 years, children gradually improved in their understanding of deictic status (e.g., "last week" is in the past, "tomorrow" is in the future) and relative order (e.g., "tomorrow" comes before "next week"). It was not until after 7 years however, that an understanding of remoteness from the present (e.g., how much further "next week" is than "tomorrow") emerged. These authors also indicated that these latter skills develop around the time clock reading is emphasized in school and that explicit instruction may facilitate these repertoires.

## **RFT Extension**

Acquisition of conventional time concepts can take years to be fully mastered. The clock and calendar systems are socially-constructed arbitrary systems that require explicit instruction (Nelson, 1996b). As such, future research might investigate whether difficulties with time concepts in these conventional systems correlate with deficits in AARR and whether training derived relations when learning conventional time units offers benefit to educators and practitioners. Research could compare using an RFT approach focusing on systematically training temporal AARR at various levels of complexity with existing educational approaches to determine whether the former might lead to faster acquisition of key clock and calendar terms. Research might also investigate whether an RFT approach can be successfully used to train these relatively complex skills at younger ages than previously observed or when deficits are present.

Using these conventional systems successfully requires a number of relational skills, not only temporal frames. Duration words are related with their equivalent lengths of time, and these words are also related to each other as well as to other non-duration words in various ways including comparatively, hierarchically, and deictically (see for example Table 2.2). An RFT approach would suggest that relating time words and conventional units in a variety of ways is critically important to a verbal understanding of time. Children often receive repeated exposure to some of these terms in educational settings such as by reciting the days of the week or months of a year during a calendar time. Friedman (1990) noted that although children can verbally list names of days and months for example early in childhood, it is not until a later age that they begin to use these terms more flexibly. A young child may be said to "know" the days of the week, but this does little to evaluate in what contexts they are able to use these terms relationally. RFT provides a more systematic way to evaluate the understanding and use of these terms by emphasizing instruction on key relational patterns involved, use of MET to train these patterns, and testing of relevant derivation and transformation of function.

While conventional time units can be related in a variety of ways, certain mathematical skills are also clearly needed for mastery of the clock and calendar systems. On one level, acquiring basic equivalence relations between durations and arbitrary words is useful (e.g., 1 week is 7 days), but this is also likely insufficient on a broader level. This relation for example does little to help a person derive subsequent equivalencies for other durations such as how many days 2, 3, or 4 weeks is. It is a quantified comparison relation (i.e., a week is 7 times more than a day) that enables a person to derive answers to these examples. Without such numerical relational skills, direct training would still be necessary to solve novel problems involving durations.

## Table 2.2

Example re	lational	patterns	with	time	units

Relational Pattern(s)	Examples
Coordination	An hour is the <i>same as</i> 60 minutes. 7 days is the <i>same as</i> 1 week. November is the <i>same as</i> "11."
Distinction	24 hours is <i>different than</i> 365 days, 12 months, and 1 year. Weekdays are <i>not</i> weekends. Leap years are <i>different than</i> other years.
Opposition	Morning is <i>opposite to</i> evening. Noon is <i>opposite to</i> midnight. 3:00 PM <i>is opposite to</i> 3:00 AM.
Comparison	A year is <i>more</i> time than a week. 30 seconds is <i>shorter than</i> 1 hour. An hour is <i>longer than</i> a second.
Temporal	April is <i>after</i> March. Spring is <i>before</i> summer. 11:30 AM is <i>earlier than</i> 2:00 PM.
Hierarchical	Years <i>contain</i> months, months <i>contain days</i> , days <i>contain</i> hours. Halloween is <i>in</i> October. Winter is a <i>type of</i> season.
Deictic	<i>Today</i> is Thursday, <i>tomorrow</i> is Friday. <i>This</i> month is June, <i>last</i> month was May. <i>Now</i> it is the afternoon, <i>later</i> it is the evening.
Relational Network	If right <i>now</i> it <i>is</i> Christmas, New Year's is <i>sooner than</i> Valentine's Day. <i>Tomorrow</i> morning is <i>before next</i> week. <i>Last</i> month <i>was</i> February, so <i>this</i> month is <i>longer</i> .

Y. Barnes-Holmes et al. (2001) suggested that a history of AARR is needed prior to solving number-series problems commonly used with identifying numerical patterns, which shares similarity to deriving mathematical relations between time concepts. Developmental research has shown that comprehension of relative order between duration words (e.g., a year is more than a week) develops earlier in children than absolute duration (e.g., how much longer a year is than a week) (Tillman & Barner, 2015). Similarly, understanding relative order of deictic time words (yesterday happened after

last week) precedes remoteness from the present (e.g., how much further from the present last week or yesterday are) (Tillman et al., 2017). Although this work did not test for derived relations specifically, from an RFT view, the results are understandable as simpler instances of AARR would be needed prior to deriving relations that are made more precise with quantification.

Training derived relations might also be integrated with learning time estimation of events in younger children even before certain mathematical skills are established. One curricular activity used in educational settings is for students to match different activities of varying durations to their estimated length in time units such as, seconds, minutes, and hours. MET could be included with teaching how much time each of these words refer to. Multiple events that take around a few seconds, minutes, and hours relevant to the child's life might first be selected. The instructor would then explicitly train relations of coordination between the events and corresponding time units using standard reinforcement and error correction procedures. Once mastered, novel events that the child has not had previous training with or exposure to, might then be used for generalization and to test for derivation. The instructor could train a relation of coordination between a novel event (e.g., a bike ride to the store) and a previously trained event that takes a similar amount of time (e.g., eating dinner). The child could then be asked how much time it would take to get to the store and a derived relation would be demonstrated if the novel event is matched to the correct time unit. This could be extended beyond just matching as well, such as by using a timeline-based task representing a continuum of "less time" to "more time." Comparative relations between events could be trained and derived relations tested for. A transformation of function could then be assessed by having the child correctly sequence pictures of the events along the timeline.

## **Self-Control**

### Summary

Another key repertoire related to an understanding of and engagement with varying temporal durations is self-control. From a behavioral perspective, self-control has been understood as choicemaking behavior in which a person chooses a larger delayed reinforcer over a smaller immediate reinforcer (Schweitzer & Sulzer-Azaroff, 1988). Self-controlled responding is an adaptive skill as delayed outcomes are likely to be encountered in the natural environment and may also support social behaviors like sharing and cooperation (Stromer et al., 2000). Impulsive behavior has been described as the inverse: choosing the smaller-sooner reinforcer over the larger-later reinforcer.

In the behavior analytic literature, the presence of additional antecedent stimuli that signal delays to reinforcement has been suggested as a factor that may influence self-control responses. A few studies have found that adding signals, such as hand gestures, timers, or verbal instructions to wait may assist with reducing impulsive responding (Grey et al., 2009; Vollmer et al., 1999). Newquist et al. (2012) however, found conflicting results as rules and timers were insufficient for increasing self-control responses in three pre-school aged children. Whiting and Dixon (2015) highlighted a variety of behavior analytic approaches that have successfully been used to improve

self-control responses, such as progressive schedules of reinforcement (Schweitzer & Sulzer-Azaroff, 1988), presenting concurrent activities during the delay (Binder et al., 2000), introducing choice and an illusion of control (Dixon & Tibbetts, 2009), and integrating group contingencies (Dixon & Holcomb, 2000).

Developmental research has explored self-control responding in children with methods such as delay of gratification tasks (Mischel et al., 1989). In these tasks, a child is typically given the choice between receiving a smaller preferred item immediately (e.g., one sticker) or waiting for a period of time and receiving more of the preferred item (e.g., multiple stickers). Research with children has shown that the likelihood of choosing the delayed reward in this context correlates with age (Mahy et al., 2020; Miller et al., 1978; Mischel & Mischel, 1983; Mischel et al., 1989; Moore et al., 1998; Prencipe & Zelazo, 2005). Different explanations for these age-related differences have been suggested such as the development of inhibitory control and improvements in Theory of Mind skills that facilitate consideration of one's future desires relative to present desires (Mahy et al., 2020).

Researchers have also explored this topic in connection with the phenomenon of delay discounting, which refers to the tendency for individuals to devalue delayed rewards (Madden & Bickel, 2010). There is evidence that discounting rates are decreased on delay discounting tasks when participants are given a cue paired with the larger-delayed reward to imagine upcoming positive future events (Daniel et al., 2015; Daniel et al., 2013; Peters & Buchel, 2010). By verbally contacting the future events in a vivid manner (i.e., episodic future thinking), it is believed that the value of the delayed reward is altered and thereby influences choice-making (Benoit et al., 2011). For instance, Peters and Buchel (2010) had participants construct a list of upcoming planned events in their lives and then used these events as tags that appeared during a delay discounting task. That is, on trials to select smaller-immediate or larger-delayed rewards, statements such as "vacation Paris" or "birthday John" appeared on the screen indicating the event they had planned for that day. Data showed decreases in delay discounting rates in the experimental group that received these statements during the task. Other research has shown that the type of imagined future events has differential results on delay discounting rates. Liu et al. (2013) showed that imagining future preferred events (e.g., a party) showed decreased discounting rates, while aversive events (e.g., getting sick) increased discounting rates, and presumed neutral events (e.g., washing clothes) produced no difference compared to a control group.

Another strategy that has been shown to decrease discounting is having persons connect to and choose for their "future-self." Bartels and Urminsky (2011, p. 193) argued that when a person perceives their future-self as, "more closely connected to the current-self in terms of sharing important psychological properties, such as beliefs, values, and goals, the decision maker is more motivated to act patiently, furthering the later-self's welfare." Across multiple experiments, the authors demonstrated that instructing participants on continuity in identity between the self-now and

the self-later influences discounting. Other future-self studies that have seen reductions in discounting have used different methods, such as showing participants a virtual avatar depicting a future representation of themselves before completing a discounting task (Hershfield et al., 2011; Kuo et al., 2016). In their systematic review of behavioral trainings and manipulations to decrease delay discounting, Scholten and colleagues (2019) found that these types of future-oriented manipulations were among the most promising research directions.

# **RFT** Extension

Examining the role of temporal and other relational frames more directly may provide further insight into how self-control responding develops in childhood as well as language-based procedures for training such responses. Existing behavioral approaches to self-control, such as working with direct contingencies, may require extensive training. Further investigating verbal interventions to teach self-control may be worthwhile if it requires less time to reach similar outcomes or can augment other procedures (Whiting & Dixon, 2015).

Twohig et al. (2007) suggested that with a developed AARR repertoire, "self-control rules" such as, "if (behavior) then (better consequence) after (delay)," can be used by self or others. Following this type of rule necessitates at least three relations: temporal (a sequence of behavior, then X amount of time, then consequence), coordination (between words and events), and comparison (evaluating the immediate and delayed consequences). A person may understand a rule like this, though to follow the rule a transformation of function is also needed to provide the behaviorcontrolling properties to the rule (Harte et al., 2020). In this case, transformation of function via temporal and coordinate relations allows for the verbally constructed events to have stimulus functions (i.e., the likelihood of demonstrating a behavior in the moment may be influenced by temporally framing a preferred future event as coming after the behavior). The transformation of functions through comparison relations decreases the impact of the smaller-immediate consequences and increases the impact of the larger-delayed consequences. Accordingly, a sufficient reinforcement history for rule-following in accordance with these verbal relations is necessary for this type of rule to be effective (D. Barnes-Holmes et al., 2001). Little empirical work on establishing this type of rulegoverned behavior in children has been done, though some work has shown that autistic children can be trained in preliminary rule-following repertoires with antecedent-behavior (Tarbox et al., 2011) and behavior-consequence relations (Wymer et al., 2016).

Theoretically, once acquired, the capacity to respond in accordance with such self-control rules may then be useful at different points of time in influencing choices between smaller-sooner or larger-later consequences. Events can be put in relational networks that describe temporal sequences or behavioral contingencies *before* a person has the opportunity to respond. If a person with a sufficient relational repertoire is told ahead of time (or self-derives) that a larger amount of a preferred item comes after waiting, while a smaller amount comes after not waiting, these rules may transform the function of each choice and thereby alter the likelihood of engaging in the self-controlled

response. Comparing these networks in a visual format (e.g., behavior mapping) may help make the relations more salient, particularly for younger children (Figure 2.4). Secondly, some research on selfcontrol training has also shown that rule statements can be used in the form of mediational responses during delays to enhance self-control (Binder et al., 2000; Toner & Smith, 1977). For instance, children can be taught to quietly or covertly repeat a verbal response such as, "When I wait quietly, I get what I want" (Hanley et al., 2007, p. 284) to promote delay tolerance. Other concurrent activities can be used to a similar effect (Binder et al., 2000), though verbal behavior of this type may be advantageous for a child with a sufficient relational repertoire. Such behavior can be used in any type of situation, does not rely on the availability of additional materials, and opportunely directs focus towards the larger-delayed consequence. AARR might also be beneficial after the consequence for a choice has been experienced. Even if the immediate choice was still selected, an individual with an AARR repertoire can describe their choice of selecting the smaller option and relationally compare it to the larger-delayed option. In turn, the selected item may be transformed to have less reinforcing properties as a result of this comparative framing (D. Barnes-Holmes et al., 2001; Stewart et al., 2012). Engaging in AARR after the consequence is experienced and the resultant transformation of function may then serve as an additional source of influence to alter subsequent future responding. More simply, a person "reflects" on their choices after the fact and considers behaving differently in similar future contexts.

## Figure 2.4





Using an RFT-based approach to self-control training has been recommended as a future approach as it provides a technical way to account for the influence of verbal behavior on choice making (McKeel & Dixon, 2014). RFT suggests that as an individual increasingly acquires AARR, it

becomes possible to transform the function of the immediate and delayed consequences through language. Dixon and Holton (2009) showed that transforming the function of stimuli may affect how individuals respond to immediate and delayed reinforcers. In their experiment, five participants with a gambling disorder first completed a delay discounting task on a computer in which two irrelevant colored stimuli were presented alongside the smaller-immediate and larger-delayed options. Participants then completed a relational responding task in which the same two arbitrary colors were trained as contextual cues for "better than" and "worse than" relations with sets of quantitative gambling stimuli. Once relational training and testing was complete, participants again completed the delay discounting task that incorporated the arbitrary stimuli. Due to the relational history, the stimulus functions of the colored stimuli were now transformed in accordance with "better than" and "worse than" relations. Results showed that participants subsequently made more larger-delayed responses compared to baseline. This study provided initial evidence that impulsive responding may be altered by the transformation of stimulus functions in the absence of reinforcement for choosing self-control responses.

This RFT-based approach could be extended to other frames. It has been suggested that temporal relational frames are of special importance to delayed responding as, "learning to describe environmental sequences and delays may considerably reduce the interference of alternative events in accounting for delayed consequences" (Hayes, Gifford, et al., 2001, p. 99). For example, it might be investigated whether training children both directly on nonarbitrary duration estimations, as well as on derivation through temporal AARR of previously untrained durations, might alter delay-discounting behavior.

Research might also look at the methods used with the future-oriented manipulations from an RFT approach. In episodic future thinking manipulations (e.g., Peters & Buchel, 2010), temporal frames in particular may play a key role in enabling a person to verbally-construct hypothetical future events. From an RFT view, imagining future scenarios in "vivid" detail might be explained in terms of contextual control by the displayed phrases/instructions producing temporal AARR that allows momentary transformation of perceptual stimulus functions. As a result, the relative reinforcing value of the immediate and delayed options during the task are altered. In future-self manipulations, the "self" is being related in different ways that can alter discounting rates. Bartels and Urminsky (2011) for example, showed that the participants that read a short passage about how the person they are and the traits they have remain stable over time were more likely to select larger-delayed options with actual gift cards than participants that read a passage on how the person they are and the traits they have change over time. From an RFT view, this could involve relating relational networks in specific ways. Presenting relations of coordination between the deictic (interpersonal-temporal) relations of self-now and self-later may support self-control responding in contrast to presenting a relation of distinction. There has as yet been no RFT work that has modeled such processes in terms of transformation of function through temporal relations. Such research might support additional

Chapter 2

understanding of the key processes involved and lead to new practically oriented insight into how such exercises can improve choice making.

## **Planning and Time Management**

## Summary

Planning as a skill includes identifying a terminal goal and the steps or resources needed to reach the goal, initiating and monitoring progress, and completing the plan (Najdowski, 2017). In a review by McCormack and Atance (2011) on the development of planning in children, the authors suggested that planning involves three forms of flexible thinking. The first is representing time in an event-independent-manner. This refers to representing different points in a temporal sequence in a flexible way that allows for their reordering, including selecting which actions should occur at any point in the sequence. For example, doing action A before action B may lead to outcome C, which may be a different outcome than if action A is done after action B. Effective planning then, may involve thinking of not only which actions are needed to achieve a goal, but also in which order those actions should occur (McCormack & Hoerl, 2017). Secondly, executive function components, such as inhibitory control, are suggested as important, though they say that more research is warranted. Third, self-projection (or temporal perspective taking), refers to the ability to shift to alternative past or future perspectives that are not the here-and-now (Buckner & Carroll, 2007). McCormack and Atance suggest that these three skills can be used as a framework to examine the research on the development of children's performance on planning tasks.

A few different methods have been used to study planning in children. Tower tasks, such as the Tower of London (Shallice, 1982), involve moving objects around within a certain problem space from their initial position to a goal position in the least number of moves possible while adhering to certain rules. Planning skills are considered relevant to tower tasks since thinking of possible sequences of moves and their outcomes (without manipulating the objects) would presumably lead to better performance. Luciana and Nelson (1998) tested children between 4-8 years on problems at various difficulty levels (i.e., the minimum number of moves needed to complete the task) and found an increased ability and efficiency in solving these problems with age. Some research has indicated that these tasks are particularly difficult though for children under 5 years (Bull et al., 2004; Kaller et al., 2008).

Another approach is the route-planning task, in which children are given a model location (e.g., a store) and toy character, and need to plan the most efficient route through the location while completing the objectives (e.g., shopping for selected items). Gauvain and Rogoff (1989) compared performance between 5- and 9-year-olds on this task and found older children planned more in advance and produced more efficient routes. Simplified variations of this method have also been designed for use with younger children (Carlson et al., 2004; McColgan & McCormack, 2008). For instance, McColgan and McCormack (2008) tested 3-5-year-olds on a single trial planning task in which a doll visits a model zoo and wants to take a picture of a kangaroo. The children were tasked

Chapter 2

with identifying a location in which a camera can be stored so that the doll can pick it up along the path. Only the 5-year-olds reliably passed the task by selecting a location on the path that the doll would encounter before rather than after reaching the kangaroo.

The aforementioned approaches examine planning in the context of solving an in-the-moment problem. Other studies have however investigated children's planning for more distant or hypothetical future situations. In one study, Atance and Meltzoff (2005) presented 3-5-year-olds with stories and pictures of future events that would evoke a certain physiological state. The researchers asked the children to select an item they would need to take with them and to vocally explain the choice. For example, when a scene depicting a waterfall was shown, a correct response was to select the raincoat from an array with distracters and to provide an explanation such as, "I might get wet." Across all ages, children tended to select the correct item while 4- and 5-year-olds. In a follow up experiment, the researchers replaced one of the distracters with an item that was semantically associated with the picture scene (e.g., a picture of rocks for the waterfall scene). Performance in 3- and 4-year-olds was significantly affected as there was an increased tendency to select the semantically associated item.

A related skill set, time management, involves: a) scheduling activities based on duration estimations of each activity, b) checking the time during the activity to determine if one is on or off track with the schedule, and c) continuing or adjusting one's behavior or the schedule to be on time for subsequent scheduled obligations (Najdowski, 2017). Said another way, a person engages in selfmanagement of their own behavior within a given time frame (Barkley, 2013; Claessens et al., 2007). Conceptually, time management could be approached as a broader repertoire that involves the previously described skills of sequencing, duration estimation, use of conventional time systems, selfcontrol, and planning. A few prerequisite skills to time management that have been noted include telling time, making and following a schedule, and estimating the duration of activities (Dawson & Guare, 2010). Between 8-11 years children perform tasks such as planning simple school projects, keeping track of changing daily schedules, and may plan for how to earn money. By 11-14 years, children are tasked with following complex and changing school schedules, planning and completing long-term projects, estimating time needed to complete tasks, and adjusting their schedule (Dawson & Guare, 2009).

Presently, behavior analytic research investigating the training of these repertoires is scarce. Multiple curricular or educational resources have been developed however, that aim to improve planning and/or time management, among other skills (Cannon et al., 2021; Dawson & Guare, 2010; DiPipi-Hoy & Steere, 2016; Najdowski, 2017). For instance, Najdowski (2017) suggests that practitioners can utilize standard behavior analytic principles such as reinforcement, prompt fading, chaining, and MET to train these repertoires. The teaching approach suggested for planning involves training learners to first identify a short-term goal (e.g., clean the bedroom) or long-term goal (e.g., school project), identify the needed materials, task analyze the relevant steps, and then initiate,

monitor, and complete the plan. For time management, learners are taught to identify and predict how much time activities take, and then make and follow a schedule. For each of these repertoires, different activities and routines are used as exemplars to promote generalized responding as opposed to only learning a specific task.

## **RFT** Extension

Mainstream psychological approaches to skills associated with planning and time management are often examined in context of executive function (Barkley, 2011, 2012). Executive function refers to "self-directed actions needed to choose goals and to create, enact, and sustain actions toward those goals" (Barkley, 2012, p. 60). Some authors have conceptualized that executive function could instead be approached as a subset of rule-governed behavior by examining the conditions in which people, "select among available rules or generate new ones, follow rules when they are available even though they conflict with other sources of behavioral control, and change them when they no longer work" (Hayes et al., 1996, pp. 292-293). One RFT-based observational study has explored patterns of relational framing that may occur during a common task of executive function (Wisconsin Card Sorting Test; WCST). Tyrberg et al. (2021) presented the WCST to adult participants and asked them questions during the task about their responses (e.g., "What do you think you can do now?" "Why do you think that was correct/incorrect?"). The authors recorded responses and analyzed transcripts by identifying relational patterns based on the types of contextual cues emitted by participants. The authors reported that participants appeared to engage in deictic and temporal framing during critical points of executive function activity (category shifts during the task). Continued research in this domain may lead to further clarity on conceptualizing executive function related abilities in behavioral terms and whether targeting temporal and/or deictic framing for intervention leads to changes in performance on measures like the WCST.

Regarding the literature on planning in children, two areas are suggested for future investigation. One, McCormack and Atance (2011) note that research is lacking on the relation between children's understanding of "before/after" and their planning abilities. For instance, routeplanning tasks (McColgan & McCormack, 2008) involve manipulating events in before-after sequences, so an accurate use of these terms may be of benefit to successfully pass these tasks. RFT research might address this limitation by training these words as contextual cues for temporal framing and evaluating whether this influences performance on such planning tasks. Similarly, performance on temporal framing tasks could be compared to scores on future-thinking measures, such as the Children's Future Thinking Questionnaire (CFTQ) which measures the future-oriented skills of planning, prospective memory, episodic future thinking, saving behavior, and delay of gratification (Mazachowsky & Mahy, 2020). Secondly, some experimental tasks in the cognitive developmental literature investigate planning to obtain a goal in the present environment (Carlson et al., 2004), while others investigate planning for a hypothetical future goal (Atance & Meltzoff, 2005). RFT research

might look to investigate the types of relational repertoires needed for each of these contexts and more specifically to what extent temporal framing plays a role.

Conceptually, temporal or causal relations are important in rules and this is particularly true for rules around planning and time management. An advantage of planning actions ahead of their occurrence is that it allows for individuals to verbally contact contingencies that are not in the immediate environment or have not previously been directly experienced. By verbally constructing multiple contingencies, a person can then select the action(s) for the present moment that will best achieve the intended outcome(s). Relationally, this type of verbal problem-solving is made possible through temporal and comparison (evaluative) frames (Hayes, Gifford, et al., 2001; Hayes et al., 2007). Temporal framing puts behaviors and consequences into particular sequences, and these sequences can then be related to alternative sequences through comparison framing. By relationally responding in this manner, a person can then select the most useful response(s) relevant to the context. Engaging in further relational responding, (e.g., a "pros and cons" activity), could abstract additional features of the context that may help with effective planning and decision making (Hayes et al., 1996).

Adaptive rules can be used to plan and manage time in other ways. For instance, one aspect to managing time includes prioritizing activities and deciding at which points in time each should be completed (DiPipi-Hoy & Steere, 2016). By prioritizing, a temporal order of activities can be constructed. For example, one well-known strategy is to sort activities based on their urgency and importance (i.e., the Eisenhower matrix). Tasks that are sorted as both "urgent" and "important," are selected to be completed sooner than the activities that are sorted elsewhere. AARR is key to successfully using a rule-based strategy such as this. Activities are related in frames of coordination or distinction with the verbal relata "urgent" and "important." Sorting based on "urgency" also involves framing how soon tasks need to be completed relative to now or another point in time (deictic, temporal) and therefore which events are more or less urgent (comparison). The relative "importance" of a given activity is context-dependent and involves relating events comparatively to other events.

A potential challenge in managing one's time is that preferred activities may interfere with activities that are neutral or aversive. Temporal framing may be useful in this instance, as it allows the person to verbally contact the differential consequences of completing each task. The person can think through what possible outcomes could happen in the future if each activity is completed (or not completed) sooner or later. This may help to transform how "important" a given activity is since neutral or nonpreferred tasks are then framed as coming before a delayed, but more preferred consequence (e.g., "If I can finish [aversive activity] today, then later this week I will have more time for [preferred activity]). Hierarchical framing with values-based strategies may also help to form adaptive rules for managing various activities in a time frame. In the RFT literature, a value has been defined as the highest point in a hierarchical relational network of life activities (Barnes-Holmes, Barnes-Holmes, McHugh, et al., 2004). Acting in accordance with one's values then involves the

transformation of consequential functions through this hierarchical relation (Plumb et al., 2009). Performing certain less-preferred tasks can be hierarchically related as belonging to a particular value (e.g., "be persistent" or "responsible"). Framing in this way may lead to transforming the function of the lesser preferred tasks to become "more important" to the person as they become instances of a verbally constructed reinforcer.

However, time management requires more than simply creating or responding to a sequence of tasks in a particular temporal order (i.e., making and following a schedule). Relational responses such as relating durations as shorter or longer than others, following rules that include quantification, and deriving whether one's progress over specified durations is sufficient to stay on schedule are likely key to effectively managing one's behavior relative to time. RFT research could examine to what extent AARR repertoires, including especially various levels of temporal framing, are helpful for training planning and time management and in what ages these skills might best be taught.

In applied contexts, one way that a focus on AARR can be integrated within existing behavioral training approaches for planning and time management is with a "leading questions" prompting method (Najdowski, 2017). Najdowski suggests that instead of simply providing the learner a prompt for the answer during a task, leading questions are models for the kind of verbal behavior in which learners might engage while carrying out the steps of a task. Leading questions for planning might include questions like, "What are all the steps your plan includes?" "In what order will you do them?" or "What might happen if you did step X before step Y?" For time management, leading questions could involve asking, "How much time is that activity?" "How long did a similar activity take last time?" or "Which activity takes a longer amount of time to do?" (Table 2.3). From an RFT perspective, these leading questions help to teach a learner to engage in certain patterns of AARR before, during, or after they complete different steps of a task. In these examples, hierarchical relations help to identify possible behaviors involved in a complex task. Temporal relations bring a specific sequence to those behaviors and help to identify alternative consequences based on the order. Coordination relations help the learner to accurately estimate needed time, and comparison relations help the learner to prioritize order when thinking about which activities take more or less time. These prompts can then be faded to facilitate the learner's engagement in similar relational responses in the presence of the natural discriminative stimuli alone. Future research could investigate whether training learners to engage in particular relational framing patterns at different points during complex tasks supports acquisition of generalized planning and time management repertoires.

Table 2.3

Skill Area	Example Verbal Prompts	Relation
Planning	"What are all the steps your plan includes? "In what order will you do these steps?" "What might happen if you do step X before step Y?"	Hierarchical Temporal Temporal
Time Management	"How much time is that activity?" "Which activity takes a longer amount of time to do?" "How long did a similar activity take last time?"	Coordination Comparison Coordination/deictic

Example Planning and Time Management Relational Responses

### Conclusion

In summary, behavioral research has historically investigated human responses to time through respondent conditioning and operant conditioning paradigms, such as with performance on reinforcement schedules. Although this work provides some insight, behavioral approaches to human responding to time as an abstract verbal concept has largely been absent in the literature. In this chapter, RFT was introduced an approach that could be used to extend work in this area. Relational framing, and particularly temporal relational framing, was suggested as key to abstract or verbal time. A conceptual overview of temporal framing was provided and the present state of RFT research into this skill was discussed. Research shows that temporal relations are relevant to sequenced responding in the context of rule-based control, though empirical work is currently limited to typically developing adults with established relational repertoires. A range of important time-related repertoires that have been explored from various psychological approaches was then reviewed. Particularly, the discussion focused on the areas of temporal sequencing, timing, acquisition of the clock and calendar systems, self-control, planning and time management. The discussion also explored how an RFT approach focused in particular on temporal relational responding might amplify and extend the existing research in each of these areas, with a particular emphasis on potential improvements in training and supporting such repertoires in children.

With the advancements of RFT research on language, rules, and derived stimulus relations, it is possible to extend a functional behavioral approach into the key domain of a human understanding of time. RFT sees derived relational responding, and notably temporal framing, as critical in this respect. Investigating how this repertoire develops as well as how it may best be trained and supported may offer new insight as well as many practical applied advances.

Chapter 3.

Assessing Temporal Relational Responding in Children

Portions of this chapter have been accepted for publication:

Neufeld, J., Stewart, I., & McElwee, J. (2023). Assessing temporal relational responding in young children. *The Psychological Record*. <u>https://doi.org/10.1007/s40732-023-00534-4</u>

Acquiring an understanding of how events are temporally related to each other, both with respect to actual experienced events as well as with respect to events in a more abstract sense (e.g., as represented in a calendar) is critically important in young children's development, allowing them to communicate more precisely about events in their world and to advantageously organize their behavior in time. Examining how temporal understanding is acquired in childhood may ultimately help inform practical intervention to teach such skills when they are deficient. As discussed in Chapter 2, an RFT approach could be used to investigate several temporal skill sets. The focus of the present chapter is to explore one of these areas and provide a starting point for additional research. Arguably, a fitting starting point is to investigate performance on temporal NARR and AARR tasks across a variety of ages with contextual cues such as "before" and "after."

Research into children's temporal understanding has heretofore mainly stemmed from cognitive developmental and psycholinguistic perspectives. Clark (1971) examined comprehension of "before" and "after" in 3- to 5-year-old children by requiring participants to perform sequences of actions with toys when instructions using these terms were given. Findings indicated that comprehension of both terms improved with age and that based on error patterns, children appeared to acquire an understanding of "before" earlier than "after." Clark (1971) reported that when unsure of the meaning of the temporal term, the children appeared to use an order-of-mention strategy— responding as if the order of events matched the order of mention in the sentence. Using an order-of-mention strategy would suggest that less errors would likely occur with sentences in which the actual order of events and the order in which they are mentioned match (e.g., "The girl kicked the ball before she patted the dog," or "After the girl kicked the ball, she patted the dog") compared to sentences in which the event order and the order of mention do not match ("Before the girl patted the dog, she kicked the ball," or "The girl patted the dog after she kicked the ball").

More recent research has extended findings on "before" and "after" comprehension in children. McCormack and Hanley (2011) showed 3- to 5-year-olds pairs of video clips of two noncausal actions in a particular order (e.g., brushing teeth, then washing face) and the experimenter said a sentence using either "before" or "after" describing the sequence. Participants were then asked to point to the video pair on the screen that either matched or not with the sentence they had been given. Children performed better with matching-sentences and only 5-year-olds performed above chance levels with non-matching sentences. Blything et al. (2015) sought to determine the age at which children begin to shift towards relying on "before" or "after" terms themselves as cues for temporal order instead of relying on background knowledge of events or an order of mention strategy. They showed young children sequences of an animated character performing two actions while the experimenters manipulated the sentence structure that narrated the order of the actions. Children were then asked to select the picture of the activity that came first. Results suggested that 3- to 4-year-olds performed better on "before" sentences compared to "after," while sentence variation and performance on a test for working memory (digit-span task) predicted performance in 4- and 6-years-

Chapter 3

olds. The authors noted the difficulty of discriminating effects of memory from those of language development.

The production of temporal terms has been shown to appear as young as 2 years, while the comprehension and accurate use of these terms gradually improves over childhood (Busby Grant & Suddendorf, 2011; French & Nelson, 1985). Errors in comprehension of these terms in particular contexts have been demonstrated up to early adolescence. For instance, Pyykkönen and Järvikivi (2012) tested understanding of multiple events in children between the ages of 8 and 12 years. Children were given a questionnaire on order of events that included a variety of sentences depicting simultaneous or sequential events while the temporal terms' sentence placement varied across trials. Results demonstrated that the children's accuracy on comprehension questions about the order of events was affected by the position of the temporal terms "before" and "after" within the sentences. While performance generally increased with age, even the 12-year-old children did not perform to levels of accuracy shown by an adult comparison group.

Some developmental theorists have pointed to the acquisition of language as critical in a child's acquisition of temporal understanding. For example, Nelson (1996a, 2007a, 2007b) has argued that it is through verbal interactions, such as by sharing past and future experiences or through direct instruction of conventional time systems, that children learn how to temporally organize events and reference themselves in time. Moore et al. (2014) evaluated Nelson's interpretation of the relation between time concepts and language and intellectual skills in an aspect of their study with children between 5 and 10 years of age. Participants were tested on time concepts and their performance compared with that on standardized measures of verbal and intellectual abilities (PPVT-4, TONI-3). A variety of tasks was used to evaluate temporal event ordering abilities such as placing cards depicting events on a spatial layout indicating how near or far they were from the present, labeling time concepts, sequencing picture cards of events in forward and backward order, and answering questions about the relative order of related events. Results showed that performance on the temporal cognition tasks were significantly predicted by verbal abilities, thus providing empirical support for a relationship between language ability and temporal understanding.

Critically, a limitation of the current literature is that minimal research has been done investigating the environmental variables that influence the acquisition of temporal understanding. While some research has shown that feedback and other instructional arrangements can improve understanding of temporal order in sentences (Amidon & Carey, 1972; Ehri & Galanis, 1980), studies are largely descriptive in nature. Turning towards a behavioral approach may assist with addressing this limitation on training these skills due to an emphasis on both prediction and influence of behavior (Biglan & Hayes, 1996). One particularly relevant approach in this regard is Relational Frame Theory (RFT; Hayes, Barnes-Holmes, et al., 2001; Stewart, 2016), a contextual behavior analytic account of human language and cognition. Substantial empirical research now attests to the efficacy of RFT as a basis for teaching basic and complex language skills to children and adults with and without

Chapter 3

disabilities (Gilroy et al., 2015; Kirsten et al., 2022a; Ming et al., 2018; Newsome et al., 2014; M. O'Connor et al., 2017; Tarbox et al., 2011; Wymer et al., 2016).

RFT suggests that arbitrarily applicable relational responding (AARR) is a key generalized operant underpinning human language and cognition. Relational responding refers to responding to one stimulus in terms of its relation to another stimulus. Nonarbitrary relational responding (NARR) is based on the formal properties of the stimuli (such as selecting a stimulus because it looks larger than another stimulus) and has been seen in many nonhuman species as well as in humans (Giurfa et al., 2001; Harmon et al., 1982; Ming et al., 2018). AARR, in contrast, is based not on the formal properties of the stimuli being related but instead on contextual cues that specify the relation irrespective of formal properties; for example, if I am told that coin A is worth 'more than' coin B then I will treat coin A as larger in value and preferable to coin B even though they are similar in size. AARR has thus far been shown unequivocally only by humans (Dugdale & Lowe, 2000; Hayes, 1989; Lipkens et al., 1993). RFT suggests that humans show a variety of different patterns of AARR (frames) including coordination (sameness), distinction (difference), opposition, comparison (more/less), spatial relations (e.g., above/below), temporality (e.g., before/after), hierarchy (e.g., member of/class of), and deixis (Hayes, Fox, et al., 2001). Framing allows the child to derive novel untaught relations between events (e.g., if told that A is more than B and B is more than C, I could derive that A is more than C and C is less than A) and hence is key to acquiring generative language (Stewart, McElwee, et al., 2013). Critically from both a theoretical as well as a practical standpoint, RFT studies have provided substantial evidence that AARR is closely linked with language and cognitive abilities (Barnes et al., 1990; Barnes-Holmes et al., 2005; Cassidy et al., 2016; Cassidy et al., 2011; Colbert et al., 2018; Devany et al., 1986; Dickins et al., 2001; Hayes & Stewart, 2016) and that it is possible to establish a variety of different patterns of AARR or frames (Barron et al., 2019; Belisle, Stanley, et al., 2020; Berens & Hayes, 2007; Dunne et al., 2014; Kirsten et al., 2022b; Mulhern et al., 2018; Rehfeldt & Root, 2005).

While there is a variety of types of relational framing, all frames share three core properties. Mutual entailment refers to a bidirectional relation between two stimuli, such that if a participant is taught that A is before B, then they derive that B is after A. Combinatorial entailment refers to the combining of relations, allowing for new derivations. For example, if a participant is taught that A is before B and that B is before C, then they may derive that A is before C and that C is after A. The third property of relational framing is the transformation of function, in which the psychological function of a stimulus participating in a derived relation is changed or transformed based on the other stimuli to which it is related as well as the nature of the relation. For instance, if stimulus event B has an appetitive function and is derived as being after event A, then the functions of A may be transformed such that anticipatory responding occurs during this event.
From an RFT perspective, children acquire particular patterns of AARR through multiple exemplar training (MET), which involves exposure to a variety of exemplars in which the stimuli and/or situations differ but in which the specific relational cues for that pattern remain the same. Through a history of such reinforcement from their socioverbal community, children abstract the relevant relational cues that indicate how stimuli are to be related (Hughes & Barnes-Holmes, 2016) and thus acquire generalized patterns of AARR.

In addition, while NARR and AARR can be described as distinct classes of behavior, acquisition of AARR is typically preceded by and supported by the acquisition of NARR (see Stewart & McElwee, 2009). In other words, the child might first learn to relationally respond in the presence of particular contextual cues based primarily on nonarbitrary properties of the stimuli involved but with sufficient exemplars, their relational response can come to be applied to arbitrary stimuli irrespective of the physical properties involved. Indeed, there is now empirical evidence in the case of a number of frames showing that assessing and/or training NARR can promote acquisition of AARR (Berens & Hayes, 2007; Zygmont et al., 1992).

With respect to temporal AARR, as in the case of other relational frames, RFT suggests that this repertoire is established through MET and also that children learn temporal NARR before learning AARR. More specifically, RFT suggests that children likely first learn to relationally respond with respect to events actually experienced in time (NARR) and subsequently learn to derive temporal relations between events without having to actually experience them; for example, if given two hypothetical events A and B, and told that A is after B, they could derive that B is before A (AARR; Hughes & Barnes-Holmes, 2016).

With regard to RFT research on temporal relational framing, O'Hora and colleagues (2004) provided the first relevant study, which was focused on relational responding in the context of instructional control. They trained and tested 12 university students on a computer-based relational responding protocol that involved "before/after" and "same/different" relations. Two arbitrary symbols were trained as contextual cues for "before" and "after" by reinforcing participants' selection of a statement that included one of the arbitrary symbols that accurately described an observed sequence of shapes. Participants were subsequently tested for instructional control on a task that involved pressing computer keys in a particular order specified by the arbitrary relational networks involving the "before/after" and "same/different" cues. Eight of the twelve participants showed responding in accordance with the arbitrary relational networks and participants who demonstrated instructional control in a second experiment additionally showed generalized responding to novel stimulus sets. A set of follow up studies (O'Hora et al., 2005, 2008) extended this work by showing that performance on the relational responding protocol correlated with performance on standardized measures of intellectual ability in university students. Other research relevant to temporal relational responding has focused on comparing accuracy and response latencies to arbitrary "before" and "after" cues on tasks of nonarbitrary sequenced responding. Overall, studies have shown that adults

respond slower on "after" trials than on "before" trials (Hyland et al., 2012; 2014) and with less accuracy (Brassil et al., 2019; McGreal et al., 2016).

Existing RFT studies on temporal relational responding have been limited to investigation with typically developing adults with already developed relational framing repertoires. Research on the acquisition of temporal relations in young children without such advanced repertoires on the other hand is almost non-existent. Thus far, just one study has looked at temporal framing in children. Kirsten and Stewart (2022) examined a number of different varieties of relational framing in a cross section of young children between 3-7 years and in doing so provided some evidence that temporal framing may emerge later in development than other types of framing. However, one of the limitations of this study was that, because it involved surveying performance of young children on a number of different frames, the scope of testing in the case of particular types of framing was very limited. As such, these data on temporal framing in young children are perhaps not quite as informative as they might otherwise be.

Despite the limitations of past research on temporal framing, however, this work has provided at least a starting point for further exploration, such as by evaluating temporal relations in different populations. The aim of the present study is to investigate performance on tasks of temporal relational responding in young, typically developing children between the ages of 3 and 8 years. The present study utilized an RFT-based protocol to assess temporal relational responding at increasing levels of complexity. This involved testing children on tasks of nonarbitrary temporal relations and arbitrary temporal relations, including assessing for properties of mutual and combinatorial entailment, and transformation of function. We intended that this exploratory study would be an initial step that might help facilitate future research to examine the acquisition of temporal relational responding in more detail and to design appropriately targeted training procedures if this repertoire was found to be deficient based on the level of performance typically observed at a given age.

The present study, which was conducted online, utilized adaptations of existing RFT methods to assess temporal relations. The relational evaluation procedure (REP) is an effective and efficient method to assess various patterns of relational framing (Cassidy et al., 2011, 2016). The REP involves presenting a particular contextual cue (e.g., SAME) in between two or more arbitrary stimuli on the screen (e.g., "QUB is the SAME as TIG") and asking participants questions about the presented relation (e.g., Is TIG the SAME as QUB?"). Kirsten and Stewart (2022) recently adapted the REP for younger children who were not yet readers or had minimal reading skills by using colored shapes instead of nonsense words and single letters as contextual cues instead of full words, alongside audio options (e.g., [Red circle] [S] [Blue circle] to indicate that "red is the same as blue."). A similar format was used in the present study as the age of participants was comparable. Additionally, the present study adapted the test for instructional control used by O'Hora and colleagues (2004) to test for a transformation of function effect in accordance with temporal relations. Instead of using computer keyboard presses in an order specified by a relational network, observable and

discriminable motor actions were included. We anticipated this as an advantageous modification given the ages of participants and as the study was conducted online.

#### Method

### **Participants**

Twenty-five typically developing children (13 female, 12 male) between the ages of 3 and 8 years (M = 5 years 6 months, SD = 17.87 months) participated in the study. These included five 3-year-olds (M = 42.49 months, SD = 3.63 months), six 4-year-olds-olds (M = 56.26 months, SD = 3.77 months), four 5-year-olds (M = 66.35 months, SD = 1.56 months), four 6-year-olds (M = 77 months, SD = 3.82 months), five 7-year-olds (M = 87.92 months, SD = 3.23 months), and one 8-year-old (100.5 months). All participants were recruited through personal contacts in the United States and Ireland. Ethical approval for the study was first obtained through the institutional research ethics committee prior to conducting the study. Verbal and written consent was obtained from each participants' caregivers and verbal assent was obtained from the children themselves before starting a session.

### **Setting and Materials**

All sessions took place online through the use of a video conferencing platform with end-toend encryption. Sessions were conducted at times preferred by each family and with the child and an accompanying family member participating via a networked device in a suitable room in their living residence. All materials were created in Microsoft PowerPoint and displayed on a 13-inch MacBook Pro laptop via the screen-share feature of the video-conferencing platform.

Each child was tested on an initial preassessment and a temporal relations assessment (TRA). The preassessment was used to test the child's ability to vocally label all stimuli used in the TRA, answer yes/no questions, and follow single and multi-step directions. The TRA included two subtests: nonarbitrary temporal relations and arbitrary temporal relations. The nonarbitrary subtest materials included images of eight different common items (e.g., an apple, ball, cat, etc.). The arbitrary subtest included images of four different colored circles (blue, green, red, and yellow), as well as four images of simple motor actions (clapping, raising arm, touching nose, and waving) (see Procedure for complete description of materials used across each test).

### Procedure

At the onset of each session, the researcher first tested the audio, video, and screen-sharing features with the participant and their caregiver to ensure all features functioned properly before proceeding. Caregivers were instructed that the purpose of the test was to explore how children would respond to the presented activities and that they as caregivers should not provide answers or guidance to the children at any point during the assessment. Caregivers were asked to intervene only in the event of problems with the audiovisual display. For all test sections, the researcher first waited for participants to demonstrate an observing response (e.g., looking at the screen) prior to beginning the trial. It was determined that if any break in attending to the stimuli occurred (e.g., lag in video call,

participant looking away), the researcher would regain the participant's attention by stating their name and asking if they are ready to see the example again before proceeding. Once attending to the stimuli was re-established, the trial presentation for each test section continued as described.

#### Preassessment

During the first session, children were first administered the preassessment to determine suitability for inclusion in the study. The preassessment included three stages: tacting the stimuli used in the TRA, answering Yes/No questions, and performing simple motor responses. To test for tacting of the TRA materials, the child was shown common items or colors (one stimulus on the screen at a time) and asked by the examiner on each trial, "What is this?" A correct response was scored if the child accurately vocally labelled the stimulus shown on the screen (e.g., said "cat" when an image of a cat was shown). A total of 12 trials was conducted consisting of 8 pictures of common stimuli that were used in the nonarbitrary section of the TRA, and images of 4 colored circles that were used in the arbitrary section of the TRA. The child was then exposed to the same stimuli but tested on their ability to answer Yes/No questions. The examiner showed one stimulus on the screen at a time and asked the child a Yes/No question such as, "Is this an apple?" A total of 12 questions was presented, 6 to which "yes" was the correct response and 6 to which "no" was the correct response.

Following this, children were tested on following single and multi-step instructions with four different motor actions (clap hands, raise arm, touch nose, wave) relevant to the transformation of function section of the TRA. A total of 6 trials was presented, including each of the 4 single actions, 1 probe for two-step actions, and 1 probe for 3-step actions. In the case of each of the individual actions, the child was shown an image on the screen depicting that action and asked to demonstrate it (e.g., "Show me clapping."). A correct response was scored if the child acted out the action specified. If an error occurred, the researcher provided a model of the action and presented another opportunity to correctly perform the action. After each individual action was correctly demonstrated, the child was then asked to perform more than one (i.e., either 2 or 3) of the four actions in succession. A correct response was scored if the child performed each of the actions stated by the examiner but they were not required to perform them in a particular order. In total, the preassessment consisted of 30 individual trials. Participants were required to score 11 out of 12 trials correct on each the tacting and Yes/No sections, and 5 out of 6 trials correct on the motor actions section (total criterion of 27 out 30 trials correct, 90% correct) in order to continue with the study. During initial recruitment, one child did not pass the preassessment and was therefore excluded from the study. All the remaining children (n=25) passed the preassessment.

#### Temporal Relations Assessment (TRA)

The Temporal Relations Assessment included two major subsections: (i) nonarbitrary temporal relational responding (i.e., responding under the control of physically temporally related events) and (ii) arbitrarily applicable temporal relational responding (i.e., responding under the control of contextual cues that specify temporal relations between events). The assessment was

delivered using a combination of visual materials, auditory recordings, and orally presented instructions. For each trial, the particular visual pictures and/or words relevant to answering the question were presented to the participant. Once the materials were displayed, the principal researcher then vocally presented the pre-determined question.

Following completion of the preassessment, children were assessed on the TRA over the course of 1 to 2 sessions that lasted in total between 20-45 minutes per child. Session durations varied depending on breaks given during testing and the age of the child. The TRA was used to test responding to temporal relations at nonarbitrary and arbitrarily applicable levels across a total of 64 trials. Children were first tested on 16 trials of nonarbitrary relations, consisting of 8 questions with 2-image sequences and 8 questions with 3-image sequences. The testing of arbitrarily applicable relations for combinatorial entailment, and 16 questions for transformation of function (8 questions for transformation of function with mutually entailed relations and 8 questions for transformation of function of function with combinatorial entailment relations). For all trials, each correct response was scored as "1" while an incorrect response was scored as "0."

Throughout all sections of the assessment, no feedback was provided following correct or incorrect responses. In order to maintain engagement, non-specific verbal praise was provided on a schedule of around every four trials (e.g., "you're working really hard"). In between each section, the researcher briefly engaged the participant in conversation about preferred topics unrelated to the test. This was done to maintain a level of rapport and to provide brief breaks between tasks. Additionally, after test sections the researcher asked participants if they would like a brief break before resuming or if they would prefer to continue with the activities.

**Nonarbitrary Testing.** At the onset of the nonarbitrary section, participants were instructed that they would be shown a series of images and be asked questions about which image came before or after the other(s). Participants were instructed to give a "yes" or "no" vocal response to each question and were told that they could request to see the image sequence again if needed. The 16 trials included both 2-image and 3-image trials and included four trial types. There were 4 "before" trials with "yes" as the correct response, 4 "after" trials with "yes" as the correct response, 4 "before" trials with "no" as the correct response, and 4 "after" trials with "no" as the correct response (Table 3.1). Each of the stimuli used in the nonarbitrary section appeared in all sequential positions and in all trial types (see Appendix A for a full list of the nonarbitrary trials used).

Trial	2-Image Sequence			3-Image Sequence		
Туре	Observed sequence	Question	Answer	Observed Sequence	Question	Answer
1	A then B	Is A before B?	Yes	A then B then C	Is A before C?	Yes
2		Is B before A?	No		Is C before A?	No
3		Is A after B?	No		Is A after C?	No
4		Is B after A?	Yes		Is C after A?	Yes

**Table 3.1**Nonarbitrary Trial Types

In the 2-image nonarbitrary relations trials, a participant was shown a single stimulus (e.g., fish) on the screen for 1s with a simultaneous audio clip producing the name of the item. The image was then removed and a blank screen was shown for 1s, and then a second stimulus was displayed on the screen for 1s (e.g., shoe) again with an accompanying audio clip, and then this second image was removed and a blank screen shown again (Figure 3.1). The experimenter then presented a question regarding the order of the observed sequence using either a "before" or "after" cue (e.g., "Was the shoe before the fish?"). The same procedure was used for trials of 3-image sequences with the only difference being the addition of a third image. That is, an image was shown on the screen for 1s accompanied by an auditory stimulus, followed by a blank screen for 1s, and then a second stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third stimulus appeared for 1s, followed by a blank screen for 1s, and then a third

### Figure 3.1



#### Example Stimulus Presentations

*Note.* From left to right, the figure illustrates an example stimulus arrangement for a two-stimulus nonarbitrary trial, a three-stimulus nonarbitrary trial, a mutual entailment trial, and a combinatorial entailment trial.

Arbitrary Testing: Mutual and Combinatorial Entailment. Prior to beginning the arbitrary relations section, participants were shown sample items of the contextual cues used to indicate "before" and "after" respectively. For all arbitrary trials, the letter "B" was used to cue a "before" relation, and the letter "A" was used to cue an "after" relation. The researcher presented the letter "B" on screen and said, "B means before," and subsequently asked, "What does B mean?" This was repeated by a similar procedure involving the letter "A." Participants were then presented with a sample relational statement for both of the contextual cues. In the case of the "before" cue the researcher showed, for example, [blue circle] [B] [green circle] on the screen and read it aloud to the participant and a similar procedure was conducted in the case of the "after" cue. The researcher then presented two trials in which the participant was asked to read aloud the relational statement (one trial with a "before" cue and one trials with an "after" cue). If participants failed to accurately read aloud the relational statement, the researcher modeled the correct response, reviewed the contextual cues in isolation, and then re-presented a sample relational statement for each contextual cue. Participants had to read aloud the statement accurately on a single-trial probe for each cue before proceeding.

Participants were then tested for arbitrary relations. For all arbitrary trials, the relational statements were presented in a multimodal format as they were both visually displayed on the screen as well as being stated aloud. During trials of mutual entailment, the participant was shown an image consisting of 2 colored circles with a contextual cue in between them (Figure 3.1). An accompanying audio clip was played which stated the relational statement out loud (e.g., "blue is before red"). The researcher then asked the participant a question about the relation between the A-B stimuli (e.g., "Is red after blue?"). Participants were required to give a "yes" or "no" vocal response. During trials of combinatorial entailment, participants were shown an image consisting of 3 colored circles with interspersed contextual cues (Figure 3.1). An audio clip stated the relational statement aloud (e.g., "blue is before red, red is before yellow") and then the researcher asked the participant a question about the A-C relation (e.g., "Is yellow after blue?"). Again, participants had to give a "yes" or "no" vocal response. For both mutual entailment and combinatorial entailment sections, the 16 trials for each section were broken down into 8 unique trial types, examining both directly trained and entailed relations (Table 3.2). Each of the trial types was presented twice as the stimuli were systematically rotated so that they each appeared in all stimulus positions and in all trial types (see Appendix A for a full list of arbitrary trials used).

Mutual Entailment			Combinatorial Entailment		
Statement	Question	Answer	Statement	Question	Answer
A before B	Is A before B?	Yes	A before B before C	Is A before C?	Yes
	Is B before A?	No		Is C before A?	No
	Is A after B?	No		Is A after C?	No
	Is B after A?	Yes		Is C after A?	Yes
A after B	Is A before B?	No	A after B after C	Is A before C?	No
	Is B before A?	Yes		Is C before A?	Yes
	Is A after B?	Yes		Is A after C?	Yes
	Is B after A?	No		Is C after A?	No

Table 3.2

Mutual and Combinatorial Entailment Trial Types

Arbitrary Testing: Transformation of Function. In order to assess for transformation of function (ToF), the experimenter first trained a discriminative function in each of the arbitrary stimuli. Each of the 4 colored circles was shown one at a time alongside an image of a particular action (clap, raise arm, touch nose, wave). For example, the experimenter showed the red circle, and said, "This means clap," while modeling clapping. Participants were then instructed to copy the action modeled. This was repeated with the other three colors and actions. Subsequently, participants were probed across four trials (i.e., one trial per color) on their ability to demonstrate the correct action when shown the corresponding color. The researcher presented a colored circle on the screen alongside a visual that showed each color-response combination and instructed the participant, "Do this one." Social praise was delivered for correct responding and corrective feedback was used if an incorrect response was emitted. Participants were required to respond correctly to each of the 4 trials before proceeding to ToF testing. If any errors occurred, trials were re-presented until the participant independently demonstrated the corresponding action for each stimulus.

Once the participant had demonstrated each of the stimulus-response combinations correctly, they proceeded to the ToF testing. The 16 ToF trials included 8 mutually entailed ToF trials (ToF-ME) and 8 combinatorially entailed ToF trials (ToF-CE). For all trials, a visual sample was presented on the left side of the screen that showed each color-response combination and the relational statement was presented on the right side of the screen (see Figure 3.2). During the ToF-ME trials, a relational statement with either the "before" or "after" cue (e.g., red before blue) was displayed while an audio clip stated the relational statement out loud. The participant was then told to do the actions in the correct order. If the participant demonstrated the two motor-actions in the order that the relational statement indicated then this was recorded as a correct response. For ToF-CE trials, the same procedure was used with the difference that the relational statement involved 3 stimuli and participants were required to perform 3 motor-actions in the order that corresponded with the

relational statement. "Before" cues were used in 8 trials (4 ToF-ME trials and 4 ToF-CE trials) and "after" cues were used in 8 trials (4-ToF ME trials and 4 ToF-CE trials).

#### Figure 3.2

Example Stimulus Presentation for Transformation of Function Trials



#### **Interobserver Agreement and Procedural Fidelity**

Interobserver agreement (IOA) and procedural fidelity data were collected by trained research assistants for 5 of the assessment sessions (20% of participants) for both nonarbitrary and arbitrary relational responding tests. A second observer attended the video conferencing session from a different physical location than either the primary researcher or the participant. IOA was calculated on a trial-by-trial basis for each session and ranged from 94% to 100% (M = 98%). Procedural fidelity for each trial was assessed according to a fidelity checklist and was scored as 100% across observed sessions.

### Results

#### Preassessment

Age cohort results from each section of the preassessment and overall score are shown in Table 3.3. Within the 3-4 years cohort, each participant scored 100% correct on the tacting section. One participant from the 3-4 years group scored 92% correct (11 out of 12 trials) on the Yes/No responding section, and two participants scored 83% correct (5 out of 6 trials) on the motor responses section. All other participants scored 100% correct on all sections of the preassessment.

### Correlations

Table 3.4 shows Spearman's rank correlations between age at the date of testing and scores on the TRA, including total score and subscale scores for nonarbitrary and arbitrary sections.<sup>7</sup> Significant correlations were found between age and total score on the TRA ( $r_s = .72, p < .001$ ), between age and nonarbitrary temporal relational responding ( $r_s = .72, p < .001$ ), and between age and arbitrarily applicable temporal relational responding ( $r_s = .64, p < .001$ ). A significant correlation was

<sup>&</sup>lt;sup>7</sup> A nonparametric test was employed so as to be conservative given the relatively small sample size involved.

also found between scores on the nonarbitrary and arbitrarily applicable temporal relations subtests ( $r_s = .77, p < .001$ ).

#### Table 3.3

Age Cohort Mean Scores on the Pre-assessment

Age Cohort	Tact	Yes/No	Motor Actions	Total
3-4 years	100%	98%	93%	98%
4-5 years	100%	100%	100%	100%
5-6 years	100%	100%	100%	100%
6-7 years	100%	100%	100%	100%
≥7 years	100%	100%	100%	100%

### Table 3.4

Spearman's Rho Correlations for Age and Temporal Relations Assessment (TRA)

2	3	4	
.915**			
.943**	.766**		
	2  .915** .943**	2 3  .915** .943** .766**	2 3 4  .915** .943** .766**

*Note.* \*\* Correlation is significant at the 0.01 level (2-tailed). Total = total relational responding score; NARR = nonarbitrary relational responding score; AARR = arbitrary applicable relational responding score

#### Figure 3.3

Total Scores on the Temporal Relations Assessment (TRA) vs. Age



#### **Relational Responding Scores and Age**

Figure 3.3 graphs the correlation between the total score for each participant on the TRA (out of a total of 64 possible points) and age in months. The R<sup>2</sup> value indicates the proportion of variance in total score that can be attributed to age (.53, low to moderate). Note that the 4-5-year-old cohort featured two of the most extreme outliers one of which was 2 standard deviations below the best-fit line. The percentage of correct responses across all sections of the TRA for each participant is shown in Table 5. Total scores on the TRA total assessment ranged from 30% to 83% correct (M = 57%, SD = 8.05). On the nonarbitrary section, scores ranged from 31% to 100% correct (M = 75%, SD = 3.5); on the mutual entailment section, scores ranged from 31% to 69% (M = 61%, SD = 2.20); on the transformation of function section, scores ranged from 0% to 100% (M = 44%, SD = 3.58).

Table 3.5 shows the average percentage of correct responses for each cohort both for the TRA as a whole and for each assessment section. The mean scores on the nonarbitrary temporal relations section showed a clear improvement between the youngest and oldest participants (M = 55% to 94%) correct, respectively). Performance on the nonarbitrary section was similar for the 3-4 years and 4-5 years cohorts (M = 55% and 58% correct, respectively) but then a substantial improvement is seen in the 5-6 years cohort (M = 80% correct). Performance continued to improve in the 6-7 years (M =89% correct) and  $\geq$  7 years cohorts (M = 94% correct). Mean scores on the mutual entailment section also showed a general trend of improvement with age, with the exception of the 6-7 years cohort, who scored the lowest out of any age cohort (M = 53% correct). However, between 3-7 years, group means on the mutual entailment section remained near chance-level responding. For the combinatorial entailment section, no clear improvement with age was apparent and the means for all age groups were near chance-level responding. Mean scores on the ToF-ME and ToF-CE sections each showed improvement between the youngest and oldest participants (M = 38% to 65% correct and 28% to 65% correct, respectively). The 3-4 years and 4-5 years cohorts scored similarly on the ToF-ME section (28% and 29% correct, respectively) and the ToF-CE section (28% and 21% correct, respectively). The 5-6 years and 6-7 years cohorts also scored similarly on the ToF-ME section (56% and 53% correct, respectively) and on the ToF-CE section (47% correct for both cohorts).

ID	M/F	Age (mo)	NARR (%)	ME (%)	CE (%)	ToF-ME (%)	ToF-CE (%)	Total (%)
3-4 ye	3-4 years age cohort							
3	F	39	8 (50%)	7 (44%)	9 (56%)	0 (0%)	0 (0%)	24 (38%)
4	М	39	7 (44%)	8 (50%)	8 (50%)	4 (50%)	3 (38%)	30 (47%)
5	М	43	11 (69%)	7 (44%)	7 (44%)	4 (50%)	4 (50%)	33 (52%)
2	М	44	9 (56%)	13 (81%)	5 (31%)	2 (25%)	2 (25%)	31 (48%)
1	F	48	9 (56%)	11 (69%)	8 (50%)	5 (63%)	2 (25%)	35 (55%)
Mean		42.5	8.8 (55%)	9.2 (58%)	7.4 (46%)	3 (38%)	2.2 (28%)	30.6 (48%)
SD		3.63	1.48	2.68	1.52	2	1.48	4.16
4-5 ye	ars age co	ohort						
6	М	50	5 (31%)	6 (38%)	8 (50%)	0 (0%)	0 (0%)	19 (30%)
8	F	53	11 (69%)	9 (56%)	9 (56%)	3 (38%)	4 (50%)	36 (56%)
11	F	57	8 (50%)	8 (50%)	8 (50%)	1 (13%)	0 (0%)	25 (39%)
10	F	58	7 (44%)	10 (63%)	11 (69%)	3 (38%)	0 (0%)	31 (48%)
7	М	59	15 (94%)	14 (88%)	9 (56%)	4 (50%)	3 (38%)	45 (70%)
9	М	59	10 (63%)	8 (50%)	6 (38%)	3 (38%)	3 (38%)	30 (47%)
Mean		56.3	9.3 (58%)	9.2 (58%)	8.5 (53%)	2.3 (29%)	1.7 (21%)	31 (48%)
SD		3.77	3.50	2.71	1.64	1.51	1.86	8.97
5-6 ye	ars age co	ohort						
12	М	65	15 (94%)	12 (75%)	8 (50%)	5 (63%)	4 (50%)	44 (69%)
14	F	66	16 (100%)	10 (63%)	9 (56%)	4 (50%)	4 (50%)	43 (67%)
15	F	66	13 (81%)	9 (56%)	9 (56%)	4 (50%)	4 (50%)	39 (61%)
13	М	69	7 (44%)	10 (63%)	5 (31%)	5 (63%)	3 (38%)	30 (47%)
Mean		66.4	12.8 (80%)	10.3 (64%)	7.8 (48%)	4.5 (56%)	3.8 (47%)	39 (61%)
SD		1.35	4.03	1.26	1.89	0.58	0.5	6.38
6-7 ye	ars age co	ohort						
17	F	73	14 (88%)	10 (63%)	9 (56%)	6 (75%)	3 (38%)	42 (66%)
18	F	75	15 (94%)	7 (44%)	8 (50%)	4 (50%)	4 (50%)	38 (59%)
16	М	78	14 (88%)	9 (56%)	7 (44%)	3 (38%)	4 (50%)	37 (58%)
19	F	82	14 (88%)	8 (50%)	8 (50%)	4 (50%)	4 (50%)	38 (59%)
Mean		77	14.3 (89%)	8.5 (53%)	8 (50%)	4.3 (53%)	3.8 (47%)	38.8 (61%)
SD		3.82	0.50	1.29	0.82	1.26	0.5	2.22
≥7 yea	irs age col	hort						
22	М	85	15 (94%)	12 (75%)	8 (50%)	4 (50%)	4 (50%)	43 (67%)
24	М	85	15 (94%)	11 (69%)	8 (50%)	4 (50%)	4 (50%)	42 (66%)
23	F	88	13 (81%)	8 (50%)	10 (63%)	4 (50%)	4 (50%)	39 (61%)
21	F	90	16 (100%)	12 (75%)	7 (44%)	4 (50%)	4 (50%)	43 (67%)
20	F	92	16 (100%)	13 (81%)	8 (50%)	8 (100%)	8 (100%)	53 (83%)
25	М	101	15 (94%)	12 (75%)	8 (50%)	7 (88%)	7 (88%)	49 (77%)
Mean		90	15 (94%)	11.3 (71%)	8.2 (51%)	5.2 (65%)	5.2 (65%)	44.8 (70%)
SD		5 89	1 10	1 75	0.98	1 84	1 84	5.15

Table 3.5

Participants' Scores on the Temporal Relations Assessment (TRA)

#### Age Related Performance across Contextual Cues

An analysis of performance across temporal trial types (i.e., "before" versus "after") was also conducted to further investigate the results found, particularly with respect to the arbitrarily applicable relations sections of the assessment. Figures 3.4-3.6 show the average correct responses per age cohort and relational response type (nonarbitrary, mutually entailed, combinatorially entailed, and transformation of function) based on the contextual cue used (i.e., "before" versus "after"). The NARR data in Figure 4 refer to which word was used in the question for that section of the assessment. For mutual entailment, combinatorial entailment, and transformation of function, the "before" data in Figures 5 and 6 represent trials in which the relational statement read, "A before B" and "A before B before C" (see trial types in top half of Table 3.2). "After" data represent trials in which the relational statement read, "A after B" and "A after B after C" (see trial types in bottom half of Table 3.2).

For the nonarbitrary-2 stimuli section (Figure 3.4), the 3-4-year-old cohort responded with slightly higher accuracy on "before" trials compared to "after" (M = 60% and 63% correct, respectively) as did the 4-5-year-old cohort (55% and 50% correct, respectively). The 5-6-year-old cohort however, performed with higher accuracy on "after" trials (M = 81% correct) compared to "before" trials (M = 63% correct). The mean scores for the 6-7-year-old cohort were the same across both trial types (M = 94% correct), while the  $\geq$  7-year-old cohort scored higher on "before" compared to "after" trials (M = 100% and 92% correct, respectively). For the nonarbitrary-3 stimuli section (Figure 4), the 3-4-year-old cohort scored higher on "after" trials (M = 55% correct) compared to "before" (M = 50% correct). The 4-5-year-old cohort performed better on "before" trials (M = 67% correct) compared to "after" trials (M = 54% correct). The mean scores for the 5-6-year-old cohort were the same across both trial types (M = 88% correct). The 6-7-year-old group performed slightly better on "before" trials than "after" (M = 88% and 81% correct, respectively) as did the  $\geq$  7-year old-group (M = 96% and 88\% correct, respectively).

On the mutual entailment section (Figure 3.5), there was an improvement on "before" trials with age with the exception of the very oldest group (M = 60%, 71%, 78%, 91% 77% correct, respectively moving from youngest to oldest). On "after" trials performance remained near chance level responding in the three youngest age groups (55%, 44%, and 50% correct, respectively) but then in the 6-7-year-old cohort the mean scores were substantially lower than had been the case for the previous three groups (M = 16%) and then performance for the oldest cohort was the highest among the cohorts (M = 65%). As regards a comparison between performance on "before" and "after" it might be noted that difference in mean scores across "before" and "after" mutual entailment trials became increasingly pronounced from the 3-4-year-old cohort to the 6-7-year-old cohort.

As regards combinatorial entailment (Figure 3.5), performance on "before" trials was lowest in the 3-4-year-old group (48% correct) and somewhat better in the 4-5 and 5-6-year-old groups (67% and 63%, respectively). The two oldest cohorts both showed high levels of correct responding on "before" trials, with the 6-7-year-olds performing better (M = 97%) than the  $\ge 7$  years group (M = 81%). In contrast, performance on "after" trials showed a gradual disimprovement with age moving from the 3-4 to the 5-6-year-old cohorts (M = 45%, 40%, 34% respectively) before showing a much lower performance from the 6-7-year-old age group (3% correct) and then a slight increase again in the oldest cohort (21% correct). As in the case of mutual entailment trials, the difference in mean scores on "before" and "after" trials became increasingly pronounced between the 3-4-years group up to the 6-7-years group.

Results on both transformation of function trial types (ToF-ME and ToF-CE; Figure 3.6) show that the participants 5 years and older demonstrated high levels of accuracy with "before" trials while mean scores on "after" trials for the transformation of function sections remained low across all age cohorts. It is noted that only two participants (P20 and P25 who were both in the  $\geq$  7 years group) scored high on the "after" trials in the transformation of function sections (100% and 75% correct, respectively).

### Figure 3.4

Average Percentage Correct per Age Cohort on Nonarbitrary Relational Responding (NARR) Trials



**NARR** Trials



Average Percentage Correct per Age Group on Mutual Entailment and Combinatorial Entailment Trials





### Figure 3.6

Average Percentage Correct per Age Group on Transformation of Function Trials



# **Transformation of Function Trials**

### Discussion

The present study aimed to examine the development of temporal relational responding in young children across a range of ages. Specifically, children between 3 and 8 years of age were tested

on an assessment of nonarbitrary and arbitrary temporal relations. To date, RFT research has minimally investigated temporal relational frames and the research that has been done has primarily investigated this repertoire in adults with an existing repertoire.

The present study expands on prior RFT research in this area and provides new empirical data on the acquisition of temporal relational responding patterns in children. Overall performance on the TRA was shown to increase with age. Significant correlations were found between age and total score on the assessment as well as between age and performance on both the nonarbitrary and arbitrary relations sections. Performance on the nonarbitrary temporal relations section steadily increased with age as average scores for each age group tended to be higher than for younger groups. Children 5 years and older in particular showed high levels of responding on nonarbitrary trials. Other patterns were observed on the arbitrary relations section, which included trials for mutual entailment, combinatorial entailment, and transformation of function. Performance on mutual entailment and transformation of function trials generally increased between the youngest and oldest participants, though errors were still made by the oldest participants. Accuracy on combinatorial entailment trials did not appear to improve with age as all cohorts performed near chance level responding. However, when analyzed according to trial types, notable differences in performance between "before" and "after" trials were observed (Figures 3.4-3.6). This pattern will be elaborated on below.

One key finding from the present study was that different patterns were observed across ages when comparing responding in accordance with nonarbitrary temporal relations and responding in accordance with arbitrary temporal relations. Namely, with the nonarbitrary section of the assessment, accuracy on "before" and "after" questions improved similarly with age (see Figure 3.4). However, on arbitrary sections of the assessment, better accuracy was consistently observed across all ages on trials that included "before" as the cue for the trained relation compared to trials that included "after" as the cue for the trained relation (see Figures 3.5 and 3.6). For example, trials that showed a relational statement such as, "red before blue" and included questions like, "Is blue before red?" or "Is blue after red?" were on average responded to with greater accuracy than trials that showed a relational statement such as, "green after yellow" with questions like, "Is green before yellow?" or "Is green after yellow?"

While performance on arbitrary trials with "before" as the trained contextual cue generally improved with age, the pattern of performance on arbitrary trials with "after" as the trained contextual cue was more complex. For the mutual entailment section, mean scores on "after" trials remained around chance level responding between 3-6 years. A substantial drop in mean accuracy on these trials was observed in the 6-7-year-old cohort (M = 16% correct) followed by a substantial improvement to the highest-level performance of any cohort in the  $\geq$  7-year-old group (M = 65%correct). For the combinatorial entailment section, a somewhat similar pattern was observed for the younger age groups to the extent that the 3-6-year-olds showed low levels of correct responding, and the 6-7-year-olds again showed a substantial drop in correct responding (M = 3% correct) compared

to the 3-6-year-olds. However, in this case while the oldest cohort did show an increase in mean accuracy (M = 21% correct) compared to the 6-7-year-old group their level of performance remained low. For transformation of function trials, children 5 years and older demonstrated high levels of accuracy with "before" trials but all ages had low levels of accuracy on "after" trials.

These results suggest firstly that trials in which "after" was the stated relation were more challenging than trials in which "before" was the stated relation across all arbitrary sections. This coheres with previous work with adults which also found that participants tended to respond slower and with less accuracy with "after" relations (Brassil et al., 2019; Hyland et al., 2012, 2014; McGreal et al., 2016). One possible reason for why both children and adults might be less accurate with "after" than "before" relations is as follows. Temporal framing is acquired gradually, as with all frames. Particularly, there may be functionally different aspects of the relational pattern to be acquired. Children might learn "before" relations than "after" relations. It is also plausible that children receive more practice deriving an entailed "after" relation based on "before" than vice versa. Further research is suggested to investigate potential differences between the acquisition of these contextual cues in children.

One specific issue affecting the acquisition of temporal relational responding that might contribute to differential performance with respect to "before" and "after" contextual control (and one previously hinted at in cognitive developmental research by Clark, 1971) is the potential of a conflict between a nonarbitrary temporal relation controlling the response and the arbitrary temporal relation controlled by the contextual cue. For instance, when a participant was told the trained relation, "red is before blue, blue is before yellow," the experienced temporal order of the stimuli in terms of what was heard auditorily (and possibly observed visually) was "red"..."blue"..."yellow." This experienced sequence coheres with the stated arbitrary temporal relation. However, when told, "red is after blue, blue is after yellow," the experienced sequence is the same but in this case, the arbitrary relation conflicts with the nonarbitrary relation. Previous RFT research has shown that such conflicts can lower levels of correct derivation of the arbitrary relation (Kenny, Barnes-Holmes, et al., 2014; Kenny, Devlin, et al., 2014; Stewart et al., 2002).

If the relational statement is presented visually then the same potential issue arises but in that case there is also a spatial dimension involved. To illustrate, Figure 3.7 below shows relational statements with both "before" and "after" contextual cues with the added question, "Is yellow after red?" With the "before" relational statement, a person may correctly derive that yellow is after red based purely on the arbitrary cue, or they might also relate yellow as after red based on the spatial relation of left-to-right (because yellow is further to the right of red). Either way, there is congruence between the temporal order according to the contextual cue and the spatial order. This congruence is not the case when "after" cues are used in a statement. For example, in Figure 7 on the right side, a person may correctly derive that red is after yellow if they respond in accordance with the arbitrary

contextual cue. If, however, they are responding in accordance with a spatial relation of left-to-right, they will answer this question as though yellow is after red (because yellow is further to the right of red). In other words, there is incongruence between the order based on the arbitrary contextual cue and the spatial order.

### Figure 3.7

Comparison of Relational Statements Using "Before" and "After" Contextual Cues



An interaction between spatial and temporal relations may be sensitive to one's individual reinforcement history. For instance, while terms like "before" and "after" are often used for temporal relations, they are also used for spatial relations. Barnes-Holmes and colleagues (2001) suggested that young children's misuse of prepositions (e.g., "can I have any reading *behind* dinner," Pinker, 1990) might be explained by children initially developing more general relational frames, which are then shaped to be more specific through interactions in the verbal community. Prepositions for temporal and spatial relations in particular bear similarity and may initially participate in a more general frame before they are acquired as contextual cues for distinct temporal and spatial relations.

In English-speaking socioverbal communities, people read from left-to-right and children are typically taught to sequence items from left-to-right, thereby possibly establishing a learning history that stimuli to the left may come "before" stimuli to the right. Indeed, some recent work has shown that a directional preference (e.g., left-to-right) for ordering events gradually emerges in childhood (Tillman et al., 2022; 2018). This too is arbitrary as a left-to-right order is based on social convention, and research from other domains has showed humans spatially represent time in other ways such as right-to-left (Fuhrman & Boroditsky, 2010) top-to-bottom (Boroditsky, 2001) and even according to cardinal directions like east-to-west (Boroditsky & Gaby, 2010). Further, research has shown that people can be trained to spatially represent time in different ways (Hendricks & Boroditsky, 2017). Nevertheless, if a history of reinforcement is established for reading or ordering from left-to-right, then it is likely the person will actually experience the stimuli in this order in a nonarbitrary sense. This means that they may orient to the image on the left first, and then scan towards the right. Correctly answering "after" trials is potentially more difficult, as the arbitrary temporal relation is incongruent with how the stimuli are experienced nonarbitrarily if reading from left-to-right.

If a person were to respond in this manner (based on the spatial-order), this would likely produce a response pattern where questions that involve stimulus arrangements like that on the left of Figure 3.7 are answered correctly and stimulus arrangements like that on the right of Figure 7 are answered incorrectly. That is, a child could score 50% correct on all arbitrary trials by responding to the spatial order of presentation without using the "before" and "after" cues. Accordingly, the relational response may be under stimulus control of an observed spatial relation instead of under the control of the contextual cue. Additionally, the same pattern would occur on a transformation of function trial. If relating the stimuli based on a spatial relation (i.e., further left is "before" and further right is "after"), a person may act out the actions in sequence based on the colors from left-to-right regardless of the arbitrary cue presented. Indeed, participants 5 years and older showed high levels of accuracy to ToF "before" trials, while nearly all failed the "after" trials. As an anecdotal example, P12 accurately read aloud the relational statement during ToF trials, (i.e., vocally stated that "blue is after green") yet performed the actions according to the spatial left-to-right order (blue...then green).

Apart from the difference in performance on "before" and "after" trials one other interesting aspect of the current data is the decrement in performance on "after" trials in the 6-7-year-old group in particular compared to the younger groups. For the ME trials this substantial drop in performance for the latter cohort is juxtaposed with a significant improvement in the oldest cohort. In the CE trials there is a similar drop in performance in the 6-7-year-olds that again is followed by an improvement in the oldest group, though in the case of the CE trials this improvement was not a substantial one. One possible explanation for this pattern is that the youngest groups are simply guessing (as evidenced by the fact that they are showing a correct rate of around 50%) whereas the 6-7-year-olds are responding systematically but under the wrong stimulus control. More specifically, it might be that they are in fact responding under the pattern of stimulus control discussed in the previous paragraphs. For example, if participants were responding on "after" trials not in accordance with the arbitrarily applicable temporal relation but in accordance with the nonarbitrary relation between the stimuli in the trained statement (whether temporal or spatial) then they would get zero or close to zero correct. In contrast, the obvious improvement in performance on ME "after" trials in the oldest cohort could be due to the fact they have by now acquired the repertoire of appropriate derivation in accordance with temporal ME and this overrides to some extent potential influence by nonarbitrary relations. In the case of the CE relations even the oldest cohort did not show correct responding for "after" relations, suggesting that in this case this cohort had yet to acquire this repertoire.

The present study adds to the body of RFT research more broadly in a number of ways. It expands work on temporal relations to a new population and provides data for the development of temporal framing in children. It expands on the results by Kirsten and Stewart (2021) relative to temporal frames by assessing this frame more specifically and adding a test for transformation of function with temporal relations. The test for transformation of function was adapted from O'Hora et al. (2004) and provides an initial approach to assessing this relational frame property with temporal

relations in children. As has been discussed, the current work also highlights potential key differences in the acquisition of "before" vs "after" with respect to contextual control. Finally, a secondary benefit of the present study is that the relational assessment was carried out entirely online. As the use of online learning models continues to grow, future RFT research may find utility in further exploring online formats (e.g., synchronous, asynchronous) for testing and training of relational skills in children.

The present data also provides new insight on non-RFT based research on "before" and "after" comprehension in children. Research on the comprehension of "before" and "after" has historically been limited to examining how their placement in a sentence affects performance. The present study however, sought to examine these terms in the context of contextual control over relational responding (both nonarbitrary and arbitrary). From an RFT view, the cognitive developmental or linguistic studies that involve answering questions about observed actions coming "before" or "after" others (Blything et al., 2015; McCormack & Hanley, 2011) are likely assessing responding under the control of nonarbitrary relations, whereas studies involving questionnaires or hypothetical scenarios rely on AARR (Pyykkönen & Järvikivi, 2012). Distinguishing between NARR and AAAR offers new insight on the acquisition of temporal understanding in children as the primary source of stimulus control over temporal NARR is the actual experience of sequential change, while temporal AARR is under control of the contextual cue itself. In the present study, the NARR data indicated that performance with both terms similarly improved with age. The AARR data however, showed that deriving temporal relations in the presence of "after" cues.

There are a number of limitations of the current study that could be addressed in future research. First, this study did not include any norm-referenced measures of cognitive or language abilities to which performance on the TRA could be compared. Other RFT studies have shown strong correlations between performance on relational tasks and standardized measures of cognitive or language abilities (Kent et al., 2017; Kirsten & Stewart, 2022; Mulhern et al., 2017). By including similar measures in future work, this line of research could be extended specifically to temporal relations with children.

A second consideration is the relatively small sample size and relatively limited range of participant ages in the present study. Using a larger sample across all age cohorts would improve the external validity of results and might provide further insight into developmental patterns. In addition to the size of the sample, future research should also look at expanding the assessment to older age groups. While overall performance on the temporal relations improved with age, error patterns with arbitrary relations were still present in the oldest participants within the present study. As previously mentioned, even adults show some differences between derivations based on "after" versus "before" relations. Conducting this assessment with a wider age range might help to further clarify at what age(s) children might begin to more reliably demonstrate mastery of these types of relational tasks.

Another consideration for future research is to address the relatively limited trial types used. In the present study, only yes/no questions were used for the nonarbitrary, mutual entailment, and combinatorial entailment sections. Future research might include additional response topographies, such as tacting the correct contextual cue on trials or selection-based responses to more fully examine the relational repertoires. Secondly, the nonarbitrary section used only short (1s) durations between stimuli. Future work could present questions of nonarbitrary temporal relations with events that happened over a longer time frame (e.g., a few minutes ago). Additionally, the current study exclusively used multimodal relational statements (i.e., participants saw the relational statement while hearing it spoken aloud). It remains unclear to what extent each of these modalities might have facilitated performance. Future work might compare performance across visual, auditory, or multimodal presentations of relational statements.

The present study also included different stimuli across test sections with more concrete stimuli used on nonarbitrary trials and only colored shapes on arbitrary trials. Concrete stimuli were selected for the nonarbitrary trials to provide more salient exemplars and a clearer discrimination between stimuli within and across trials since the task involved the presentation/removal of stimuli. The use of colored shapes for the arbitrary trials was however adopted from recent empirical work using this REP format (Kirsten & Stewart, 2022; Kirsten et al., 2022a, 2022b). We suggest that future research utilize the same stimuli across both nonarbitrary and arbitrary tasks, whether it be more meaningful or abstract stimuli. Further, the present study only used contextual cues between the relata ("A before B"; "B after A") to be consistent with the REP format. However, in natural environments, these cues may also be contacted in the first part of sentences (e.g. "Before B, A"; "After A, B"). As previously noted, other psychological research has showed performance differences with these arrangements in children (McCormack & Hanley, 2011) and this could be evaluated from an RFT perspective.

Additionally, only "before" and "after" cues were used to assess nonarbitrary and arbitrary temporal relations. While these two words are important to contextually controlled temporal relational responding, investigating performance with additional words might be desirable to closer examine the flexibility, fluency, and breadth of the relational repertoire (Luciano et al., 2009). Words like "first" and "last" are other commonly used words to which young children may often be exposed. Some linguistic research has shown that young children may more easily follow instructions with "first" and "last" (e.g., "Move the red car first; move the blue car last,") compared to "before" and "after" ("Move the red car before you move the blue car) due to the absence of a subordinate clause (Amidon & Carey, 1972). In the present study, "before" and "after" were specifically selected because they advantageously describe a relation between stimuli in a manner similar to that seen in other RFT studies on relational framing, whereas "first" and "last" more so describe a feature of a stimulus or deal with quantification of the (temporal) dimension. Nonetheless, these terms could be used for example, by giving a relational context (e.g., "if A is after B and B is after C…") and assessing for a

transformation of ordinal functions using "first" and "last" ("...Which is first/last?"). Other types of temporal relations could also be examined by using experienced past events ("Which was a longer time ago, your birthday or Christmas?") or hypothetical future events ("Which comes sooner, your birthday or Christmas?"). Future research might integrate these terms as well as other commonly used words into protocols of assessment and training of temporal relational responding.

Based on the present results, future research might seek to investigate the relationship between cues for spatial and temporal relations further as time is often represented spatially. In the present study, visual stimuli were presented horizontally and read from left-to-right. As previously discussed, notable differences in performance were observed between the arbitrary temporal contextual statements in the current study. For instance, did the participants that responded correctly to "after" trials (such as those to the right in Figure 3.7) respond under control of the contextual cue for a temporal relation or might they have responded based primarily on a reversed spatial relation? In other words, was a rule derived about the spatial order (e.g., right-to-left order) which thereby altered responding? More investigation is needed to examine to what extent arbitrary spatial relations might interfere or compete with stimulus control under the temporal contextual cues in children.

A final recommended research direction is to examine training procedures for temporal relational responding. The aim of the present study was to explore how children at different ages perform with relational tasks when assessed and did not include a specific training component. One direction might be to investigate directly training both temporal NARR or AARR repertoires directly. For instance, the current study found that deriving arbitrary relations when given an "after" cue to be particularly difficult for participants. Future research could look to train this skill in particular. Another direction might consider to what extent training a temporal NARR repertoire facilitates acquisition of temporal AARR. In the current study, the NARR repertoire appeared to hit mastery prior to the AARR repertoire (which in fact with respect to some specific patterns was not seen in any of the cohorts) and they strongly correlated with each other. RFT research on other relations has shown that NARR is an important precursor ability to AARR and may help support the acquisition of particular frames (Berens & Hayes, 2007). Similarly, another direction could be to examine whether training other particular frames help facilitate the acquisition of temporal framing. Investigating further which relational repertoires are important precursors could help pinpoint optimal relational training sequences that involve temporal relations.

### Conclusion

Despite the limitations, the present study provides insight into children's performance on tasks of nonarbitrary and arbitrary temporal relational responding. The current data expands the research base on temporal frames to a new population and adds to the growing RFT literature on relational responding abilities more generally. The results can be used as a starting point for further investigation into how contextual control is acquired for temporal relations and how these repertoires may be best trained in children.

Chapter 4.

Developing Procedures for Training Temporal Relational Responding in Children

Existing research on temporal skill sets—and more specifically temporal relational responding—has minimally investigated the environmental variables that influence the acquisition of temporal understanding. While some research has shown that feedback and other instructional arrangements can improve understanding of temporal order in sentences (Amidon & Carey, 1972; Ehri & Galanis, 1980), studies are largely descriptive in nature. Turning towards a behavioral approach may assist with addressing this limitation on training these skills due to an emphasis on both prediction and influence of behavior (Biglan & Hayes, 1996). One particularly relevant approach in this regard is relational frame theory (RFT; Hayes, Barnes-Holmes, et al., 2001; Stewart, 2016), a contextual behavior analytic account of human language and cognition. Substantial empirical research now attests to the efficacy of RFT as a basis for teaching basic and complex language skills to children and adults with and without disabilities (Gilroy et al., 2015; Kirsten et al., 2022a; Ming et al., 2018; Newsome et al., 2014; M. O'Connor et al., 2017; Tarbox et al., 2011; Wymer et al., 2016)

RFT research has also developed methods that can be used to *predict* behavior (e.g., the Implicit Relational Assessment Procedure, Hughes & Barnes-Holmes, 2013; Kelly & Barnes-Holmes, 2013). Indeed, one aim of the previous study in Chapter 3 was to provide initial data on how we might generally expect a child to respond at a given age on tasks of temporal relational responding. Critically, however, an important implication of derived relational responding being a form of generalized operant behavior is that it can be specifically taught. As such, an RFT approach can also be used to *influence* behavior—a central component of the scientific goal of a functional contextual perspective. Since AARR can both be established and strengthened, the procedures used to develop this skill can be studied to identify effective ways of influencing behavior (Barnes-Holmes, Barnes-Holmes, & McHugh, 2004; Barron et al., 2019; Berens & Hayes, 2007; Dunne et al., 2014; Gilroy et al., 2015; Ming et al., 2018; Mulhern et al., 2018; Rehfeldt & Root, 2005). As such, this goal of predicting and influencing behavior, can be applied specifically to training derived temporal relations.

From an RFT view, the acquisition of relational responses is not happen-stance or the product of a developmental stage. Similar learning contexts that often happen in natural interactions can be created and used in a systematic way to teach AARR to children. One such teaching approach that has successfully been used is multiple exemplar training (MET; Barnes-Holmes, Barnes-Holmes, et al., 2004). The use of MET has been shown to be an effective strategy to establish arbitrarily applicable relational responding in children in the presence of contextual cues for coordination (Rehfeldt & Root, 2005), distinction (Dunne et al., 2014), comparison (Berens & Hayes, 2007), opposition (Barnes-Holmes, Barnes-Holmes, et al., 2004), hierarchy (Ming et al., 2018), and advanced frames such as deixis (Gilroy et al., 2015) and analogy (Carpentier et al., 2002).

Applying MET to relational skills involves rotating a variety of stimuli while keeping the targeted contextual cues the same. With temporal relations, training could for instance involve varying stimuli used alongside the specific contextual cues "before" and "after." Across a range of exemplars, a learner is reinforced for responding consistent with the relational pattern and provided

error correction as necessary. Once criterion is reached with one set, the learner is then exposed to a novel set of stimuli and tested for relational responding in the absence of reinforcement procedures.

No work has thus far investigated the training of temporal relations in young children. The aim of the present study is to pilot the efficacy of a MET protocol based on RFT to train temporal relational responding in young children. Specifically, the present study sought to train mutually and combinatorially entailed temporal relations with "before" and "after" cues.

### Method

## **Participants and Setting**

Three children agreed to participation and two were included in the study. P1 was a female aged 6 years and P2 was male aged 6 years. Both participants were typically developing and attend schools in the west of Ireland. Participants at this age were recruited based on their performance on a previously administered test of temporal relations (Neufeld et al., 2023). Each participant demonstrated high accuracy on nonarbitrary temporal relations tasks but not with arbitrary temporal relations tasks. A third child was omitted from the study due to passing mastery criterion on baseline probes. Ethical approval for recruitment of participants was first obtained from the research ethics committee at the host institution of the lead researcher. Parental consent for participant.

All sessions were held online through a video conferencing platform with end-to-end encryption (Zoom). Sessions were held at convenient times during the day for both families via a networked device in their household. All probe and training were administered by the primary researcher. Prior to beginning each session, an accompanying family member was present to assist with ensuring all audio and visual features functioned properly. At the onset of a session, the accompanying family did not remain on-screen but remained in a nearby room in their home. All sessions were held up to 3 times per week, with each session lasting between 15-20 minutes.

#### **Experimental Design**

A combination noncurrent multiple baseline design across participants and a multiple probe design across responses was used for the current study. Prior to sessions, participants were preassigned to a baseline length of 3 or 5 sessions. Predetermined baselines were used in order to improve the internal validity of the nonconcurrent design (Christ, 2007; Watson & Workman, 1981). **Materials** 

Session materials were displayed on a 13-inch MacBook Pro running Microsoft PowerPoint via the screen-share feature of Zoom. The materials used for probe or training sessions included sets of images of common stimuli. Four different stimulus sets were included, with each set including 4 unique stimuli (see table 4.1). During intervention sessions, tokens were used via ClassDojo in which a virtual avatar was administered points.

Set 1	Set 2	Set 3	Set 4
Red	Cat	Square	Apple
Blue	Bird	Circle	Banana
Yellow	Fish	Star	Pizza
Green	Pig	Triangle	Ice cream

Table 4.1Multiple Exemplar Training (MET) Stimulus Sets

### Measurement

Accuracy of responding was measured on first trial probes across each trial-type for both mutually entailed responses and combinatorial entailment responses. Each trial type is shown below in Table 4.2. For each trial, a response was scored as correct if the participant gave a "yes" vocal response when the answer was yes, or a "no" vocal response when the answer was no. Scores were totaled out of 16 and then converted to a percentage for both mutual entailment and combinatorial entailment trials.

### Table 4.2

Mutual and Combinatorial Entailment Trial Types

Relational Level	Trial Type	Relational Statement	Question	Answer
Mutual Entailment	1	A1 before B1	Is A1 before B1?	Yes
	2	A2 before B2	Is B2 before A2?	No
	3	A3 before B3	Is A3 after B3?	No
	4	A4 before B4	Is B4 after A4?	Yes
	5	B1 before A1	Is A1 before B1?	No
	6	B2 before A2	Is B2 before A2?	Yes
	7	B3 before A3	Is A3 after B3?	Yes
	8	B4 before A4	Is B4 after A4?	No
	9	A1 after B1	Is A1 before B1?	No
	10	A2 after B2	Is B2 before A2?	Yes
	11	A3 after B3	Is A3 after B3?	Yes
	12	A4 after B4	Is B4 after A4?	No
	13	B1 after A1	Is A1 before B1?	Yes
	14	B2 after A2	Is B2 before A2?	No
	15	B3 after A3	Is A3 after B3?	No
	16	B4 after A4	Is B4 after A4?	Yes
Combinatorial Entailment	1	A1 before B1 before C1	Is A1 before C1?	Yes
	2	A2 before B2 before C2	Is C2 before A2?	No
	3	A3 before B3 before C3	Is A3 after C3?	No
	4	A4 before B4 before C4	Is C4 after A4?	Yes

5	C1 before B1 before A1	Is A1 before C1?	No
6	C2 before B2 before A2	Is C2 before A2?	Yes
7	C3 before B3 before A3	Is A3 after C3?	Yes
8	C4 before B4 before A4	Is C4 after A4?	No
9	A1 after B1 before C1	Is A1 before C1?	No
10	A2 after B2 after C2	Is C2 before A2?	Yes
11	A3 after B3 after C3	Is A3 after C3?	Yes
12	A4 after B4 after C4	Is C4 after A4?	No
13	C1 after B1 after A1	Is A1 before C1?	Yes
14	C2 after B2 after A2	Is C2 before A2?	No
15	C3 after B3 after A3	Is A3 after C3?	No
16	C4 after B4 after A4	Is C4 after A4?	Yes

### Procedure

#### Baseline

For all baseline sessions, participants were exposed to each of the 16 mutual entailment trials, followed by each of the 16 combinatorial entailment trials, for a total of 32 trials per session. During each baseline session, one of the relational statements was presented on the screen (see Table 4.2). The researcher then read aloud the presented relational statement (e.g., "Square is before star") and subsequently asked the participant a question about the relation (e.g., "Is star after square?"). Once the participant responded, the screen was changed to the next trial and the procedure repeated until all trials were presented. During baseline, no reinforcement was provided for correct responses and no error-correction was provided for incorrect responses. A session ended once each of the trials type had been presented.

#### Intervention

Following baseline, participants were exposed to training on mutually entailed relations with only one of the four stimulus sets with each of the 16 different trial types. The presentation of each trial was identical to the baseline condition, apart from programmed consequences for responses. During training, correct responses were reinforced on a continuous schedule of reinforcement (FR1) with specific verbal praise (e.g., "Nicely done! [A] is after [B]!"). and token delivery. Tokens were administered in the form of points, which were given to a virtual avatar selected by the participant in ClassDojo that was visible to the participant via screenshare. Tokens were then able to be exchanged for preferred activity such as a video or game once the session goal criterion was met (Berens & Hayes, 2007; Mulhern et al., 2018). For each session, a goal criterion was designated based on performance from the previous session (e.g., if the total score from the previous session was 9/16 trials correct or 9 points, the goal criterion for the current session was 10/16 trials correct, or 10 points). Following incorrect responses, the researcher gave contingent feedback (e.g., "No, that's not right. "[A] is not before [B], [A] is after [B]") and re-presented the trial.

Each training session continued until the participant responded correctly on each of the unique trial types. Participants progressed to testing of ME and CE relations across all stimulus once training criterion was met (14/16 trials correct across 2 sessions). If no upward trend was demonstrated across five data paths, participants were then exposed to training with only "before" trials (see trial types 1-8 in Table 2). Once criterion was met with "before" cues (7/8 trials correct across 2 sessions), participants were then exposed to training with only "after" cues (see trial types 9-16 in Table 2). Once criterion was met with "after" cues (7/8 trials correct across 2 sessions), participants were then re-exposed to the original trial block consisting of all 16 trial types across both contextual cues.

### **Generalization Probes**

Once mastery criterion was reached with the targeted repertoire, participants were exposed to testing across all stimuli sets for both ME and CE trials. Generalization sessions were identical baseline sessions. No programmed consequences were administered for correct or incorrect responses. A participant was considered to pass a test for ME or CE if they responded correctly on 14/16 trials across 1 session. If a participant passed a test for one repertoire (e.g., ME) but failed the other (e.g., CE), they progressed to intervention with the failed repertoire. If the participant failed the repertoire that was just trained (e.g., ME), they were re-exposed to training at that same level with a new stimulus set (e.g., ME stimulus set 2). Once a participant passed generalization tests for both ME and CE, training ceased, and they progressed to a maintenance condition.

#### Maintenance

Approximately four weeks after training and generalization was completed, participants were tested again with both ME and CE trials. The maintenance session was identical to baseline and generalization sessions.

#### **Interobserver Agreement and Procedural Fidelity**

Interobserver agreement (IOA) and procedural fidelity data were collected by a trained research assistant. The research assistant attended 22% of the online sessions from a different physical location and their kept their audio and visual off during data collection. Both IOA and procedural fidelity data were collected on a trial-by-trial basis. IOA data was calculated by taking the total number of agreed upon trials divided number of total trials and was 99%. For procedural fidelity, trials were scored as correct or incorrect according to a fidelity checklist that included antecedent and consequence presentation for each condition. A procedural fidelity score was calculated by taking the total number of trials implemented correctly divided by the total number of trials and was 100%.

#### Results

Figure 4.1 shows percent correct on ME and CE trials for both participants during baseline, intervention, generalization, and maintenance conditions. During baseline, both participants scored near chance level responding for both repertoires. Following intervention, both participants reached criterion level responding and passed tests for generalization with ME and CE trials.





Note. ME: mutual entailment; CE: combinatorial entailment.

### **Participant 1**

P1 scored 50% correct for both ME and CE trials during each of the three baseline sessions. P1 entered training with ME and required 3 sessions to meet criterion. During the subsequent generalization test, P1 passed the test for ME (94% correct) and failed the test for CE (56% correct). Accordingly, P1 entered training with CE and required 3 sessions to meet criterion. During the

following generalization test, P1 then passed tests for both ME and CE (100% correct each). In the maintenance session approximately four weeks later P1 again scored 100% correct with both ME and CE.

### **Participant 2**

During baseline, P2's average percent correct for ME trials was 51% correct (range 50–56% correct) and 50% correct for CE (no range). Accordingly, P2 entered training with ME. As ME stimulus set 1 was introduced, no upward trend was observed and responding was variable (range 50-88% correct) across 8 sessions. As a result, P2 was then introduced to only "before" trials with the ME stimulus set 1 and required 2 sessions to meet criterion. P2 then was introduced to only "after" trials with ME stimulus set 1 and required 6 sessions to meet criterion. Following this, the standard ME trial set (including both "before" and "after" trials) was reintroduced. P2 required 5 sessions to meet criterion. During the subsequent generalization test, P2 failed tests for ME (63% correct) and CE (50% correct). P2 therefore was reintroduced to ME trianing but with stimulus set 2 and required 8 sessions to meet criterion. During the next generalization test, P2 passed tests for both ME and CE (100% correct each). P2 maintained performance with ME and CE (100% correct each) four weeks later during the maintenance session.

### **Performance across Contextual Cues**

Figure 4.2 shows performance of both participants on trials with "before" as the stated contextual and trials with "after" as the stated contextual cue. The series labeled with "test" include the combined percent correct across both ME and CE with each contextual cue. The series labeled with "train" includes only the targeted repertoire during intervention.

P1's average percent correct on "before" trials during baseline was 98% correct and was 0% correct on "after" trials. During ME training, performance with both cues was identical across each of the three intervention sessions. During the first generalization test, P1 responded correctly on 56% of "before" trials and correctly on 94% of "after" trials. Across the three CE training sessions, P1 scored 75%, 88%, and 100% correct on "before" trials and 75%, 100% and 88% correct on "after" trials. During the second generalization test, P1 scored 100% correct with both contextual cues and this performance maintained post-training.

P2's average percent correct on "before" trials during baseline was 100% correct and was 3% correct on "after" trials. During initial ME training, P2's performance on "before" trials deteriorated across sessions while performance on "after" trials remained variable with no trend. When training was adjusted to include only "before" ME trials, performance returned to baseline levels (100% correct). As training progressed to "after" ME trials only, performance gradually improved from chance level responding to mastery criterion across 6 sessions. As interspersal of both contextual cues was reintroduced, variability in performance was observed across the 5 sessions with both "before" (range 50% to 100% correct) and "after" (range 63% to 100% correct) cues. During the first generalization test, P2 responded correctly on 44% of "before" trials and 69% of "after" trials. As ME

training was re-introduced (stimulus set 2), variability in performance was again observed (range 38%-100% correct with "before" and 50%-100% correct with "after"). However, across these sessions, the data series patterns for each trial type more closely mirrored each other across sessions in contrast ME set 1. During the second generalization test, 2 scored 100% correct with both contextual cues and this performance maintained post-training.

# Figure 4.2

Percent Correct on "Before" Trials and "After" Trials in Each Participant across All Phases



#### Discussion

This research aimed to pilot a testing and training procedure for mutually and combinatorially entailed temporal relations in two, 6-year-old typically developing children. Results showed that following implementation of MET, both participants met training criterion, passed tests for generalization, and maintained performance four weeks post-training.

While baseline performance was similar for both participants, performance during intervention revealed variation in the amount and type of trials needed to meet criterion and pass generalization tests. P1 required only 6 intervention sessions to pass generalization tests for both ME and CE. P2 however, required substantially more learning opportunities (29 intervention sessions) to pass generalization tests for ME and CE. Additionally, P2 required additional training sessions in which the contextual cues were isolated, as well as an additional stimulus set. Although P1 required minimal sessions, she required direct intervention for both ME and CE to pass generalization tests for each repertoire (i.e., training in ME alone did not result in passing a test for CE). P2, in contrast, require more sessions but only needed training on ME to subsequently pass a generalization test for CE.

Results also showed differences in responding on "before" and "after" trials during baseline that this was changed with intervention (Figure 4.2). Baseline responding for both participants was similar to the pattern observed in 6-year-olds by Neufeld et al. (2023). That is, accuracy on "before" trials was at a mastery level for both ME and CE while accuracy on "after" trials was near 0% correct for both ME and CE. The current study extends this work by introducing a training component that revealed that this response pattern was changeable via MET. By the end of testing and training, accuracy with both contextual cues reached 100% correct and maintained post-training. Consistent with other findings, the current data suggest that relational responding in accordance with arbitrary cues can come under antecedent and consequential control.

Another interesting finding in terms of contextual control with "before" and "after" cues is that promoting relational flexibility may be a relevant response dimension to examine more closely in terms of training designs. Relational flexibility refers to the extent to which a given instance of AARR can be modified by the current context (Barnes-Holmes et al., 2020). The more readily a person derives relations during shifting relational contexts, the more flexible the relational response. In the current study, this consideration was seen more evidently in the case of training for P2. Baseline levels were high accuracy with "before" cues and low with "after" cues. Upon introducing MET for ME with both cues, her accuracy with "before" cues specifically deteriorated across sessions. Due to this trend, "before" cues were interspersed during training, two rounds of MET were required to meet criterion. It could be argued that trial blocks involving multiple contextual cues require more relational flexibility than having to respond to a single cue within the trial block, and thus may be more difficult to acquire. It is unclear if training with only one cue and subsequently mixing cues (see

e.g., Berens & Hayes, 2007) would have been any more efficient or not. Future work could more systematically compare training designs in terms of how and when to program for flexibility to optimize skill acquisition with temporal AARR.

Certain limitations of the current study should be noted. Foremost, the current study utilized a nonconcurrent baseline design. Although nonconcurrent designs contain the baseline logic properties of prediction and replication, the verification of intervention effects is excluded (Cooper et al., 2020). The design is unable to control for history effects that may coincide with introduction of the independent variable and thus are arguably weaker than concurrent designs (Ledford & Gast., 2018). Additionally, the current investigated responses with only 2 participants. Future work should utilize a concurrent multiple baseline design with a higher number of participants to increase validity.

Other considerations for future work include specifics relative to the relational training protocol. The present study did not include any testing or training in terms of a transformation of function (ToF) consistent with the derived temporal relations. ToF is present in any derivation of a mutually or combinatorially entailed relation. However, ToF as a concept is distinguished from derived relational responding for practical reasons since a host of different functions can be transformed beyond simply derivation of relations (Belisle, Stanley, et al., 2020; Finn & De Houwer, 2021; Perez et al., 2015). The current study tested for contextually controlled mutually and combinatorially entailed relations (i.e., a ToF in the *relational* context). Future work should include a component to test for a stimulus function transformation with nonarbitrary properties of the environment within a particular *functional* context (i.e., whether behavior would be sequenced in accordance with the given relation).

Additional considerations relative to the training design are also suggested. Sessions in the present study involved 16 trials for ME and 16 for CE. Given the repetitive nature of the task itself, prolonged exposure to this many trials—particularly if many sessions are required—may lead to fatigue for learners at younger ages. It is worth considering whether similar outcomes could still be obtained using a lower number of trials, or in more naturalistic contexts. Second, the current study employed images of commonly known stimuli. Learners may have differing learning histories with each stimulus used and thus may have impacted acquisition. Future work could consider using different stimuli for the relational statements. Lastly, the present work incorporated both audio and visual presentation of the relational networks. That is, they were stated aloud and were shown on-screen. It is unclear whether using this combination of sense modalities facilitated or hindered acquisition.

In summary, the current study revealed an intervention design that can potentially be used for improving performance with arbitrary "before" and "after" relations at different levels of complexity in young children. Future work should replicate applications of this protocol as well as address the present limitations. Continued research is needed in this area and the current data offer encouraging first steps.

Chapter 5.

Training Temporal Relational Framing in Young Children

This chapter has been accepted for publication:

Neufeld, J., Stewart, I., & McElwee, J. (2023). Training temporal relational framing in young children. *Journal of Contextual Behavioral Science*, 28, 81-90. https://doi.org/10.1016/j.jcbs.2023.03.013

Responding to the ways in which events are related in time is important to understanding event sequences and organizing one's behavior. Terms such as "before" and "after" are often used to indicate the temporal order of events, and hence accurately responding to event order requires responding appropriately in the presence of such terms.

A range of cognitive developmental and linguistic studies have explored "before" and "after" comprehension in young children (Blything et al., 2015; Clark, 1971; Keller-Cohen, 1987; McCormack & Hanley, 2011). These studies have typically involved presenting sentence variations describing an event sequence such as, "She brushed her teeth before she read the book"; "After she brushed her teeth, she read the book"; "Before she read the book, she brushed her teeth"; or "She read the book after she brushed her teeth." In one version of a "before/after" comprehension task, children observe a sequence of events (e.g., an animated character brushes her teeth and then reads a book). They then answer questions about the event order, such as by pointing to which event came first or last, or by pointing to the pair of images that matches the order described in the sentence. In a variation, children are instead tasked with acting out the events in the order specified in the sentence (e.g., "Take a blue circle after you take a red circle"; Trosborg, 1982). The data show that performance improves with age and that young children tend to perform better on sentences that state the events in the same order as how they were observed compared to sentences that state the events in the reverse order (Blything et al., 2015; Clark, 1971; Keller-Cohen, 1987; McCormack & Hanley, 2011). A limitation is that these studies mostly emphasized developmental differences and minimal research has investigated arranging conditions to improve performance when a child does not respond coherent with the terms.

One exception is that Amidon and Carey (1972) showed that 5-year-olds' performance on an act-out version of the task can be improved. They found that providing corrective feedback following errors (e.g., "No, that's not right" and model the correct response) improved performance. Additionally, they observed that providing vocal prompts in which the temporal term is stated more loudly did not improve performance. However, the training procedure used was relatively limited in scope and methodological description.

Behavior analytic research investigating the acquisition and training of "before" and "after" comprehension is also scarce. One approach that offers utility in this regard is relational frame theory (RFT; Hayes, Barnes-Holmes, et al., 2001). RFT is a contextual behavioral account of language and cognition that suggests that derived relational responding (DRR) is a key operant at the core of complex human behavior. Humans learn to derive relations between stimuli in various relational patterns (frames), including similarity, distinction, comparison, hierarchy, temporality, etc. Research with children has found that DRR is relevant to important language and cognitive-oriented skills, such as analogy (Carpentier et al., 2002), categorization (Mulhern et al., 2017), and perspective taking (McHugh et al., 2004).

From an RFT perspective, temporal frames are particularly relevant to following instructions involving temporal terms and ordering events in time (Neufeld & Stewart, 2023). Temporal relational framing involves responding to one event as being temporally related to another event. Like all frames, temporal frames are brought under the control of antecedent stimuli (contextual cues) that indicate the relational pattern. In this case, terms like "before" or "after" specify a frame of temporality. Temporal (and other) frames have three key properties. Mutual entailment is demonstrated when taught a relation such as, "A is after B," and then the reverse relation, "B is before A," is derived without further training. Combinatorial entailment is demonstrated when multiple entailed relations are combined such as, "A is after B, and B is after C," and then additional untaught relations are derived such as, "A is after C, or C is before A."

The third property, transformation of stimulus function (ToF), occurs when a function of a stimulus in a relational network is changed or transformed via derivation of relations through that network. Consider an instruction similar to certain "before/after" comprehension tasks (Trosborg, 1982): "Take a blue circle after you take a red circle." A child given this instruction may now take the colored circles in a particular order (i.e., red followed by blue). In other words, the stimuli have acquired sequential functions based on the derived relational network presented. From an RFT perspective, following this instruction involves both equivalence relations (between the words and events or objects) and temporal relations (cues specifying the order). If a child derives all the necessary relations specified in the instruction, they can "understand" the instruction. This may serve as a potential source of influence over behavior even without a direct learning history of the events or sequences described in the instruction.

Indeed, this RFT conception of instructional or rule-based control has been modeled empirically. O'Hora et al. (2004) trained and tested university students using a computer-based relational responding protocol. Arbitrary symbols were first trained as cues for "before/after" and "same/different" relations. After the cues were established, participants were tested for instructional control. This involved the presentation of relational networks including the arbitrary cues, and participants had to press computer keys in the correct sequence specified by the relational network. Across two experiments, 8 of 12 participants demonstrated sequenced responding in accordance with the relational networks involving "before/after" and "same/different" cues. Results provided evidence that sequences of responses can come under relational contextual control.

Later studies expanded this work by examining accuracy and response speed on nonarbitrary ordering tasks involving "before" and "after" (Brassil et al., 2019; Hyland et al., 2012; McGreal et al., 2016) For instance, McGreal et al. (2016) and Brassil et al. (2019) used a go/no-go task in which adults observed a sequence of stimuli on a computer screen one at a time (e.g., A...B). A statement about the sequence with either "before" or "after" then appeared on-screen and participants responded to whether the statement accurately described the sequence (e.g., B AFTER A; press key to affirm
statement; B BEFORE A; no response to reject statement). A consistent finding across studies was that participants tended to respond faster on "before" cue trials than on "after" cue trials.

At present, research involving temporal relations has predominantly been restricted to experimental studies with adults with already well-established temporal framing repertoires and no work has yet been done to train derived "before/after" relations in children. However, RFT considers DRR an overarching, generalized operant (Hughes & Barnes-Holmes, 2016) and accordingly reasons that frames can be established and strengthened via multiple exemplar training (MET). MET involves interchanging a variety of stimuli as the relata, while the contextual cues remain the same. Differential reinforcement is provided for responding and children learn to relationally respond in accordance with the cues. Several studies have demonstrated that MET is an effective method for improving DRR in children across various frames (Barnes-Holmes, Barnes-Holmes, Smeets, et al., 2004; Berens & Hayes, 2007; Dunne et al., 2014; Kirsten et al., 2022b; Mulhern et al., 2018; Weil et al., 2011).

One format through which MET can be realized is the relational evaluation procedure (REP; see Cassidy et al., 2011; Stewart et al., 2004). This procedure requires participants to evaluate arbitrarily applicable relations given particular contextual cues (Whelan & Schlund, 2013). For example, a "relational statement" such as, "TIB is bigger than LEP" might be presented on-screen and participants could be tasked with either confirming or denying related relations (e.g., "Is LEP smaller than TIB?") by pressing either "yes" or "no." Across trials, relational statements could be rotated along with the targeted cues. This type of format is advantageous as it allows for a range of exemplars to be efficiently presented without extended baseline training (Kirsten et al., 2022b).

Studies have shown that this is an effective method to strengthen relational responding in children, and that such training can improve performance on standardized measures of cognitive abilities (Cassidy et al., 2011, 2016; Colbert et al., 2018; Hayes & Stewart, 2016). Many such studies thus far have used words as stimuli, which limits the benefits to children who are literate. Recent work has however successfully adapted this type of procedure for use with children who have yet to acquire a reading repertoire (Kirsten & Stewart, 2022; Kirsten et al., 2022a, 2022b). In this approach, colored shapes were used as stimuli instead of nonsense words, and single letters were used for contextual cues. For example, a relational statement of [red circle] S [blue circle] is displayed, which can be interpreted as, "red is the same as blue" and questions about the relational network can be asked.

The aim of the present study was to expand research on relational framing in children by training temporal frames with "before" and "after" cues in 5-year-old children using MET. A protocol similar to Kirsten et al. (2022a, 2022b) was used to train arbitrary mutually and combinatorially entailed relations. Additionally, an adapted version of the O'Hora et al. (2004) protocol was used to examine ToF in accordance with temporal relations.

#### Method

### **Participants and Setting**

Three children participated. P1 was a female aged 5 years 5 months, P2 was a male aged 5 years 2 months, and P3 was a male aged 5 years 2 months. Participants were selected based on prior findings in which (i) this age group failed tests for arbitrary temporal relations (Neufeld et al., 2023) and (ii) children aged 5 years may have difficulty with "before" and "after" but their performance can be improved with training (Amidon & Carey, 1972). All participants were typically developing and attended schools in the west of Ireland. Ethical approval was first obtained from the research ethics committee at the host institution of the lead researcher. Parental consent and individual assent were obtained for each child.

All sessions were held online through Zoom at convenient times for the families via a networked device in their household. The primary researcher administered all sessions. Prior to each session, an accompanying family member assisted with ensuring all audiovisual features functioned properly. At the session onset, the family member did not remain on-screen but remained in a nearby room in their home. Sessions were held up to 3 times weekly, with each session lasting between 15-20 mins.

### **Experimental Design**

A combination multiple probe design across participants and responses was used for this study. A multiple probe design was selected to reduce threats to internal validity and so that participants were not exposed to undesirably prolonged baseline conditions. Due to the number of trials and difficulty of the tasks, repeated testing could lead to undesirable behavior (Gilroy et al., 2015; Ledford & Gast., 2018). Participants were first exposed to baseline conditions in which all repertoires were tested: mutual entailment (ME), transformation of function via mutual entailment (ToF-ME), combinatorial entailment (CE), and transformation of function via combinatorial entailment (i.e., P1) entered intervention with one repertoire (i.e., ME) while the other repertoires remained untrained, and the two other participants remained in baseline. For P2 to enter training, P1 had to meet the training criterion with the targeted repertoire and for P3 to enter training, P2 had to meet the training criterion. For each participant, once training criterion was met with one repertoires remained in baseline (e.g., CE, ToF-CE).

# Materials

Session materials were displayed from a 13-inch MacBook Pro running Microsoft PowerPoint via the screen share feature of Zoom. Individual stimuli included images of colored circles, and the letters "B" or "A" representing the contextual cues, "before" and "after," respectively (Figure 5.1). For ME and ToF-ME trials, two colored circles with a contextual cue between them were used as the relational statements. For CE and ToF-CE trials, three colored circles with contextual cues between them were used as the relational statements. One set of four different colors was used for baseline and generalization sessions (red, blue, yellow, green) and a separate set of four different colors was used for training sessions (purple, black, orange, white).

Images of simple motor actions were also included for ToF trials. For all ToF-ME and ToF-CE trials, the left side of the screen showed each color-action combination, and the relational statement was presented on the right of the screen (Figure 5.2). One set of four actions was used for baseline and generalization sessions (clap, raise arm, touch nose, wave) and a different set of four actions was used for training sessions (thumbs up, cross arms, open mouth, touch head).

# Figure 5.1

Example Relational Statements



*Note.* The left side of the figure show a mutual entailment trial with a "before" contextual cue (top left) and "after" contextual cue (bottom left). The right side of the figure shows a combinatorial entailment trial with a "before" contextual cue (top right) and "after" contextual cue (bottom right).

#### Figure 5.2

Example Stimuli Arrangement for Transformation of Function Trials



*Note.* The figure shows a sample stimulus arrangement on a transformation of function-combinatorial entailment trial (ToF-CE) with "before" contextual cues.

### Measurement

Accuracy of responding was measured on trial probes across each repertoire (Table 5.1). For both ME and CE, a response was scored as correct if the participant gave a "yes" vocal response when the answer was yes, or a "no" vocal response when the answer was no. Scores were totaled out of 8 and converted to a percentage. For ToF-ME and ToF-CE trials, a response was scored as correct if the participant demonstrated the motor actions in sequence consistent with the order specified by the relational statement. For both ToF-ME and ToF-CE, there were four trials with "before" as the cue and four trials with "after." Scores were totaled out of 8 and converted to a percentage.

### Table 5.1

Relational Level	Relational Statement	Question/Instruction	Correct Response
Mutual Entailment	A before B	Is A before B?	Yes
		Is B before A?	No
		Is A after B?	No
		Is B after A?	Yes
	A is after B	Is A before B?	No
		Is B before A?	Yes
		Is A after B?	Yes
		Is B after A?	No
Transformation of	A before B	Do these in order.	Action A then B
Function (ME)	A after B	Do these in order.	Action B then A
Combinatorial Entailment	A before B before C	Is A before C?	Yes
		Is C before A?	No
		Is A after C?	No
		Is C after A?	Yes
	A after B after C	Is A before C?	No
		Is C before A?	Yes
		Is A after C?	Yes
		Is C after A?	No
Transformation of	A before B before C	Do these in order.	Action A then B then C
Function (CE)	A after B after C	Do these in order.	Action C then B then A

### Trial types for Each Relational Response

### Procedure

### Screening and Pre-Training

Participants were first screened on their ability to a) tact all stimuli used in the study, b) answer yes/no questions about the stimuli, c) perform one-step and multi-step motor actions that were included on ToF trials, and d) answer questions about nonarbitrary temporal relations ("Was A before/after? B" when shown images in sequence). Participants were required to score at least 90% correct on the tacting, yes/no, and motor actions tasks, and at least 75% correct on the nonarbitrary relations task. All participants passed the screening.

Following screening and prior to baseline, participants were introduced to the contextual cues, "B" and "A." The researcher displayed the letter "B" on-screen, stated that "B means before" and then asked the participant, "What does 'B' mean?" The same procedure was repeated with the contextual cue, "A." Participants were then shown a sample relational statement with each of the cues, such as [blue circle] A [green circle]. The researcher "read" the relational statement out loud and then asked the participant to read the same statement aloud.

Next, the researcher trained a discriminative function (for a particular motor action) with each of the colored circles used on the ToF trials. Participants were first shown one of the four colored circles on the screen, along with an image of a particular action (top row of Figure 5.3). The researcher stated, "This [color] means [action]" while also demonstrating a model of the action. This was repeated with the three remaining colors. Participants were then taught to demonstrate the corresponding action when shown the color (alongside a visual depicting each color-action combination, bottom row of Figure 5.3). For instance, the researcher presented the color green, and issued the instruction, "Do this one." Social praise was delivered for a correct response and corrective feedback was given for an incorrect response. Each participant was required to respond with the correct motor action with each of the four colors before proceeding. All participants responded accurately with each color-action combination during initial training.

### Figure 5.3

Example Flow of Presented Stimuli During Color-Action Pre-Training



*Note:* Each box represents a separate display on-screen. The order of presentation reads left-to-right on the top row and then left-to-right on the bottom row.

#### Baseline

For all baseline sessions, participants were exposed to each of the 8 ME trials, then 8 ToF-ME trials, followed by 8 CE trials, and then 8 ToF-CE trials, for a total of 32 trials per session. During

each baseline session, one of the relational statements was presented on-screen at the onset of a trial and then the researcher read aloud the presented relational statement (e.g., "Red is before blue"). The participant was then asked a question about the relation (e.g., "Is blue after red?") during ME and CE trials. During ToF trials, participants were instructed to perform the actions in the order specified by the relational statement (i.e., "Do these in order"). Once the participant responded, the screen was changed to the next trial and the procedure repeated. During baseline, no feedback was provided for correct or incorrect responses. A session ended once each of the trials was presented.

### Intervention

Following baseline, participants were exposed to training in phases: ME then ToF-ME, then CE, then ToF-CE. For all training sessions, trial presentation was identical to baseline, with the exception of programmed consequences for responses. Prior to training, the researcher obtained a list of preferred interests for each participant via caregiver report to establish potentially reinforcing activities. At the onset of training sessions, the researcher asked the participant what kind of activity they would like to do following the teaching activities (e.g., a video or a game). If the participant did not state a preference, the researcher provided a sample of options based on their interests. During training, correct responses were reinforced on a continuous schedule of reinforcement with specific verbal praise (e.g., "Nicely done! [A] is after [B]!") and token delivery. Tokens (points) were given to a virtual avatar selected by the participant in ClassDojo, which was visible to the participant via screen share. Tokens could then be exchanged for access to the selected preferred activity once the session goal criterion was met (Berens & Hayes, 2007; Mulhern et al., 2018). For each session, a goal criterion was designated based on performance from the previous session (e.g., if the previous session score was 6/8 trials correct or 6 points, the goal criterion for the current session was 7/8 trials correct or 7 points). Following incorrect responses, the researcher gave contingent feedback (e.g., "No, that's not right. "[A] is not before [B], [A] is after [B]") and re-presented the trial.

Each training session continued until the participant responded correctly on each of the unique trial types for the targeted relational response. Participants progressed to generalization testing across all repertoires once training criterion was met (100% correct for 1 session or 85% correct across 2 consecutive sessions). It was also determined that if no upward trend was demonstrated across five data paths (Greer & Ross, 2008), participants would then be exposed to training with only "before" cues. Once criterion was met with "before" cues, participants would then be trained with only "after" cues. Once criterion was met with "after" cues, participants would then be re-exposed to the original trial block consisting of trials with both cues. No participant required this additional training phase.

### Generalization

In between each training phase, participants were tested across ME, ToF-ME, CE, and ToF-CE. Generalization probes were identical to baseline in which no programmed consequences were administered. The same untrained stimulus set used in baseline was also used for generalization

probes.

A participant was considered to pass a generalization probe if they scored above 85% correct for that repertoire. Depending on performance during generalization probes, participants continued in one of three possible ways: progress to the next training phase, skip ahead to a latter training phase, or repeat the previous training phase. For example, if the trained skill was ME and the participant passed the generalization test for ME but failed testing with the next untrained skill (i.e., ToF-ME), they proceeded to training at that level. Alternatively, if the participant passed a test for the trained skill of ME and also passed the generalization test for the untrained skill of ToF-ME, they skipped ahead and entered training at the next level that was failed (e.g., CE). Lastly, if a participant was trained in ME but then failed the generalization test for ME, they were then re-exposed to training at the same level. *Maintenance* 

Following completion of all training and generalization testing, each participant was subsequently tested again approximately four weeks later on ME, ToF-ME, CE, and ToF-CE. The maintenance session was identical to baseline and generalization sessions.

### **Procedural Fidelity and Interobserver Agreement**

Procedural fidelity and interobserver agreement (IOA) data were collected by a trained research assistant for each participant across each condition. The research assistant attended the sessions from a different physical location and kept their audio and video off during observations. Procedural fidelity and IOA data were collected in 30% of sessions. Procedural fidelity was assessed on a trial-by-trial basis alongside a fidelity checklist, in which a trial was scored as correct or incorrect. For a trial to be scored as correct, all criteria had to be adhered to according to the condition and trial type, including trial presentation and consequences administered. Procedural fidelity was found to be 99%. IOA data was calculated on a trial-by-trial basis by taking the number of agreed upon trials divided by the number of total trials for a given session; IOA was 99%.

#### Results

#### **Overview**

Results for all participants are shown in Figure 5.4, which displays percentage of trials correct for each repertoire: ME, ToF-ME, CE, and ToF-CE for baseline, training, generalization, and maintenance conditions. An analysis of baseline data indicated no participant demonstrated the target relational repertoire across consecutive sessions. When MET was introduced, each participant met training criteria for the targeted relational response(s) and passed tests for generalization. As this study involved multiple components, we will first discuss results for each participant for clarity.

# Participant 1

In baseline, P1's average percent correct for ME was 56% correct (range 38–88% correct); ToF-ME was 47% correct (range 38–50% correct), and both CE and ToF-CE were 50% correct in each session. During the first baseline session, P1 showed a high level of accuracy on ME trials (88% correct), however this stabilized to near chance levels during the subsequent three baseline sessions.

Accordingly, P1 entered training at the ME level. P1 required eight sessions to reach mastery criterion (100% correct), and subsequently passed the generalization test for ME (100% correct), as well as for the untrained ToF-ME (88% correct). During this generalization probe, P1 failed tests for CE (75% correct) and ToF-CE (37% correct). As a result, P1 then entered training for CE and required five sessions to meet criterion (two consecutive sessions at 88% correct). P1 subsequently passed a generalization test for ME, ToF-ME, and CE (each 88% correct) but failed the test for the untrained skill of ToF-CE (50% correct). Following four training sessions for ToF-CE, P1 met criterion (100% correct) and passed generalization testing for all responses. During the maintenance session, P1 maintained three of the four the relational responses ( $\geq$  85% correct), with ToF-CE returning to baseline level (50% correct).

# **Participant 2**

P2's average percent correct in baseline for ME was 53% correct (range 50–63% correct), ToF-ME was 50% correct (no range), CE was 45% correct (range 38–50% correct) and ToF-CE was 50% correct (no range). P2 entered training for ME and required two sessions to meet criterion (100% correct). P2 passed the generalization test for ME (100% correct), while the other untrained responses remained near baseline level. Training proceeded to ToF-ME, and P2 required three sessions to meet criterion (100% correct). P2 subsequently passed a generalization test for all responses, including the untrained responses for CE and ToF-CE. During the maintenance session, P2 scored 75% correct on ME and ToF-ME, 88% on CE, and 100% on ToF-CE.

### **Participant 3**

For P3, his average percent correct in baseline for ME was 56% correct (range 50–63% correct), ToF-ME was 46% correct (range 38–50% correct), CE was 56% correct (range 50–63% correct), and ToF-CE was 47% correct (range 38–50% correct). P3 also began training at the ME level. P3 required four training sessions to meet criterion (100% correct) and passed the generalization test for ME (100% correct) as well as for the untrained ToF-ME (88% correct). During this test, P3 failed generalization probes for CE and ToF-CE (each 50% correct), so intervention progressed to training CE. P3 required three sessions to meet the training criterion (100% correct) but did not pass the generalization test for CE (75% correct) or ToF-CE (63% correct), while performance with ME and ToF-ME remained at 100% correct. Accordingly, P3 was then re-exposed to training at this level and required two additional training sessions to meet criterion with CE trials (100% correct). On the subsequent generalization test, P3 scored 75% correct on ME and 100% correct on each of the other relational responses, including the untrained ToF-CE. During the maintenance session, P3 maintained a high level of accuracy for all relational responses (100% correct on ME, ToF-ME, and CE, 88% correct on ToF-CE).

# Figure 5.4



Combined Multiple Probe Design Across Participants and Responses

*Note*. ME: mutual entailment; ME-ToF: transformation of function of mutually entailed relations; CE: combinatorial entailment; CE-ToF: transformation of function of combinatorially entailed relations.

# Figure 5.5

Participant Responses During Baseline and Testing Sessions



*Note*. All Trials: includes all responses with both contextual cues across all skills. Before Trials: includes only responses on "before" trials across all skills. After Trials: includes only responses on "after" trials across all skills.

#### **Performance Across All Trials**

Performance across all responses on baseline and testing sessions for each participant is shown in Figure 5.5. That is, the "all trials" data series depicts the overall percent correct across ME, ToF-ME, CE, and ToF-CE trial probes. Each participant's overall performance in baseline remained near 50% correct responding (averages of 51% correct for P1 and P3, 49% correct for P2). As training in the relational responses was introduced, performance improved in each participant and as subsequent responses were trained, overall accuracy improved. Accuracy across all relational responses slightly decreased post-training in P1 and P2 but remained at a high level (84% correct for both participants) while P3's accuracy across all trials was highest during post-training (97% correct).

# Performance on "Before" and "After" Trials

Figure 5.5 also shows overall performance for each participant on trials with "before" as the contextual cue in the relational statement compared to trials with "after" as the contextual cue in the relational statement. During baseline for each participant, accurate responses on "before" trials were consistently and substantially higher than "after" trials. P1 averaged 81% and 21% correct on "before" and "after" trials, respectively; P2 averaged 92% and 7% correct on "before" and "after" trials, respectively; and P3 averaged 83% and 20% correct on "before trials" and "after" trials, respectively. As training was introduced and progressed for each participant, performance on "after" trials significantly improved. During the last generalization session for each participant, P1 scored 100% and 88% correct on "before" and "after" trials, respectively; and P3 scored 94% correct on trials for both contextual cues. During the maintenance session, P1's performance maintained at 100% correct for "before" and "after" trials and decreased to 69% correct on "after" trials. P2 scored 88% and 81% correct on "after" trials.

#### Discussion

This study aimed to train temporal relational framing based on "before" and "after" cues across multiple levels of complexity in three, typically developing 5-year-olds. Results showed that following intervention, each participant met training and generalization mastery criteria for each relational response, and that overall accuracy maintained four weeks later. The present data also showed that each participant had greater difficulty with "after" trials compared to "before" trials at baseline. Following training, each participant demonstrated contextually controlled relational responding in accordance with both cues.

The present study adds to previous research showing that DRR is trainable through MET. Although this has been demonstrated with a variety of other relations (Barnes-Holmes, Barnes-Holmes, Smeets, et al., 2004; Berens & Hayes, 2007; Dunne et al., 2014; Mulhern et al., 2018; Weil et al., 2011), this is the first study to train temporal frames in children. Prior studies have shown that temporal relational responding can come under arbitrary contextual control, but such work has been

limited to adult participants (Brassil et al., 2019; Hyland et al., 2012, 2014; McGreal et al., 2016; O'Hora et al., 2004, 2005, 2008). Further, the present study adds to the literature showing that the REP is a useful format that can be adapted for use with young children (Kirsten et al., 2022a, 2022b).

Results also highlighted different training needs across participants. P1 required direct training on ME and ToF-ME in order to pass generalization tests for these responses. Once ME and ToF-ME were trained, P1 subsequently passed a generalization test for CE without requiring training, but still required direct training for ToF-CE. For P2, once ME was trained, he still required directing training for ToF-ME. Once both ME and ToF-ME were trained, he then passed generalization tests for CE and ToF-CE without direct training. P3 was similar to P1 in that after ME training, he passed a generalization tests for ToF-CE in order to pass generalization tests. Nonetheless, in the case of all participants, MET was effective for training each relational response when needed.

There are at least two implications of this result. First, it suggests that testing each relational response is an important consideration in interventions targeting derived relations. One tool that offers utility in this regard is the hyperdimensional multilevel (HDML) framework, which distinguishes between increasingly complex levels of relational responding (see Barnes-Holmes et al., 2020). The current study examined two types of derivation that are posited as separate levels of the HDML, that is, ME (Level 1) and CE (Level 2). Thus, these types of derivation are sufficiently different from each other such that, for example, even if ME has been acquired, this might not be a robust indicator of the likelihood of soon thereafter acquiring CE. The current data provided some indication of this separation for two of the three participants for whom training in ME alone did not lead to demonstrations of CE. At the same time, it is important to note that within the HDML separate levels of relational behavior are not necessarily "stages" but rather, lower levels of relating may provide a relevant historical context for the higher-level patterns of relating (Barnes-Holmes et al., 2020, p. 609). In the case of P2 who did not require direct training in CE, it is possible that this response was already at least partially established in his repertoire and that training with arbitrary "before" and "after" cues during ME facilitated CE responding as well.

Second, it should not be assumed that if an individual is trained in particular patterns of DRR, that a certain transformation of function in terms of nonarbitrary stimulus properties will always occur. Belisle, Stanley, et al. (2020) suggested that in some RFT studies, ToF with nonarbitrary stimulus properties is assumed to occur but may not be warranted. As the authors noted, without this ToF effect in terms of the physical environment, there is no referential meaning of the established arbitrary relational networks. To that end, we sought to test for ToF throughout the study. Results showed that even when particular relational responses were established, participants on occasion still required direct training to show corresponding ToF. P2 required training on ToF-ME even after ME training, while P1 and P3 both required training on ToF-CE after CE training. However, in some circumstances, training in arbitrary relations was sufficient for ToF to also appear. Future research

Chapter 5

should further investigate demonstrations of ToF across patterns and levels of DRR, and incorporate other stimulus functions, particularly in young populations in whom such patterns of responding may still be emerging.

Another key finding was that clear differences in responding were observed between "before" and "after" cues prior to intervention, and that this was changeable via MET for all participants. Each participant showed high accuracy on "before" trials and low accuracy on "after" trials in baseline, which is consistent with prior findings (Neufeld et al., 2023). Accuracy on "after" trials significantly improved only when training was introduced. While the improvement in performance on "after" trials was more pronounced, improvements on "before" trials were also observed but to a smaller degree given the already high level of accuracy in baseline. An at least partially previously acquired repertoire of relationally responding in the presence of "before" cues might have accelerated the acquisition of the complete repertoire of "before/after" relations. This may have provided a foundation upon which the latter might be built, in contrast to a situation in which a participant had a low or non-existing repertoire of responding to both "before" and "after" cues. Future work could empirically examine this issue. Overall, this data indicates that DRR in accordance with arbitrary "before" and "after" cues can be facilitated through relational training procedures. This result is consistent with other RFT studies, which suggest DRR is an operant behavior that can be strengthened and come under arbitrary contextual control (Barnes-Holmes et al., 2004; Berens & Hayes, 2007; Healy et al., 2000; O'Connor et al., 2011).

The response patterns observed with "before" and "after" cues might also be analyzed through the HDML framework. Relational responses can be examined across multiple dimensions and particularly relevant here is the dimension of coherence: the extent to which a derived relational response is consistent with other patterns of relating. One possible reason for the greater accuracy on "before" trials is that there is higher coherence than on "after" trials. On "before" trials, there is coherence between the description of a sequence in the relational statement (e.g., "[red] B [blue]") and the actual nonarbitrary appearance of stimuli in time (e.g., hearing or reading, "red" then "blue"). In hearing the statement, the nonarbitrary temporal relation is given. In reading the statement, the reader produces the temporal order based on the given spatial order in the statement (i.e., orienting to stimuli from left-to-right). Even if a child does not have a reading repertoire, they might still have a learning history in which a linear arrangement of stimuli will evoke orienting to the stimuli from leftto-right. Socioverbal communities represent time in a specific spatial order, such as writing direction (Fuhrman & Boroditsky, 2010), and this preference becomes increasingly pronounced between early childhood and adulthood (Tillman et al., 2022). Because "before" trials offer higher coherence than "after" trials, participants may be more likely to derive the correct answer in the former than in the latter.

The present study also contributes to the wider literature on temporal responding in children. Studies examining "before" and "after" comprehension in children have, from an RFT view, typically

relied on nonarbitrary temporal relations when children are tasked with reporting on an observed sequence of events (e.g., Blything et al., 2015). Additionally, studies have mostly been descriptive in terms of age-level performance with little emphasis on antecedent and consequential variables. The current study addresses both limitations by providing a method to examine responding with "before" and "after" cues at a more symbolic level (DRR) and demonstrating that correct responding could be increased through operant training procedures.

There are limitations to the present study that future research should address. First, results indicated that not all targeted repertoires maintained post-training at a high level in each participant. For P1, her performance on ME, ToF-ME, and CE maintained at a high level of accuracy but her performance on ToF-CE returned to baseline level (50% correct). More specifically, her pattern of responding on ToF-CE during the maintenance probe was 100% correct with "before" trials and 0% correct with "after" trials. For P2 during the maintenance probe, he performed worse on the presumably simpler trials, ME and ToF-ME (75% correct on both), compared to CE (88% correct) and ToF-CE (100% correct). One possible explanation for this performance is that a practice effect may have occurred during the maintenance probe. Since trials were presented in the order of ME, ToF-ME, CE, then ToF-CE, P2 may have been becoming re-familiarized with the tasks that were previously learned. Future RFT research should examine to what degree patterns of DRR maintain over time consistent with intervention versus how those stimuli were related prior to intervention. Additionally, incorporating a testing condition in which the different levels of relational responses are interspersed (i.e., not presented in a particular order) could offer a more robust indication of skill mastery.

A second possible critique of this study is that a more stringent mastery criteria could have been used, such as by incorporating response fluency (Binder, 1996). This consideration may be relevant to the prior limitation regarding maintenance of performance. Training relational responses to a higher level of fluency may have in turn impacted skill maintenance. Including response rate or response latency as measures could also offer greater insight into comparing performance with "before" and "after" cues in children. For instance, existing RFT studies with nonarbitrary temporal relations tasks incorporated a response latency measure, which found that adults tended to respond slower on "after" trials (Brassil et al., 2019; Hyland et al., 2012; McGreal et al., 2016). This could be extended to children and with solely arbitrary relations in future work. It is recommended that future research on relational framing in children consider mastery factors such as fluency, flexibility, and retention of relational responses into their training designs.

Future studies might also consider other variations in terms of stimuli employed or responses examined. For instance, the letters "B" and "A" were used as "before" and "after" contextual cues in this study. This was consistent with the approach of Kirsten and Stewart (2022) in which initials were selected as cues as opposed to the full names (e.g., the text "before" and "after") to make it easier for young children to "read" the relational statements. One consideration in terms of the initials used in

this study, is that children may have a pre-experimental history with typically sequencing "A" before "B" in the context of the alphabet. It might be argued that this could have made it more difficult to learn that "A" represents "after." It seems relatively unlikely that this might have been the case, however, as a function involving the sequencing of potential cues is different from a function involving contextual control over the sequencing of other stimuli. While the sequencing of letters in the alphabet might affect the former, it was the latter that was relevant in the present study. This study also only presented relational statements in a left-to-right spatial direction to simplify the procedure for children and to be consistent with other applications of the REP (Cassidy et al., 2016; Kirsten et al., 2022b). However, certain considerations might be applicable in the case of temporal framing. As children have a gradually acquired learning history of sequencing stimuli in a particular spatial direction with temporal terms like "before/after," arranging stimuli in this way may be a factor in terms of performance. Future research might consider alternative stimuli arrangements to examine any potential differences in responding. Lastly, future work could include measurement of additional responses such as orienting responses toward the stimuli (e.g., eye-tracking), nonlinear relations (e.g., "A before B, B after C"), and/or higher levels of relating described in the HDML framework.

In summary, the present study adds to the limited literature on temporal relational framing. Results from this preliminary work show that contextually controlled patterns of temporal relational framing can be trained in young children using MET. Given the prevalence and importance of timerelated skill sets in everyday life, more work is needed to comprehensively explore the acquisition and training of temporal relational repertoires and how they support more complex skill sets. Chapter 6.

Teaching Nonarbitrary Temporal Relational Responding in Adolescents with Autism

This chapter has been accepted for publication in The Analysis of Verbal Behavior.

Time, as an abstract verbal concept, is intricately woven with language (Hayes, 1992). Early in development, children learn to orient themselves in time and talk about events according to the temporal order in which those events occur (Fivush & Mandler, 1985; Friedman, 1990). As children age, there is a gradual shift towards also using symbolic systems like the clock and calendar to orient themselves in time (Tartas, 2001; Tillman & Barner, 2015; Tillman et al., 2017). The ability to verbally order events in time, and to understand temporal relations more generally, is understood from a relational frame theory (RFT) perspective as temporal relational responding. Temporal relational responding can be approached as a specific pattern of the overarching operant of relational responding: responding to one stimulus based on its relation to another stimulus.

RFT distinguishes between two forms of relational responding: nonarbitrary relational responding (NARR) and arbitrarily applicable relational responding (AARR) (Kenny, Barnes-Holmes, et al., 2014). NARR is a form of relational responding that is based on the physical or formal properties of the stimuli (Stewart, Barrett, et al., 2013). Thus, nonarbitrary temporal relations are primarily based on actual experiential change in the environment from one moment to the next. For example, consider a child that completed a puzzle and then ate a snack. When asked, "What did you do before snack time?" the correct response (puzzle) is indicated by how the events were sequentially experienced in time. In contrast, AARR is a more advanced variety of relational responding based primarily not on the formal or physical properties of related stimuli, but instead on features of the context that specify the relation (Hayes, Barnes-Holmes, et al., 2001). Hence, arbitrarily applicable temporal relations are based on contextual cues that can indicate a temporal pattern (e.g., "before," "after," "earlier," "later") without reliance on direct experience of sequential order. For instance, a child being told that historical event X happened before historical event Y might derive that Y happened after X, without seeing either event actually happen. This ability is key to understanding time at an abstract level, which allows relatively complex levels of planning and organization with respect to events (Neufeld & Stewart, 2023; Chapter 2).

Although NARR is primarily controlled by the physical relations between stimuli, contextual cues are often still involved. A child may first learn to respond to contextual cues applied to nonarbitrary relations. For example, a child may initially learn to respond to contextual cues like "before" and "after" with events they sequentially experience (e.g., brushing one's teeth before going to bed). Once generalized responding has been acquired, relational control with the contextual cues can then transfer to scenarios in which physical properties are no longer key; for example, learning that their new friend's birthday falls before their own. Some research has shown that learning a particular pattern of NARR may emerge earlier in development than its AARR counterpart (Kirsten & Stewart, 2022; Mulhern et al., 2017). Other work has highlighted that training with nonarbitrary relations may also help facilitate acquisition of AARR (Barnes-Holmes, Barnes-Holmes, Smeets, et al., 2004; Berens & Hayes, 2007). These findings are coherent with RFT predictions (Hughes & Barnes-Holmes, 2016). NARR has a similar relational pattern to AARR, but it is concrete as opposed

to abstract and thus simpler and more likely to come first, as well as to provide a useful foundation for AARR.

Most research on nonarbitrary temporal relations thus far has been conducted with neurotypical adults involving sequential responding (Brassil et al., 2019; Hyland et al., 2012; McGreal et al., 2016). As one example, Hyland and colleagues (2012) investigated accuracy and response speed with temporal relations in 20 adult university students. In their two-part study, the relative effects of temporal relational cues, "before" and "after," on temporal order judgements were evaluated. In Study 1, participants were exposed to familiar shapes and completed a temporal relations task to describe the sequence of the shapes. Participants first observed a sequence of two shapes that appeared on a screen one at a time (e.g., circle...square). Next, a "before" or "after" contextual cue appeared on the screen in between a set of shapes on both the left and right side of the screen. Participants had to select the shapes in the order consistent with the contextual cue. For example, if "after" was presented, a correct response was to select square on the left side of the screen and circle on the right side. Results indicated that participants responded significantly faster to "before" statements in comparison to "after" statements. Such findings were replicated in Study 2, in which abstract stimuli were utilized in comparison to familiar stimuli.

To date, two studies have examined temporal relational responding in children with "before" and "after" cues. Kirsten and Stewart (2022) assessed 24 between 3-7 years old on a relational responding test that included a variety of relations and levels of complexity. During the nonarbitrary temporal relations section, participants observed two sequentially presented shapes and answered questions about their order (e.g., "Which one was before/after?" or "Was stimulus 1 before or after stimulus 2?"). During the arbitrary temporal relations section, participants were presented with a relational statement that consisted of colored shapes with a "before" or "after" contextual cue between them (e.g., "[red] B [green]"). Participants were then asked questions that tested for directly trained, mutually entailed, or combinatorially entailed temporal relations. Results showed that across age cohorts, performance on nonarbitrary temporal and opposition relations was lower than that on other nonarbitrary relations (coordination, comparison, hierarchy). Similarly for the arbitrary relations section, derived temporal relations appeared to emerge later than other derived relations, and only the 6-7 years participants scored above chance.

Neufeld et al. (2023) tested 25 children between the ages of 3-8 years old in nonarbitrary and arbitrary temporal relations using a more comprehensive protocol than the one used in Kirsten and Stewart (2022). During the temporal NARR portion of the assessment, children were shown 2- and 3- image sequences of common stimuli on a screen one at a time and answered "before" and "after" questions about the image sequence (e.g., "Was apple after cat?"). A similar method to that employed in Kirsten and Stewart (2022) was used to assess arbitrary relations but involving a greater number of trials. A test for transformation of function was also included in which simple motor actions were assigned to each stimulus in the relational statement and participants had to sequence their actions in

the order specified by the relational statement. For example, when taught that red = clap, blue = raise arms, and yellow = wave and provided a relational network of "red after yellow after blue," participants were to sequence their actions by raising their arms, then waving, then clapping. Results showed that accurate responding on the NARR task appeared to emerge between 5-6 years of age and similar levels of accuracy were observed with both "before" and "after" questions across the age cohorts on the temporal NARR task. In contrast, participants performed notably better with "before" cues than "after" cues on the arbitrary trials and no age cohort fully met mastery criterion with both cues on the temporal AARR section. This points to a deficit in temporal understanding in young children that might help explain other deficits also found at these ages (e.g., such as in planning and organization), and that might be a target of teaching.

Although this work sheds some light on the topic, only typically developing participants have so far been tested for temporal relational responding, so the generalizability of these findings to other populations is not clear. In some populations such as autistic children, there may be delays with respect to learning of these key repertoires and in these cases targeted training of temporal relations may be needed. Previous research has shown that individuals with developmental or intellectual disabilities can demonstrate deficits in various forms of relational responding (Barron et al., 2019; Gilroy et al., 2015; McLay et al., 2013; Moran et al., 2015; Rehfeldt et al., 2007) and it is possible that this might apply to temporal relations also. Outside of the behavior analytic literature, Overweg et al. (2018) also found that autistic children were less accurate overall in comprehending sentences containing the temporal terms "before" and "after" compared with their typically developing peers.

A second limitation of the current literature is that no training approach has specifically improved responding with temporal relations when absent or deficient. While research evaluating and teaching temporal relational responding specifically for children with developmental disabilities has not yet been conducted, research has examined the utility and efficacy of an RFT approach for assessing and teaching other relational skills for autistic children (Barron et al., 2019; Belisle et al., 2016; 2020; Gilroy et al., 2015; Kirsten et al., 2022a; Ming et al., 2018). Hence, the RFT framework may be successful in addressing deficits that children with developmental disabilities may have in temporal relational responding by establishing and strengthening this repertoire through multiple exemplar training (MET).

From this perspective, assessing and teaching temporal relational responding is critical to supporting and boosting children's temporal understanding. The aim of the present study was to evaluate the efficacy of an assessment and teaching procedure based on an RFT framework to establish and strengthen a foundational level of nonarbitrary temporal relational responding in three autistic participants. Consistent with interventions teaching other relations, the present study used a MET procedure to establish this skill in those children showing a relative deficiency and tested for generalization of the skill. This study was also conducted over telehealth. As MET materials for teaching relational responding can often be efficiently presented in a digital format, delivering

instruction in a telehealth format was considered a unique opportunity that has minimally been explored in applied RFT research.

#### Method

### **Participants**

Informed assent and consent were obtained from the three participants and their parents, respectively. Before the study began, pseudonyms were applied instead of their actual names. Participants were three adolescents previously diagnosed with autism by a relevant professional in accordance with the Diagnostic and Statistical Manual of Mental Disorders Fourth Edition (American Psychiatric Association, 2000). Katie (15-year-old female), Anna (17-year-old female), and Jack (18-year-old male) were all enrolled in a special education school in the West of Ireland. All participants were performing at Level 2 on the Verbal Behavior Milestones Assessment and Placement Program (VB-MAPP; Sundberg, 2008) at the time of the present study. Participants communicated primarily using short and succinct sentences. When presented with yes or no questions, the participants responded accurately and often extended their answers to provide further detail. Participants had the ability to report on events of their day or previous times when prompted. None of the participants had received previous assessment for or teaching in temporal relational responding.

The criteria for inclusion in the study included: (a) diagnosis of autism; (b) prerequisite skills including accuracy in answering Yes/No questions; (c) ability to label common items; (d) ability to follow multi-step instruction; (e) failure to score 100% accuracy on an assessment probe for temporal NARR; and (f) an internet connection and equipment required to support telehealth delivery for the duration of the study. Prior to commencement of the study, parents received written information on their and their child's rights as study participants, an information sheet on the procedure of the study, consent and assent forms, and a Note of Privacy Practices in relation to encrypted two-way video calling being used in the study.

#### Setting

All sessions were conducted online via the use of a home-based laptop with a built-in webcamera and microphone linked via Zoom. The researcher followed a set of ethical guidelines for practice with online delivery of interventions adapted from Trepal et al. (2007) in addition to a second independent observer scoring a procedural reliability checklist (Davlin et al., 2011). Participants and their parents were familiar with the use of laptops and video-calling. Sessions were conducted with the participant sitting at a desk in a quiet room in their home. Each participant completed sessions using a laptop provided by their family and each participant additionally wore headphones throughout each session to minimize any potential background noise within their home. For an active participant (i.e., in continuous baseline or training phases), sessions were conducted once daily on weekdays. Session duration ranged from 10-15 min, with longer sessions reflecting more breaks being requested by the participant. For the remaining participant(s), baseline probe data were collected on the application of intervention to the preceding participant. The intervention was conducted during

extended school break and the primary researcher estimated the total duration of participation for each of the participants during this time including both initial assessment and post-training measures to be no longer than six weeks.

### Materials

All materials used during screening, baseline, training, maintenance, and generalization sessions were presented from a 13-inch HP Pavilion I3 laptop on Microsoft PowerPoint presentations via the screen share feature of Zoom. Test and training stimuli included known images of common items (correct labeling of such stimuli was a prerequisite skill included in the pre-test). The images of common items were categorized into animals, household objects, colors, personal items, and leisure activity materials. For baseline and training sessions, 50 stimulus exemplars were used (10 images per category). For generalization sessions, twenty novel exemplars were used. Ten of the novel exemplars were used for the first generalization session (2 images per category) and the other 10 novel exemplars were used for the second generalization session (2 images per category). Exemplars were of uniform size on PowerPoint 5 x 5 cm and were enclosed in a black square. Exemplar pairings were randomized prior to each session by the experimenter shuffling the slide deck into a novel order.

#### **Dependent Measures**

In baseline, intervention, post-training, and generalization phases, the dependent variable was the percentage of correct responses on nonarbitrary temporal relational responding trials. There was a total of 30 trials in each session, consisting of two different categories of trial-type<sup>8</sup>. Twenty trials posed the question "What was before/after?" On this trial-type category, a correct response was scored if the participant vocally tacted the name of the stimulus that was either before (10 trials) or after (10 trials) the other stimulus as required. The remaining 10 trials consisted of a question such as, "Was the house before the swing?" or "Was the cat after the shoe?" to which either a "Yes" or "No" response was required (5 trials with "yes" as the correct answer and 5 trials with "no" as the correct answer). A correct response was scored on this category of trial type if the participant gave a vocal "Yes" response when the relational statement accurately described the observed image sequence or a vocal "No" response when the relational statement did not describe the observed image sequence of prompts. Trial order was varied in each session. The specific order was decided by the researcher prior to the onset of each session to ensure that the correct number of trials were presented per trial-type.

#### **Interobserver Agreement and Procedural Fidelity**

<sup>&</sup>lt;sup>8</sup> The rationale for teaching multiple response forms was to help facilitate a more flexible relational response repertoire. While teaching the core relational response pattern is critical, it is also important to promote generalization to different contexts such as various task formats (Barnes-Holmes et al., 2001). Both trial types used were from the nonarbitrary temporal relations section in Kirsten et al. (2022) and incorporating multiple response forms has been suggested in previous RFT research to provide a more thorough indication of the relational repertoire (Neufeld et al., 2023).

Chapter 6

A second independent observer scored responses in 69% of sessions to assess for interobserver agreement (IOA). Trial-by-trial IOA data were calculated by taking the number of agreements and dividing by the total number of disagreements and agreements, then multiplying it by 100. IOA was calculated in 62% of baseline sessions, 60% of MET sessions, and in 100% of posttraining and generalization sessions across participants. The second observer was a teacher who received training from the primary researcher on response measurement for the procedure. The overall agreement across all sessions and participants across each condition was 100%, corresponding with an overall average of 100% IOA. Procedural fidelity data were also collected by an independent observer. The procedural fidelity measure was adapted from Davlin and colleagues' (2011) checklist. The measure included whether the researcher conducted pre-session video conferencing checks, randomized the presentation of exemplars and question type, correctly phrased and delivered questions, followed prompting, reinforcement, and error correction procedures, recorded participant responses, ensured availability of breaks throughout the session by the presence of a break-card beside the participants' laptop, and delivered additional structured breaks during training trials. Procedural fidelity was assessed alongside IOA in the same sessions (i.e., 69% of sessions across conditions) and the researcher was observed to implement the program correctly in 100% of the sessions observed.

# Design

A multiple probe design (MPD) across participants was selected for this study to evaluate the functional relation between MET and skill acquisition (Ledford & Gast., 2018). The target skill in intervention was nonarbitrary temporal relational responding between exemplars. The MPD design utilizes multiple probes for baseline data, in comparison to continuous simultaneous baseline data for each participant (Cooper et al., 2020). This design was selected to minimize threats to internal validity and reduce participant exposure to temporal relational responding assessments (Ledford & Gast., 2018). Lengthy concurrent baselines would have been unnecessary and increase the potential for frustration and/or fatigue (Gilroy et al., 2015). Furthermore, the MPD design was most appropriate to the setting, in relation to resources required to facilitate measurement of continuous baselines (Horner & Baer, 1978). Criteria for moving subsequent participants into the training phase was dependent on the preceding participant meeting mastery criteria of 100% across 2 consecutive sessions.

Prior to the onset of each session, the instructor and a family member checked the internet connection, and audiovisual and screen sharing features to ensure they were fully functional to prevent any disruptions during sessions. The instructor then requested that the family member turn off any other screens in the room that the participant was conducting sessions in if relevant. Once all connectivity and audiovisual functionality was confirmed, the family member remained in the room throughout the session. Family members were instructed to not provide guidance to the participant on how to respond and were asked to aid only in the case of technical support.

### **Temporal Relations Assessment**

Following initial screening and testing for relevant prerequisite skills, participants completed an adapted version of the temporal relational responding assessment from Neufeld et al. (2023) to determine inclusion. The adapted temporal relations assessment included tests for temporal NARR (composed of 18 trials of 2-image sequences and 18 trials of 3-image sequences) and temporal AARR (composed of 28 trials for mutual entailment, 28 trials for combinatorial entailment and 28 trials for transformation of function). If a participant scored 100% correct across all nonarbitrary trials, their participation in the study concluded. The first nonarbitrary section at which the participant failed to achieve 100% accuracy was then set as target skill for intervention (i.e., 2-image sequences or 3image sequences). The termination criterion for the assessment was at the stage at which the participant failed to achieve 100% accuracy.

#### Baseline

Participants were exposed to baseline measurement based on the order of performance on the temporal relational responding assessment, starting with the participant performing lowest overall. During baseline session trials, the experimenter presented a sequence of two stimuli. Each stimulus was presented on screen for 1 s (e.g., an image of a house appeared on screen for 1 s, followed by a blank screen for 1s, followed by an image of a swing for 1 s, followed by a blank screen). The experimenter then asked the participant a question regarding the nonarbitrary temporal relation between the stimuli presented (e.g., "What was before/after?" or "Was the house after the swing?"). Baseline procedures were similar to training trials, with the exception that the participant was not provided with any differential consequences for correct or incorrect responses. Precisely, during baseline, following the participant's response, the experimenter said "Okay," in a neutral tone. A session ended once each of the 30 trials was presented. Baseline measurement continued until three consecutive observations of stable responding or trending in the opposite direction to learning were observed. Non-consecutive baseline probes were collected for remaining participants coincident with the preceding participant(s) moving into the training phase.

#### Intervention

Prior to starting the training sessions, the participant selected a preferred activity to access on completion of the session. As in baseline, the session was composed of the same 30 trials, assessing nonarbitrary temporal relations between two stimuli. During the training phase, procedures of reinforcement, error correction, and prompting were used. If the participant provided the correct answer, immediate verbal praise was delivered, and the participant was awarded a virtual sticker (i.e., an emoji icon appeared on-screen)<sup>9</sup>. If the participant emitted an incorrect response, the researcher

<sup>&</sup>lt;sup>9</sup> Since the study was conducted via telehealth, these icons were included to complement social praise to make the feedback more interactive. We reasoned that when an intervention is delivered in-person, other contextual features like facial expression, etc. may be more salient compared to when delivered on a computer screen. Accordingly, we attempted to provide a visual component to the feedback in addition to the auditory feedback.

said, "Nice effort but let's try it again," and then presented the trial stimuli again while verbally labeling the preceding stimulus as "before" and the following stimulus as "after." The researcher then re-delivered the question. If the participant still did not answer correctly, the researcher presented the trial once again and this time delivered a full-vocal prompt for the correct answer (i.e., the researcher modeled the "yes" or "no" response or stated the name of the image that came before/after). Training continued until they met the pre-determined criterion of 100% correct responding across two consecutive sessions. Following the first participant meeting the pre-determined criteria in training, continuous baseline measurement for the subsequent participant began. This procedure was applied in the case of all three participants. Additional baseline probes for the remaining participant(s).

### Post-Training and Generalization

Each participant entered a post-training phase following mastery in the MET phase. Each participant received two post-training sessions scheduled for approximately two- and four-weeks following training, respectively. Conditions in post-training phase probes were identical to that of baseline with no error correction or prompting provided. Once criterion was met in the MET phase and each post-training probe was completed, generalization was assessed across two novel stimuli sets. Conditions in the generalization phase were almost identical to baseline and post-training with the difference being that novel exemplars were utilized to evaluate generalization of the repertoire. No prompting, error correction, or reinforcement was given in this condition, but instead, the researcher provided a neutral response following each trial as in baseline sessions.

### Results

#### **Temporal Relational Responding Assessment**

Prior to baseline probes and continuous baseline measurements, participants completed a temporal relations assessment to determine their existing level of temporal relational responding. The two-phase assessment was of increasing difficulty and criterion to progress to the subsequent phase was achieving 100% accuracy on the preceding phase. Each participant scored below 100% accuracy on the first stage of the assessment: temporal NARR with 2-image sequences. On this section, Anna scored 39% correct, Jack scored 40% correct, and Katie scored 44% correct. As 100% accuracy was not achieved on this initial stage of the assessment, the remaining assessment sections were not administered, and temporal NARR with 2-image sequences was set as the target skill for each participant.

#### **Multiple Probe Design: Overview**

Performances for all three participants are graphed as percentage of correct responses during baseline, intervention, post-training, and generalization phases is shown in Figure 1. For each participant, stable responding was observed below 50% accuracy across all baseline sessions. Following the introduction of MET, substantial and steady improvements in nonarbitrary temporal relational responding were observed across all participants until mastery criterion was achieved. In

contrast to baseline data, scores greatly increased during intervention with a consistent upward trend reaching 100% accuracy across two consecutive sessions. Similarly, during post-training and generalization probes, performance for all three participants remained at mastery levels demonstrating that performance generalized to novel exemplars and maintained post-training.

# **Individual Participant Data**

### Anna

Anna demonstrated low levels of accuracy during baseline measurement scoring on average 36% correct (range 33-39% correct), as shown in Figure 1. Fifteen sessions were required for Anna to reach the mastery criterion of 100% correct across two sessions for the target skill. Post-training phase was entered following mastery being achieved; Anna continued to exhibit high levels of accuracy during post-training scoring 100% correct across both probe sessions. The difference in correct responses between baseline and post-training phases was high, shown in the comparable data trends from baseline (36% average correct) to post-training phases (100% average correct). When tested during the generalization probes, Anna responded at 100% correct across each of the two novel exemplar sets.

#### Jack

Jack also demonstrated low levels of accuracy in nonarbitrary temporal relational responding during baseline measurement scoring an average of 39% correct (range 37-40% correct). Fourteen training sessions of MET were necessary for Jack to meet mastery criteria. There was a large difference in performance, from baseline (39% average correct) to post-training phases (100% average correct) which is demonstrated in the data change. Mastery of the skill was maintained at post-training and generalization was observed on each of the two novel exemplar sets, with Jack scoring 100% correct in both phases, respectively.

# Katie

Baseline probe data were collected for Katie when each of the two preceding participants entered the intervention phase. During baseline measurement, Katie scored on average 39% correct (range 32-43% correct) Katie required 13 training sessions to meet mastery criterion. During the post-training phase, Katie scored 100% correct across both sessions. Katie subsequently passed tests for generalization, scoring 100% correct with both novel sets of stimuli. The magnitude of difference in accuracy from baseline to post-training phases was high, from 39% correct in baseline phases to 100% correct in generalization probes.

# Figure 6.1

Percentage Correct on Nonarbitrary Temporal Relations Trials across All Phases for All Participants



#### Discussion

The present study assessed and trained nonarbitrary temporal relational responding in three autistic adolescents. RFT research has suggested that relational responding deficits may be present in autistic individuals relative to their neurotypical peers (McLay et al., 2013; Rehfeldt et al., 2007). The present study focused specifically on nonarbitrary temporal relations, which may be considered a key foundational skill for more complex, derived temporal relations. This study successfully trained this skill using multiple exemplar training (MET), thus indicating the potential efficacy of this intervention in this context.

Data were collected prior to and following MET. Each participant maintained low levels accuracy during the baseline condition. As MET was introduced, each participant demonstrated improvements in responding and subsequently met the training criterion. A high level of accuracy was maintained in each participant two- and four-weeks following completion of training. Participants were then tested for generalization, and each showed a high level of performance with two sets of novel stimuli exemplars. In summary, performance to the mastery criterion was achieved, maintained, and generalized by all three participants following completion of the MET phase for nonarbitrary temporal relational responding.

The results of the present study extend existing research suggesting that MET is an effective method of establishing and strengthening relational responding (Belisle, Stanley, et al., 2020; Berens & Hayes, 2007; Kirsten et al., 2022a; Ming et al., 2018; Mulhern et al., 2018). This research also provided the first empirical evidence for MET as an effective intervention for teaching temporal relational responding to autistic individuals. Specifically, this approach was effective in teaching three autistic participants to respond to nonarbitrary temporal relations between various exemplars with "before" and "after" contextual cues. As research to date has minimally investigated training temporal relations in autistic individuals, these data have important implications. From an RFT point of view, establishing and strengthening temporal relational responding in autistic children should ultimately result in substantial benefits as it is a fundamental skill which underpins a range of functional living and social skills including sequencing, talking about the past and future, and time management. These more complex repertoires are pivotal across personal, academic, and occupational settings. First learning to relationally respond to actual event sequences in the presence of "before" and "after" contextual cues represents an arguably important foundational skill to building these more complex repertories.

The current study furthers existing research on temporal relational responding in two primary ways. First, existing research on temporal relational responding has thus far only included neurotypical children (Neufeld et al., 2023), and has predominantly been limited to work neurotypical adults (Brassil et al., 2019; Hyland et al., 2012; 2014; McGreal et al., 2016; O'Hora et al., 2005; 2008; 2014). The present study expands the literature base to include autistic participants with demonstrated deficits of this repertoire. Second, existing research on temporal relational responding has not

Chapter 6

included a training component when this repertoire is absent or deficient. While there has been research using MET to train autistic children on other nonarbitrary relations, such as nonarbitrary sameness (Belisle, Huggins, et al., 2020; Toussaint et al., 2017) and hierarchy (Ming et al., 2018), no work has been done with temporal relations. The present study addresses this limitation by specifically demonstrating the effectiveness of a MET approach to training nonarbitrary temporal relations. Considering that mastery of abstract time concepts unfolds over several years in childhood (Friedman, 2000; Friedman & Kemp, 1998; McCormack & Hoerl, 2017; Moore et al., 2014; Pyykkönen & Järvikivi, 2012; Tillman et al., 2022), developing effective training approaches for populations that may have difficulty acquiring these skills is critical. Further research is warranted in this area to support the findings of the present study and to refine the efficacy of the training protocol.

Additionally, the present study expands the literature on nonarbitrary relational responding in humans more generally by providing initial data on nonarbitrary temporal relations. Research on responding to nonarbitrary stimulus relations in nonhumans has accumulated over several decades (Flemming et al., 2007; Ghirlanda et al., 2017; Harmon et al., 1982; Thompson & Oden, 2010; Truppa et al., 2011). However, research with humans on nonarbitrary relations beyond sameness has not necessarily been as extensive (Ming & Stewart, 2017). Considering the potential relevancy of the acquisition of nonarbitrary relations to more complex verbal repertoires, more work is needed not just with temporal relations but other nonarbitrary relations as well. Although some behavior analytic curricula, such as the Promoting the Emergence of Advanced Knowledge (PEAK) Relational Training System (Dixon, 2015, 2016) have begun incorporating nonarbitrary relational responding into their training designs, more empirical work in this area is needed.

The present study also administered the intervention via telehealth which is an increasingly popular method for behavioral service delivery. Telehealth and online learning platforms have been found to reduce the barriers autistic children face between intervention requirements and the availability of resources and intervention (Vismara et al., 2013). Prior to the global pandemic, during which the current study was conducted, research was beginning to highlight the need for telehealth to be utilized in greater abundance in providing behavior analytic interventions (Ferguson et al., 2019), clinical assessments (Dahiya et al., 2020), professional and parent training for autistic children (Wainer & Ingersoll, 2011), and supporting behaviours of concern (Lindgren et al., 2016). This study supports findings of other applied behavior analytic research which has used telehealth for the delivery of various skill teaching assessments and interventions (Gibson et al., 2010; Neely et al., 2017). As teaching and learning has been shaped indefinitely by safety procedures in place following the pandemic, research utilizing telehealth successfully with positive treatment outcomes is fundamental to the progression of the field.

### **Limitations and Future Research**

One limitation of the study is that the degree to which the skill generalized to everyday language and related repertoires was not assessed. As the purpose of this study was to provide

preliminary insight into temporal relational responding in autistic participants and whether MET would be effective in training this skill, generalization was only assessed across novel exemplar sets. The extent to which the skills strengthened in the current study generalized to the participants' everyday lives is unknown, yet the fact that generalization occurred across novel untrained stimuli sets provides encouragement for further research. Subsequent studies should include additional tests for generalization and examine whether improvements in this skill relates to other relevant repertoires (e.g., sequencing tasks, following multi-step directions). Such work is critical to expanding our knowledge regarding temporal relational responding and its relation to time understanding.

Future research could also explore additional generalization considerations, such as assessing the skill with novel instructors or in other settings. Although the current study was delivered in an online format, its structure is transferable to in person teaching also and a comparative study examining the two methodological approaches may be insightful with relation to the generalizability of the findings. Although generalization of the target skill was achieved across two novel exemplar sets in the current study, inclusion of an additional condition with a novel instructor would have been beneficial as generalization across new instructors could lead to benefits beyond the academic setting over time. However, due to the online delivery of the intervention and restrictions due to the global pandemic, training and delivery of generalization conditions by a novel instructor was not possible.

Another direction future research could take involves expanding the training program to include shifting contextual control to arbitrary temporal relations. The present study did not include arbitrary temporal relations, as the primary aim was to assess and train the more foundational skill of temporal NARR. However, learning to derive arbitrary temporal relations is particularly relevant to more complex verbal repertoires such as rule-following, problem solving, and responding to delayed consequences. This could be included in training for autistic populations. Studies involving other patterns of relational responding have shown the benefits of using nonarbitrary relations to acquire more abstract relations (Barnes-Holmes, Barnes-Holmes, Smeets, et al., 2004; Berens & Hayes, 2007). It is plausible that this may apply to temporal relations as well, but this should be explored empirically.

Lastly, future research could expand on the current study by including other demographics, forms of measurement, and additional contextual cues. The present study included only three autistic adolescents. This training could be replicated with autistic children of varying ages to examine the utility of the intervention more closely in other contexts. Future work should also include a measure of social validity to assess significance and acceptability of the training procedures. Additionally, the current study only used "percent correct" as a dependent measure. Existing studies involving nonarbitrary temporal relations with adults have also included response latency, which have shown that adults are faster on "before" compared to "after" trials (Hyland et al., 2012). As the current study did not include response latency, it remains uncertain whether the participants a) responded in a similar pattern, and b) if training affected this dimension of behavior. Future research could replicate

Chapter 6

the training protocol and additionally include a measure of response latency. Further, the current study only included total accuracy across the trial blocks. Future research should consider examining any differences in responding based on the response form (i.e., between tacting the stimulus that was before or after vs yes/no responding about a description of the observed sequence). This might add to data concerning young children's acquisition of temporal relations as well as relational responding more generally. Finally, potential future work could incorporate other contextual cues for temporal relations. While the contextual cues included in the present study ("before/after") are important, children encounter a range of other cues, (e.g., "earlier/later", "first/last") in natural environments. Future research should investigate performance with these different contextual cues and how to incorporate them in training designs to expand relational skills in learners.

# Conclusion

The present study complements and extends the results of existing temporal responding research, in that participants were autistic individuals who were assessed and trained successfully in nonarbitrary temporal relational responding with "before" and "after" cues using an RFT based approach. It contributes to the literature on relational responding by suggesting that autistic children exhibit deficits in temporal relational responding and that such deficits can be successfully addressed through MET. Such findings extend support for the position that MET is an effective training methodology for teaching relational responding in neurotypical and neurodiverse children alike. The social validity of assessing and addressing deficits in temporal relational responding for autistic children is substantial as the skill is fundamental to temporally ordering events, and other time understanding skills. The present results are encouraging as they indicate that deficits in nonarbitrary temporal relational responding can be successfully addressed in autistic adolescents. Continued research in this area is needed as such work is critical to expanding our knowledge regarding how the acquisition of verbal repertoires relate to an abstract understanding of time.

Chapter 7. General Discussion The purpose of this research program has been to extend research on temporal responding and more specifically understanding of time, using an RFT approach. Towards this end, this research has identified areas of application of RFT relevant to temporal skill sets; extended research on temporal relations to new demographics; designed an assessment procedure to measure performance on temporal relational tasks; and developed a training procedure that can be used to facilitate the acquisition of these repertoires. In the final chapter of this thesis, I provide an overview of the work that has been done and discuss implications and directions for future work. This chapter first provides an overview of key findings from the literature review and Studies 1-4, followed by the theoretical and applied implications from this research. Following this, a discussion of the limitations and directions for future research is provided.

# **Overview of Thesis**

### Narrative Review (Chapter 2)

Historically, behavioral research has examined temporal responding in terms of responses to various durations in context of different reinforcement schedules. For instance, research has shown that humans demonstrate response patterns that differ from nonverbal organisms on FI schedules of reinforcement (Lowe et al., 1978) and that this difference becomes more pronounced during childhood (Bentall et al., 1985). Other work has shown that providing training with temporal instructions can improve performance with reinforcement schedules (Droit et al., 1990) and timing tasks (Lejeune et al., 1992). However, no behavioral work has specifically examined how verbal abilities alters our experience with time.

RFT suggests that due to AARR, humans additionally experience time in a fundamentally different way. That is, humans experience a "verbal time." Instead of solely discriminating and responding to moment-to-moment changes in the environment, humans verbally create bidirectional sequences about the past or future without having had to experience them directly. One important pattern of AARR relevant to verbally responding to time is temporal relational framing. This frame can be made more precise through quantification, shares overlapping features with certain frames (e.g., comparison, spatial, deictic, causal), and is often involved in forming rules. Conceptually, it reasons that AARR more generally and temporal AARR more specifically are relevant for the acquisition of and performance with time-related skill sets. Chapter 2 sought to identify multiple areas of investigation in which RFT could potentially offer a new or supplemental approach that would advance our understanding of human temporal responding.

Temporal skills range from basic to complex, including skills such as temporal sequencing, timing, using conventional time systems, self-control, planning, and time management. Research with temporal sequencing has revealed gradual improvements with age (Fivush & Mandler, 1985; Friedman, 1990), although a question for continued exploration is to what degree does "before" and "after" understanding facilitate performance on these tasks (McCormack & Hanley, 2011). An RFT approach can augment this work as it precisely pinpoints how to operationalize and train

"understanding" of these words. Research with timing has shown that explicit timing improves with age (Droit-Volet & Coull, 2016), there is an increased reliance on counting strategies as children acquire verbal skills (Wilkening et al., 1987) and changing contextual features can distort duration estimates (Droit-Volet & Gil, 2009). RFT-based work might seek to investigate whether training equivalence relations between quantities and durations improves timing and whether facilitating stimulus function transformations alters duration estimates.

Mastering the use of the clock and calendar systems takes years in childhood and gradual improves have been observed with telling time (Friedman & Laycock, 1989), acquiring duration and deictic time words (Tillman et al., 2015; 2017), and using calendar information to talk about events (Tartas, 2001). One key area for RFT application includes embedding relational learning into clock and calendar instruction, such as by teaching the various ways that conventional time words can be related. Research on self-control has found multiple behavioral approaches that alter responding toward delayed reinforcers (Whiting & Dixon, 2015) and notably, future-oriented manipulations have shown to affect delay discounting (Scholten et al., 2019). RFT provides an approach that could help investigate verbal interventions, such as with forming self-control rules, interpreting future-oriented strategies behaviorally, and transforming the functions of immediate and delayed choices (Dixon & Holton, 2009). Lastly, research on planning has shown age-related improvements on both in-themoment (McColgan & McCormack, 2008) and hypothetical-future planning tasks (Atance & Meltzoff, 2005); while work with time management is scarce. RFT-based work could look at identifying which relational frames are most relevant at different points during planning tasks (Tyrberg et al., 2021), and embed specific relational response training during planning and time management instruction.

### Study 1 (Chapter 3)

Existing research on derived relational responding has primarily examined relations of equivalence (O'Connor et al., 2017). Although investigations of nonequivalence relations have continued to expand over the now 40-year history of RFT, temporal relations have still received little empirical attention. Accordingly, the primary aim of Study 1 was to investigate young children's performance on tasks of temporal relational responding at different levels of complexity. Twenty-five children between 3-8 years were assessed on performance with nonarbitrary temporal relations with 2-and 3-image stimulus sequences, arbitrary mutually and combinatorially entailed temporal relations, and transformation of function in accordance with arbitrary relational networks.

Results from Study 1 highlighted two key findings. One, the data showed that temporal NARR appeared to be acquired earlier than temporal AARR. A high level of accuracy (80% mean accuracy) on the nonarbitrary temporal relations tasks appeared to emerge between 5-6 years of age. In contrast, no age cohort demonstrated this level of accuracy on the presented arbitrary temporal relations tasks. Existing RFT literature has discussed that NARR may be a particularly important precursor skill to AARR (Stewart & McElwee, 2009). Empirically, evidence for this has been found

Chapter 7

with other nonequivalence relations such as comparison (Berens & Hayes, 2007) and hierarchy (Mulhern et al., 2017). The results of the current study provide tentative support that this may be true for temporal frames as well. However, replication is warranted with a larger sample size and additional assessment conditions.

A second notable finding was that there were varying differences in performance observed for NARR and AARR in responding toward "before" and "after" contextual cues. On the temporal NARR tasks, participants generally responded with comparable levels of accuracy with both contextual cues across ages. This was not the case for temporal AARR tasks. Participants regularly responded with greater accuracy on trials with "before" as the displayed contextual cue compared to trials with "after" as the displayed contextual cue for. Further, this finding was consistent across tests for mutual entailment, combinatorial entailment, and transformation of function. Whereas accuracy on arbitrary "before" trials improved with age, accuracy on "after" trials did not.

### Studies 2 and 3 (Chapters 4 and 5)

The aim of Studies 2 and 3 was to develop and refine an RFT-based training protocol to train arbitrary temporal relations (i.e., temporal relational frames) in young children. Multiple studies have shown that MET can be used to improve various patterns of AARR in children when these repertoires are still emerging (e.g., Dunne et al., 2014). However, no existing work had specifically taught temporal AARR with the contextual cues "before" and "after." The data obtained from Study 1 revealed response patterns across different ages on different tasks of temporal relational responding, which informed participant selection for intervention. Specifically, by 5-6 years of age, children tended to respond with a higher level of accuracy on temporal NARR tasks but did not pass tests for temporal AARR.

Study 2 was an initial investigation into the efficacy of a MET intervention for arbitrary temporal relational responding. Guided by findings in Study 1, two 6-year-olds were trained on temporal AARR with mutually and combinatorially entailed relations in a nonconcurrent combined multiple baseline design. The study included a screening of potential prerequisite skills; a baseline condition in which temporal AARR was assessed across multiple stimulus sets; an intervention of the targeted relations with one stimulus set; and generalization and maintenance testing across all stimulus sets. Results showed that both participants demonstrated contextually controlled responding with both "before" and "after" cues with the trained set of stimuli. MET was generally effective for both participants; however, one participant required a training condition in which the contextual cues were isolated from one another, and multiple stimulus sets to reach criterion. Both participants showed generalization to novel stimuli and maintained this performance post-intervention.

Study 3 was an extension from Study 2 and attempted to address some of its limitations. Namely, a stronger experimental design was used, and additional responses (ToF trials) were trained in slightly younger participants. Study 3 employed a combined multiple probe design across participants and responses to train temporal relational framing (mutual entailment, transformation of

function via mutual entailment, combinatorial entailment, and transformation of function via combinatorial entailment) in accordance with "before" and "after cues" in three neurotypical 5-yearolds. Additionally, Study 3 attempted to improve the efficiency of the protocol developed in Study 2 by reducing the number of trials required. A similar MET training approach was used as in Study 2 while the stimuli selected for the arbitrary relational networks (colored shapes) and transformation of function trials (images of motor actions) mirrored the format of Study 1. Baseline, training, generalization, and maintenance conditions across each repertoire were again included. As in Study 2, participants reached training criterion across repertoires following MET and showed generalized responding to novel exemplars. During the post-training condition, not all responses for every participant were maintained at the same level as compared to training (e.g., transformation of function via combinatorial entailment for P1). However, overall accuracy was maintained at a higher level than baseline for each participant.

Taken together, Studies 2 and 3 both showed that responding can come under contextual control with arbitrary contextual cues for temporal relations. In both studies, participants demonstrated mutual entailment and combinatorial entailment following MET. Each participant in Study 3 additionally learned to sequence sets of motor actions in accordance with arbitrary relational networks following intervention (i.e., a transformation of stimulus function). Both Studies 2 and 3 showed a similar response pattern as had been observed in Study 1: a much stronger baseline performance with arbitrary "before" trials compared to "after." Following intervention, a notable increase in accuracy with "after" trials was observed and participants responded in accordance with both contextual cues.

# Study 4 (Chapter 6)

Study 4 sought to extend this line of research to another demographic. Some existing research has identified that individuals with autism may present with deficits in AARR (McLay et al., 2013; Rehfeldt et al., 2007) and that they may have greater challenges understanding temporal terms compared to their same age neurotypical peers (Overweg et al., 2018). Moreover, prior studies have shown that MET has been effective in training relational responses in individuals with autism for both nonarbitrary relations (Ming et al., 2018) and arbitrary relations (Belisle, Stanley, et al., 2020). However, no work prior to this research program had investigated the acquisition of "before" and "after" relations in neurodiverse populations.

In Study 4, a multiple probe design was used to teach nonarbitrary temporal relational responding in three autistic adolescents. Each participant was first assessed using an adapted version of the temporal relations assessment that was used in Study 1. The first nonarbitrary test section in which criterion was not met was the targeted skill for intervention, and in the case of all participants this was temporal NARR with 2-image sequences. MET was used in which a variety of 2-image stimulus sequences were presented, and differential reinforcement was applied for responding in accordance with "before" and "after" cues. The outcomes were that all participants met training

Chapter 7

criterion, maintained performance post-training, and passed generalization tests with two novel stimulus sets.

Although Study 4 is limited insofar as it solely trained nonarbitrary temporal relations, it is still an important contribution. RFT emphasizes that a generalized AARR repertoire is the key operant relevant to complex human behavior but nonetheless, learning NARR across relational patterns is an important building block. Existing research on training procedures for nonarbitrary relations other than sameness (identity) is minimal. Continued work in this area may offer insight into intervention design, scope, and sequence with relational skills. If individuals have yet to acquire these more foundational relational skills (as was the case for participants in Study 4), this would be the logical point of introduction for training. Further, learning to respond to formal temporal relations in the presence of contextual cues used within the learner's socioverbal community may assist with effectively navigating sequences in their environment. Indeed, the distinction between NARR and AARR may become blurrier in naturalistic contexts as instances of relating may often be nonarbitrarily applied. For example, calendars or written activity-schedules that include "before" and "after" cues involve arbitrary stimuli, but these schedules are often tied to real events that are actually sequentially experienced.

#### **Overall Contributions and Implications**

Collectively, the totality of this research program included a review and conceptual RFT analysis of temporal skill sets, and the assessment and training of temporal relational responding repertoires in neurotypical children and autistic adolescents. This section describes the overarching contributions of this thesis to the literature, and the theoretical and applied implications for behavioral and nonbehavioral approaches to the acquisition of temporal relations.

Most notably, the current work addresses a pressing need concerning the lack of research on temporal relations identified in the RFT literature. To put this into perspective, consider the following quotes from RFT textbooks:

"The impact of such verbal temporal relations on how humans interact with their environments is immense." (Hayes, Fox, et al., 2001, p. 60)."

"Though we have acquired substantial evidence of the importance of framing in general to language and cognition, research into framing at a more detailed level is still at a relatively early stage. For example, there remains as yet little or no investigation of some unquestionably important forms of framing (e.g., spatial, temporal, logical or hierarchical) and as yet limited investigation of some others (e.g., conditional, sequential)" (Stewart, Tarbox, et al., 2013, p. 192)."

"Unfortunately, evidence of temporal framing and the history of learning needed to establish it is currently scarce. This class of relations has received far less attention than other frames in the RFT literature, with existing work focused on their implications for intelligence and rule-following, as well as their experimental induction in adult populations...It remains to be seen how these frames are initially established in the natural environment, how the unidirectional experience of change comes to be abstracted, and the bidirectional dimension of time constructed. Given their potential role in suicide (Hayes, 1992) and delayed gratification (Mischel & Ayduk, 2004), for example, closer attention to temporal relations certainly seems warranted" (Hughes & Barnes-Holmes, 2016, p. 153).
"To date, temporality has commonly been combined with other relational frames and has rarely been targeted as a relational frame in isolation within applied research with children and populations for whom this relational repertoire may be deficient...Given the frequency with which temporality is cited within the introduction of published RFT work (and indeed, the importance of the concept of time itself), it is surprising that there remains a dearth of literature which focuses on temporality alone....to date, there is no research which has focused on the facilitation of temporal relations amongst individuals who demonstrate deficits in this repertoire... The lack of published research which focuses on the facilitation of non-arbitrary and arbitrary temporal relations with populations deficient in this repertoire indicates a substantial gap in the area which future research should aim to resolve." (Mulhern, 2023, pp. 131, 145, 149).

A sobering insight from the sequence of these passages is that the importance of temporal frames was noted as early as the initial 2001 RFT textbook, yet decades have gone by with similar statements being issued concerning the minimal progression of work in this area. Prior to this research program, existing work on temporal relational responding had predominantly been restricted to neurotypical adults (Brassil et al., 2019; Hyland et al., 2012, 2014; McGreal et al., 2016; O'Hora et al., 2004, 2005, 2008) and the little work done with children had been relatively limited in scope (Kirsten & Stewart, 2022). The current research extends the literature base on temporal relations, particularly with respect to the acquisition and training of this repertoire in children. Specifically, this research on temporal relational responding extended empirical work to new demographics, provided initial data on potential developmental trends, designed a method for assessment of this repertoire, and developed an effective approach to training. Although this thesis certainly does not comprehensively fill the gaps identified in the above passages, it arguably provides a conceptual and empirical footing in which subsequent research can be built.

The current thesis also made a key contribution to the literature by extending empirical work on relational responding more generally. A core proposition in RFT is that relating is an operant that comes under contextual control. According to RFT, a generalized skill of relating is learned across a variety of opportunities, often occurring in natural language contexts. However, what is less clear, is when certain relational framing skills typically emerge in terms of age or relative to other frames. Some RFT work has begun to reveal insights in terms of general developmental trends with various relational repertoires (Kirsten & Stewart, 2022; McHugh et al., 2004; Mulhern et al., 2017). These types of findings are particularly useful for assisting in the guidance of creating programs designed to improve relational skills. Having data on when certain frames tend to emerge or in what type of sequence can help inform better age-appropriate learning targets for practitioners. The results from Study 1 in this thesis add to the literature in this area as it provides a data set on how children at different ages might respond on temporal relational tasks.

Along these lines, several studies have additionally shown that for individuals that are still acquiring certain relational skills, arranging MET testing and training conditions can lead to improvements (Barnes-Holmes, Barnes-Holmes, Smeets, et al., 2004; Belisle et al., 2016; Mulhern et al., 2018). Although MET has been shown to be effective for improving relational responses, no work had demonstrated this with temporal "before" and "after" relations. Studies 2 and 3 specifically

showed that a MET approach can be effectively extended to training arbitrary temporal relations as well. Even though AARR is considered an overarching repertoire, various relational patterns can still be distinguished from another, such as by the way stimulus function transform based on the relational pattern. As such, it is important to continue to examine AARR across different relational patterns. Nonetheless, instances of relating across different frames should still show a degree of commonality if they are to be conceptualized under a generalized repertoire. The findings from Studies 2 and 3 add to the body of work suggesting that AARR is an operant that can be strengthened and brought under contextual control.

Although RFT primarily emphasizes the importance of AARR, many early relational operants are considered to have nonarbitrary counterparts which provide foundational building blocks to purely symbolic acts of relating (Berens & Hayes, 2007; Ming et al., 2018). To that end, investigating the acquisition of these nonarbitrary relational repertoires may be particularly important, especially in contexts where individuals have significant deficits in relational responding. Refining instructional procedures for nonarbitrary relational repertoires could thereby help fill in gaps in relational skills more effectively and efficiently. Study 4 contributed to the literature by providing an initial data set on training nonarbitrary "before" and "after" relations based on an RFT approach. The findings from Study 4 additionally supported the general finding from Studies 2 and 3 that MET is an effective intervention approach.

An interesting pattern that emerged from this research program was the contrasting performance with arbitrary "before" and "after" contextual cues. As discussed in Chapter 3 (Study 1), children across ages tended to respond comparably with both cues on nonarbitrary trials. However, with arbitrary trials, "after" cues were responded to with much less accuracy than "before" cues. As a reminder to the reader, this pertains to trials in which the relational statement included "before" or "after" (e.g., [red] B [blue] or [red] A [blue]) and not the contextual cue within the question asked (e.g., "Is red before/after blue?"). In Studies 2 and 3, a similar pattern was observed on arbitrary trials, and this only changed with the introduction of MET for all participants.

This finding appears to be somewhat consistent with existing RFT work as well as other psychological research on "before" and "after" comprehension. Existing RFT experimental research has reliably found that neurotypical adults with fully developed relational repertoires still respond slower on "after" trials compared to "before" (Brassil et al., 2019; Hyland et al., 2012; Hyland et al., 2014; McGreal et al., 2016). Although the research in this thesis did not include a response latency measure, there is a similarity in the sense that better performance was observed with responding to "before" statements compared to "after" statements in participants. Secondly, developmental and linguistic studies on "before" and "after" comprehension have typically found that children perform better when given a sentence describing a temporal sequence matches the actual temporal sequence compared to when the description is reversed (McCormack & Hanley, 2011; Trosborg, 1982). Interestingly, this means that it is not simply that children understand "before" better than "after," as

either cue could be included in either matching sentences ("A is before B;" "After A, is B") or reversed sentences ("Before A, is B;" "B is after A"). The correspondence between the description and actual event sequence appears to be a larger factor in terms of performance. This parallels data from the current thesis. If performance was simply a function of participants not understanding "after," then the data would reveal errors whenever an "after" question was asked. However, this was not the case as the discrepancy in accuracy was observed relative to the type of relational statement presented, which is analogous to whether a description matches a temporal sequence or is reversed.

This observed difference between responding to "before" and "after" relational statements can be examined along the dimensions within the HDML framework (Barnes-Holmes et al., 2020). These dimensions include: coherence (How consistent is the relational response with one's learning history?), complexity (How detailed is the relational response?), derivation (How well-practiced is the relational response?) and flexibility (How readily does the relational response change as required in a new context?). A particular relational response may be analyzed along a continuum (e.g., high to low) for each of these dimensions.

The dimension of coherence is relevant in two ways. On one hand, as discussed in Chapter 5 (Study 3), "before" relational statements are more coherent than "after" as the sequence described in the relational statement corresponds with an actual nonarbitrary temporal relation (observing or hearing the stimuli in a particular order). Secondly, responding to "before" relational statements is more coherent with a likely learning history for sequencing events left-to-right. Although access to the participants' specific learning histories is not possible, this point appears relevant given the ages of participants. For example, Tillman et al. (2022) showed that directional preference for ordering temporal events begins to emerge around 5 years, coinciding with a considerable increase in exemplars for ordering in this direction (e.g., reading). As such, the socioverbal community tends to reinforce relating events in the particular way, which is more coherent with "before" relational statements than "after."

The response differences may also be examined in terms of complexity. In this thesis, complexity was manipulated by presenting relations between 2 relata (ME trials) and subsequently increasing the number of relata to 3 (CE trials)<sup>10</sup>. Relational networks involving 3 relata would be considered higher in complexity than those with 2 relata. In Study 1 and baseline of Studies 2 and 3, performances of participants on arbitrary "before" trials were typically higher than arbitrary "after" trials at both levels of complexity. In a few instances however, response patterns changed when complexity increased. During baseline of Study 3 for example, P1 responded more accurately to ME "before" trials than ME "after" trials (average of 81% and 31% correct, respectively); whereas on CE trials, P1 responded "no" to all CE questions presented (hence demonstrating chance level

<sup>&</sup>lt;sup>10</sup> Relational complexity can be defined along multiple dimensions, such as the number of relata, frames, or contextual cues in a relational network (Barnes-Holmes et al., 2020).

Chapter 7

performance). As such, manipulations in complexity may in some instances be relevant to response patterns between contextual cues.

The HDML dimension of derivation may also be relevant, though empirical testing is warranted. As discussed in Study 1 (Chapter 3), one possible partial explanation for more accurate responding toward "before" trials compared to "after" is that the degree of practice with each cue in the natural environment might be a factor. Relational responses that are well-practiced would be considered lower in derivation than be responses with less practice. Two questions for empirical testing relative to derivation include whether children receive more exposure to "before" relations than "after" relations, and whether children receive more practice deriving "after" relations when given a "before" statement than the reverse. Whether or not children do in fact receive more practice with temporal relations like this within in their socioverbal environments requires empirical investigation.

Lastly, flexibility of the relational response was invoked by introducing MET. In all participants, relational responses on "after" trials changed when the reinforcing context changed. The degree to which a participant's repertoire was flexible was more idiosyncratic, however. In Study 2 for instance, one participant found it more difficult to respond accurately on trial blocks when relational statements were mixed with both "before" and "after" cues compared to when they were presented in isolation (i.e., "before" statements only, and then "after" only). The relational responding in this example is lower in flexibility compared to a response pattern that can more readily shift between changing relational conditions.

There are both theoretical and applied implications of this finding on contextual control. Other applied RFT work has often not reported in detail on performance differences between contextual cues when a particular frame entails more than one relation (e.g., comparison, hierarchical). The data from the current thesis highlighted that this was an especially relevant consideration relevant to contextual control with temporal relations, and simply reporting on the overall accuracy with both cues would have provided an incomplete learning picture. At present, it is not clear whether this finding is similar with other frames involving multiple contextual cues. Comparing performance across contextual cues (e.g., baseline to intervention changes, rates of acquisition) with other frames could offer a more complete picture in terms of the acquisition of AARR repertoires in childhood. The applied implication of this result also suggests that practitioners working with individuals with emerging relational skills should consider collecting data on tests for derivation across various presentations of relational networks and question types. By doing so, decision making in terms of intervention adjustments could be enhanced (e.g., providing booster sessions or trial blocks with certain trial types).

Another interesting theoretical consideration is that a nonarbitrary temporal relation is technically always in effect regardless of the type of relational network presented (whether a temporal frame, or otherwise). Organisms experience events sequentially in time and this is true for verbal

Chapter 7

stimuli as well. In other words, when reading a relational network or hearing it spoken aloud, a person orients to these stimuli in sequence. Future work might consider to what degree this dynamic interacts with relational responding more generally.

Approaching responding to "before" and "after" from an operant perspective also offers application to the broader literature base. As discussed in Chapter 1, different psychological perspectives on behavior may pursue slightly different scientific goals through different means. These philosophical world views need not be in competition, but rather, can each offer a unique insight into a particular phenomenon. Existing work, primarily from cognitive developmental and linguistic approaches, has explored the comprehension of "before" and "after" in terms of age-related performance (Busby Grant & Suddendorf, 2011), arranging sentence variations (Clark, 1971), and connections to neural correlates (Ye et al., 2012) or cognitive processes (Blything et al., 2015). This work has helped inform models that align more closely with mechanistic or organicist leanings. However, one gap in the literature is that these studies have mostly been observational or descriptive, limiting what can be understood in terms of the acquisition of these skills. The current research introduced an alternative perspective (contextualistic) on this phenomenon (temporal relations) to the broader literature base which offers an advantage relative to the is research gap. Data from this thesis provided a specific methodology for the prediction and influence of an operant behavior involved in the comprehension of "before" and after." Further, this behavioral approach provides insight into distinguishing between different sources of stimulus control over "before/after" responding, which previous research has not explored. Responses may be contextually controlled by a nonarbitrary temporal relation, a solely arbitrary contextual cue, or a combination. Just as these cited findings helped inform this research program, so too can this data inform future work rooted in other philosophical assumptions. Cooperation and collaboration around a particular phenomenon rather than working in silos will hopefully lead to better research, leading to better technologies, leading to a more meaningful impact on consumers. Indeed, there are many interesting points of intersection between branches of psychology surrounding relational reasoning or learning (McLoughlin, Tyndall, & Pereira, 2020). Continued integration in this area is a promising direction.

A final major contribution from the current work is that it helped provide conceptual basis for approaching a range of other time-related skills to be approached behaviorally. Although the empirical work detailed in Chapters 3-6 focused specifically on temporal relational skills with "before" and "after" cues, there are other adjacent time-related repertoires. Friedman (2011, p. 398) described that time involves, "recurrent sequences of events, natural and conventional time patterns, invariant causal sequences, logical relations between succession and duration, the past-present-future distinction, and many others." Chapter 2 specifically highlighted the key skills of timing, clock and calendar acquisition, self-control, planning, and time management. From an RFT view, various other relational skills may be relevant to these skill sets (e.g., frames of coordination with duration units; comparison relations with immediate and delayed contingencies; relational networks involved in

Chapter 7

planning and time management). Even with temporal event sequencing, there is room for work beyond simple "before" and "after" comprehension (e.g., deictic relations with past and future sequences, causal relations, etc.). In conjunction with the literature reviewed across these domains, an RFT interpretation was presented that could allow for the development of methods to support research in these areas.

## **Limitation and Future Research Directions**

Despite the contributions offered through the current research program, several limitations were evident which should be addressed in continued work. This section provides an overview of these limitations, suggested ways to address them, as well as other potential areas of empirical investigation inspired by this work.

One limitation of this research is that the connection between temporal relational responding and other repertoires was not examined. RFT proposes that AARR is the key operant relevant to language and cognition (Hughes & Barnes-Holmes, 2016). Accordingly, language-based skills or cognitive skills might be specifically examined in relation to instances of AARR. Future work could utilize the temporal relational responding assessment used in Study 1 and compare performance relative to standardize measures of language or cognitive skills (e.g., Peabody Picture Vocabulary Test, Stanford-Binet Intelligence Scales, Bracken Basic Concept Scale) in a larger sample. Further, future work could examine whether training in temporal relational responding aids in performance on any of these measures. Alternatively, research could compare temporal relational responding performance with other behavioral repertoires not necessarily measured by standardized measures. In Chapter 2, a range of temporal skill sets were specifically identified. Future work could compare how acquisition of temporal relational responding intersects with performance on tasks designed to measure skills like planning (e.g., route-planning tasks) or self-control (e.g., delay of gratification tasks).

A second area for continued work is to address the relatively limited scope with responses and stimuli used in the testing and training protocols. In this thesis, all the arbitrary relations were presented using an adapted REP format. This particular method provides an advantage in terms efficiency (Kirsten et al., 2022a). It is unclear however, if similar performance patterns would have been obtained using alternative methods, such as a match-to-sample (MTS) arrangement. Future work could investigate temporal frames in this context to see what kind of results an MTS approach provides. Along these lines, future work might consider using different spatial arrangements for presenting relational networks, include other stimuli (e.g., either culturally relevant or more abstract stimuli), present stimuli across different sense modalities, and train across additional contextual cues (e.g., earlier/later, first/then). It is also recommended that future work include broadening the types of responses included within a MET protocol, such as tacting the relation, selection-based responses, orienting responses, other stimulus functions, and/or higher levels of relational responding in the HDML.

Future work might also look at the importance of nonarbitrary relations. In this thesis, Study 4 did not explicitly extend training beyond contextually controlled responding to nonarbitrary temporal relations. Subsequent research should examine whether such acquisition supports learning arbitrary temporal relations. For example, Berens and Hayes (2007) found that including nonarbitrary pre-trials as an intervention component supported acquisition of arbitrary comparison relating. Similar work can be extended to temporal relations. Moreover, the current protocol was limited insofar as the timescale that was used (e.g., 1s intervals). Subsequent research should incorporate the testing and training of nonarbitrary relations with longer (or varied) durations. Additionally, future work could examine the utility of using nonarbitrary relations to support the acquisition of other time skills. For example, duration units might be taught by using visual nonarbitrary comparison relations (e.g., 1 hour vs 1 minute comparison using visual bars, etc.).

Another limitation in the current work is that a social validity measure was not included for the trainings. As the training studies were preliminary investigations in terms of developing an intervention, priority was placed on first evaluating its efficacy. Future iterations of this intervention should additionally include direct consumer feedback. Participants could offer feedback in terms of their experience participating in the training and offer ideas for improvement. In addition to measuring social validity with children or their families, this could be applied to behavioral practitioners as well. For example, future work could instead have clinicians implement the RFT based protocol, and they could share their experience, perspective, or challenges implementing it.

The issue of social validity is perhaps a more widespread area of need within applied RFT work, let alone behavior analysis more generally. Given that social significance is a fundamental principal within ABA (Baer et al., 1968), this is an area that should take priority in future applied RFT work. For instance, this research line could be adapted into more naturalistic settings (e.g., play scenarios) or embed more functional response sequences (e.g., life skills) to increase social validity. One empirical example to build on utilized equivalence based instruction (EBI) to teach a generalized grocery store shopping sequence (Taylor & O'Reilly, 2000). One dyad was taught the shopping sequence in vivo across multiple grocery stores, while participants in the EBI dyad practiced the skill in one store but completed EBI with pictures of different stores. The EBI dyad demonstrated similar generalization performance compared to the multiple stores dyad when assessed in a novel store. This intervention only used equivalence relations, however, any behavior chain in which the order of actions is important (e.g., steps for cooking, etc.) could also incorporate training temporal relations.

A potential roadblock encountered by clinicians when implementing interventions targeting derived relations in practice is the complex nature of creating programs. Protocols often include a high number of discrete trials, systematic rotation of stimuli, ensuring appropriate testing conditions for derived relations, and navigating potentially less familiar terminology (Marano-Frezza et al., 2022). These are not insurmountable challenges to a clinician; however, it may still require a higher response effort on their part compared to more familiar ABA programming. Considering the emerging

evidence that relation training may enhance outcomes in ABA intervention (Dixon et al., 2023) however, increased attention in this area is appropriate.

One avenue that might help address both the concerns of social validity and clinician response effort or error, is through investing in creating technology-based programs that incorporate RFT. Digital platforms offer the benefit of automating certain tasks, which could assist with program design (e.g., selecting and arranging stimuli, data collection on trials, etc.). Additionally, some language learning software platforms infuse artificial intelligence (AI) into their design, which further reduces the burden on the instructor and could potentially enhance learning opportunities for consumers. For example, platforms such as Duolingo or ELSA embed AI, which helps identify errors made, provide feedback, adapt trial blocks, and personalize lessons to fit the needs of the learner. It is reasonable that that such an approach can fit within the scope of relational training. Although some designs currently exist such as SMART Training (Cassidy et al., 2016) or the T-IRAP (Murphy et al., 2019), there is room for more consumer-friendly designs along with integrating present technological advances. This could reduce a barrier of needing to "know" RFT well on the part of the clinician, while also broadening the scope for implementation via caregivers, teachers, or even child-directed learning.

## Conclusion

This thesis contributes to the limited research on temporal relational responding by extending empirical investigation to new demographics, and by designing testing and training procedures for improving responding. Findings from this thesis provide evidence that an RFT-based assessment procedure can be used to determine temporal relational responding skill level, which in turn can help guide intervention. This work extends existing RFT research that MET training can effectively improve relational responding performance in both neurotypical and neurodiverse individuals at multiple levels (i.e., arbitrary and nonarbitrary relations). This thesis also provides a conceptual analysis for other potential areas of RFT application within psychological research on time-related skill sets. The findings from this thesis are preliminary, however. More research is needed to replicate the results, refine the testing and training procedures, and show relevancy of these relational responses to other adaptive skill sets. Given the importance of responding to temporal features of the environment and the advantages that a verbal experience of time offers, continued work in this area is warranted.

## References

- Allman, M. J., Teki, S., Griffiths, T. D., & Meck, W. H. (2014). Properties of the internal clock: Firstand second-order principles of subjective time. *Annual Review of Psychology*, 65, 743-771. <u>https://doi.org/10.1146/annurev-psych-010213-115117</u>
- Amd, M., & Roche, B. (2018). Assessing the effects of a relational training intervention on fluid intelligence among a sample of socially disadvantaged children in Bangladesh. *The Psychological Record*, 68(2), 141-149. <u>https://doi.org/10.1007/s40732-018-0273-4</u>
- American Psychiatric Association. (2000). *Diagnostic and statistical manual of mental disorders* (4th ed.).
- Amidon, A., & Carey, P. (1972). Why five-year-olds cannot understand before and after. *Journal of Verbal Learning and Verbal Behavior*, 11(4), 417-423. <u>https://doi.org/10.1016/s0022-5371(72)80022-7</u>
- Angrilli, A., Cherubini, P., Pavese, A., & Mantredini, S. (1997). The influence of affective factors on time perception. *Perception & Psychophysics*, 59(6), 972-982. <u>https://doi.org/10.3758/bf03205512</u>
- Arstila, V., & Lloyd, D. (Eds.). (2014). Subjective time: The philosophy, psychology, and neuroscience of temporality. MIT Press.
- Atance, C. M., & Meltzoff, A. N. (2005). My future self: Young children's ability to anticipate and explain future states. *Cognitive Development*, 20, 341-361. https://doi.org/10.1016/j.cogdev.2005.05.001
- Atance, C. M., & O'Neill, D. K. (2005). The emergence of episodic future thinking in humans. *Learning and Motivation*, 36(2), 126-144. <u>https://doi.org/10.1016/j.lmot.2005.02.003</u>
- Bach, P. A., & Moran, D. J. (2008). Contacting the present moment and perspective taking. In ACT in practice: Case conceptualization in acceptance & commitment therapy (pp. 127-142). New Harbinger Publications, Inc.
- Baer, D. M., Wolf, M. M., & Risley, T. R. (1968). Some current dimensions of applied behavior analysis. *Journal of Applied Behavior Analysis*, 1(1), 91-97. https://doi.org/10.1901/jaba.1968.1-91
- Barkley, R. A. (2011). Barkley Deficits in Executive Functioning Scale (BDEFS). The Guilford Press.
- Barkley, R. A. (2012). *Barkley Deficits in Executive Functioning Scale—Children and Adolescents* (*BDEFS-CA*). The Guilford Press.
- Barkley, R. A. (2012). *Executive functions: What they are, how they work, and why they evolved.* The Guilford Press.
- Barkley, R. A. (2013). *Taking charge of ADHD: The complete authoritative guide for parents*. The Guilford Press.

- Barnes, D., McCullagh, P. D., & Keenan, M. (1990). Equivalence class formation in non-hearing impaired children and hearing impaired children. *The Analysis of Verbal Behavior*, 8, 19-30. https://doi.org/10.1007/BF03392844
- Barnes-Holmes, D., Barnes-Holmes, Y., Luciano, C., & McEnteggart, C. (2017). From the IRAP and REC model to a multi-dimensional multi-level framework for analyzing the dynamics of arbitrarily applicable relational responding. *Journal of Contextual Behavioral Science*, 6(4), 434-445. <u>https://doi.org/10.1016/j.jcbs.2017.08.001</u>
- Barnes-Holmes, D., Barnes-Holmes, Y., & McEnteggart, C. (2020). Updating RFT (more field than frame) and its implications for process-based therapy. *The Psychological Record*, 70(4), 605-624. <u>https://doi.org/10.1007/s40732-019-00372-3</u>
- Barnes-Holmes, D., Hayes, S. C., & Dymond, S. (2001). Self and self directed rules. In S. C. Hayes,
  D. Barnes-Holmes, & B. Roche (Eds.), *Relational frame theory: A post-Skinnerian account of human language and cognition* (pp. 119-139). Kluwer Academic/Plenum.
- Barnes-Holmes, D., Staunton, C., Whelan, R., Barnes-Holmes, Y., Commins, S., Walsh, D., Stewart, I., Smeets, P. M., & Dymond, S. (2005). Derived stimulus relations, semantic priming, and event-related potentials: Testing a behavioral theory of semantic networks. *Journal of the Experimental Analysis of Behavior*, 84(3), 417-433. https://doi.org/10.1901/jeab.2005.78-04
- Barnes-Holmes, Y., Barnes-Holmes, D., & McHugh, L. (2004). Teaching derived relational responding to young children. *Journal of Early and Intensive Behavior Intervention*, 1(1), 3-12. https://doi.org/10.1037/h0100275
- Barnes-Holmes, Y., Barnes-Holmes, D., McHugh, L., & Hayes, S. C. (2004). Relational frame theory: Some implications for understanding and treating human psychopathology. *International Journal of Psychology & Psychological Therapy*, 4(2), 355-375.
- Barnes-Holmes, Y., Barnes-Holmes, D., Roche, B., Healy, O., Lyddy, F., Cullinan, V., & Hayes, S.
  C. (2001). Psychological development. In S. C. Hayes, D. Barnes-Holmes, & B. Roche
  (Eds.), *Relational frame theory: A post-Skinnerian account of human language and cognition*(pp. 157-180). Kluwer Academic/Plenum.
- Barnes-Holmes, Y., Barnes-Holmes, D., Smeets, P. M., Strand, P., & Friman, P. (2004). Establishing relational responding in accordance with more-than and less-than as generalized operant behavior in young children. *International Journal of Psychology & Psychological Therapy*, 4(3), 531-558.
- Barnes-Holmes, Y., McHugh, L., & Barnes-Holmes, D. (2004). Perspective-taking and Theory of Mind: A relational frame account. *The Behavior Analyst Today*, 5(1), 15-25. <u>https://doi.org/10.1037/h0100133</u>
- Baron, A., Kaufman, A., & Stauber, K. A. (1969). Effects of instructions and reinforcement-feedback on human operant behavior maintained by fixed-interval reinforcement. *Journal of the Experimental Analysis of Behavior*, 12(5), 701-712. <u>https://doi.org/10.1901/jeab.1969.12-701</u>

- Barron, B. F., Verkuylen, L., Belisle, J., Paliliunas, D., & Dixon, M. R. (2019). Teaching "then-later" and "here-there" relations to children with autism: An evaluation of single reversals and transformation of stimulus function. *Behavior Analysis in Practice*, 12(1), 167-175. <u>https://doi.org/10.1007/s40617-018-0216-1</u>
- Bartels, D. M., & Urminsky, O. (2011). On intertemporal selfishness: How the perceived instability of identity underlies impatient consumption. *Journal of Consumer Research*, 38(1), 182-198. <u>https://doi.org/10.1086/658339</u>
- Bauer, P. J., & Mandler, J. M. (1989). One thing follows another: Effects of temporal structure on 1to 2-year-olds' recall of events. *Developmental Psychology*, 25(2), 197-206. <u>https://doi.org/10.1037/0012-1649.25.2.197</u>
- Belisle, J., & Dixon, M. R. (2021). Relational behavior and ACT: A dynamic relationship. *Behavior Analysis in Practice*, 15(1), 71-82. <u>https://doi.org/10.1007/s40617-021-00599-z</u>
- Belisle, J., Dixon, M. R., & Stanley, C. R. (2018). The mediating effects of derived relational responding on the relationship between verbal operant development and IQ. *Behavior Analysis in Practice*, 11(4), 411-416. https://doi.org/10.1007/s40617-018-0215-2
- Belisle, J., Dixon, M. R., Stanley, C. R., Munoz, B., & Daar, J. H. (2016). Teaching foundational perspective-taking skills to children with autism using the PEAK-T curriculum: singlereversal "I-You" deictic frames. *Journal of Applied Behavior Analysis*, 49(4), 965-969. https://doi.org/10.1002/jaba.324
- Belisle, J., Huggins, K., Doherty, M., Stanley, C. R., & Dixon, M. R. (2020). Generalized reflexive responding and cross-modal tactile transfer of stimulus function in children with autism. *The Analysis of Verbal Behavior*, 36(2), 233-250. <u>https://doi.org/10.1007/s40616-020-00137-y</u>
- Belisle, J., Stanley, C. R., Schmick, A., Dixon, M. R., Alholail, A., Galliford, M. E., & Ellenberger, L. (2020). Establishing arbitrary comparative relations and referential transformations of stimulus function in individuals with autism. *Journal of Applied Behavior Analysis*, 53(2), 938-955. <u>https://doi.org/10.1002/jaba.655</u>
- Benoit, R. G., Gilbert, S. J., & Burgess, P. W. (2011). A neural mechanism mediating the impact of episodic prospection on farsighted decisions. *Journal of Neuroscience*, 31(18), 6771-6779. https://doi.org/10.1523/JNEUROSCI.6559-10.2011
- Bentall, R. P., & Lowe, C. F. (1987). The role of verbal behavior in human learning: III. Instructional effects in children. *Journal of the Experimental Analysis of Behavior*, 47(2), 177-190. <u>https://doi.org/10.1901/jeab.1987.47-177</u>
- Bentall, R. P., Lowe, C. F., & Beasty, A. (1985). The role of verbal behavior in human learning: II. Developmental differences. *Journal of the Experimental Analysis of Behavior*, 43(2), 165-181. <u>https://doi.org/10.1901/jeab.1985.43-165</u>

- Berens, N. M., & Hayes, S. C. (2007). Arbitrarily applicable comparative relations: Experimental evidence for a relational operant. *Journal of Applied Behavior Analysis*, 40(1), 45-71. https://doi.org/10.1901/jaba.2007.7-06
- Biglan, A., & Barnes-Holmes, Y. (2015). Acting in light of the future: How do future-oriented cultural practices evolve and how can we accelerate their evolution? *Journal of Contextual Behavioral Science*, 4(3), 184-195. <u>https://doi.org/10.1016/j.jcbs.2015.06.002</u>
- Biglan, A., & Hayes, S. C. (1996). Should the behavioral sciences become more pragmatic? The case for functional contextualism in research on human behavior. *Applied and Preventive Psychology*, 5(1), 47-57. <u>https://doi.org/10.1016/s0962-1849(96)80026-6</u>
- Biglan, A., & Hayes, S. C. (2016). Functional contextualism and contextual behavioral science. In R.
  Zettle, S. C. Hayes, D. Barnes-Holmes, & A. Biglan (Eds.), *The wiley handbook of contextual behavioral science* (pp. 37-61). Wiley Blackwell. https://doi.org/10.1002/9781118489857.ch4
- Binder, C. (1996). Behavioral fluency: Evolution of a new paradigm. *The Behavior Analyst*, 19(2), 163-197. <u>https://doi.org/10.1007/BF03393163</u>
- Binder, L. M., Dixon, M. R., & Ghezzi, P. M. (2000). A procedure to teach self-control to children with attention deficit hyperactivity disorder. *Journal of Applied Behavior Analysis*, 33(2), 233-237. <u>https://doi.org/10.1901/jaba.2000.33-233</u>
- Blackledge, J. T. (2003). An introduction to relational frame theory: Basics and applications. *The Behavior Analyst Today*, *3*(4), 421-433. <u>https://doi.org/10.1037/h0099997</u>
- Block, R. A., Hancock, P. A., & Zakay, D. (2010). How cognitive load affects duration judgments: A meta-analytic review. *Acta Psychologica*, 134(3), 330-343. <u>https://doi.org/10.1016/j.actpsy.2010.03.006</u>
- Block, R. A., Zakay, D., & Hancock, P. A. (1999). Developmental changes in human duration judgments: A meta-analytic review. *Developmental Review*, 19(1), 183-211. <u>https://doi.org/10.1006/drev.1998.0475</u>
- Blything, L. P., Davies, R., & Cain, K. (2015). Young children's comprehension of temporal relations in complex sentences: The influence of memory on performance. *Child Development*, 86(6), 1922-1934. https://doi.org/10.1111/cdev.12412
- Boroditsky, L. (2001). Does language shape thought? Mandarin and English speakers' conceptions of time. *Cognitive Psychology*, *43*(1), 1-22. <u>https://doi.org/10.1006/cogp.2001.0748</u>
- Boroditsky, L., & Gaby, A. (2010). Remembrances of times East: Absolute spatial representations of time in an Australian aboriginal community. *Psychological Science*, 21(11), 1635-1639. <u>https://doi.org/10.1177/0956797610386621</u>
- Brackbill, Y., & Fitzgerald, H. E. (1972). Stereotype temporal conditioning in infants. *Psychophysiology*, 9(6), 569-577. <u>https://doi.org/10.1111/j.1469-8986.1972.tb00766.x</u>

- Brannon, E. M., Suanda, S., & Libertus, K. (2007). Temporal discrimination increases in precision over development and parallels the development of numerosity discrimination. *Developmental Science*, 10(6), 770-777. https://doi.org/10.1111/j.1467-7687.2007.00635.x
- Brassil, N., Hyland, J., O'Hora, D., & Stewart, I. (2019). Reversing time and size: Mutual entailment of nonarbitrary temporal and magnitude relational responding. *The Psychological Record*, 69(1), 95-105. https://doi.org/10.1007/s40732-018-0323-y
- Brislin, R. W., & Kim, E. S. (2003). Cultural diversity in people's understanding and uses of time. *Applied Psychology*, 52(3), 363-382. <u>https://doi.org/10.1111/1464-0597.00140</u>
- Buckner, R. L., & Carroll, D. C. (2007). Self-projection and the brain. *Trends in Cognitive Sciences*, 11(2), 49-57. https://doi.org/10.1016/j.tics.2006.11.004
- Bull, R., Espy, K. A., & Senn, T. E. (2004). A comparison of performance on the Towers of London and Hanoi in young children. *Journal of Child Psychology and Psychiatry*, 45(4), 743-754. https://doi.org/10.1111/j.1469-7610.2004.00268.x
- Busby Grant, J., & Suddendorf, T. (2010). Young children's ability to distinguish past and future changes in physical and mental states. *British Journal of Developmental Psychology*, 28(Pt 4), 853-870. <u>https://doi.org/10.1348/026151009x482930</u>
- Busby Grant, J., & Suddendorf, T. (2011). Production of temporal terms by 3-, 4-, and 5-year-old children. *Early Childhood Research Quarterly*, 26(1), 87-95. <u>https://doi.org/10.1016/j.ecresq.2010.05.002</u>
- Cannon, L., Kenworthy, L., Alexander, K., Werner, M., & Anthony, L. (2021). Unstuck and on target! (2nd ed.). Brookes Publishing.
- Capaldi, E. J., & Proctor, R. W. (1994). Contextualism: Is the act in context the adequate metaphor for scientific psychology? *Psychonomic Bulletin & Review*, 1(2), 239-249. https://doi.org/10.3758/BF03200775
- Carlson, S. M., Moses, L. J., & Claxton, L. J. (2004). Individual differences in executive functioning and theory of mind: An investigation of inhibitory control and planning ability. *Journal of Experimental Child Psychology*, 87(4), 299-319. https://doi.org/10.1016/j.jecp.2004.01.002
- Carmelo Visdómine-Lozano, J. (2015). A functional-contextualist account of locus of control: Generalized control expectancies as derived relational responding. *Journal of Contextual Behavioral Science*, 4(4), 312-323. <u>https://doi.org/10.1016/j.jcbs.2015.06.005</u>
- Carni, E., & French, L. A. (1984). The acquisition of *before* and *after* reconsidered: What develops? *Journal of Experimental Child Psychology*, *37*(2), 394-403. <u>https://doi.org/10.1016/0022-0965(84)90011-0</u>
- Carpentier, F., Smeets, P. M., & Barnes-Holmes, D. (2002). Establishing transfer of compound control in children: A stimulus control analysis. *The Psychological Record*, 52(2), 139-158. <u>https://doi.org/10.1007/bf03395420</u>

- Casassus, M., Poliakoff, E., Gowen, E., Poole, D., & Jones, L. A. (2019). Time perception and autistic spectrum condition: A systematic review. *Autism Research*, 12(10), 1440-1462. <u>https://doi.org/10.1002/aur.2170</u>
- Cassidy, S., Roche, B., Colbert, D., Stewart, I., & Grey, I. M. (2016). A relational frame skills training intervention to increase general intelligence and scholastic aptitude. *Learning and Individual Differences*, 47, 222-235. <u>https://doi.org/10.1016/j.lindif.2016.03.001</u>
- Cassidy, S., Roche, B., & Hayes, S. C. (2011). A relational frame training intervention to raise intelligence quotients: A pilot study. *The Psychological Record*, 61(2), 173-198. <u>https://doi.org/10.1007/bf03395755</u>
- Christ, T. J. (2007). Experimental control and threats to internal validity of concurrent and nonconcurrent multiple baseline designs. *Psychology in the Schools*, 44(5), 451-459. <u>https://doi.org/10.1002/pits.20237</u>
- Church, R. M. (1984). Properties of the internal clock. *Annals of the New York Academy of Sciences*, 423(1), 566-582. <u>https://doi.org/10.1111/j.1749-6632.1984.tb23459.x</u>
- Claessens, B. J. C., van Eerde, W., Rutte, C. G., & Roe, R. A. (2007). A review of the time management literature. *Personnel Review*, 36(2), 255-276. https://doi.org/10.1108/00483480710726136
- Clark, E. V. (1971). On the acquisition of the meaning of before and after. *Journal of Verbal Learning* and Verbal Behavior, 10(3), 266-275. https://doi.org/10.1016/s0022-5371(71)80054-3
- Colbert, D., Dobutowitsch, M., Roche, B., & Brophy, C. (2017). The proxy-measurement of intelligence quotients using a relational skills abilities index. *Learning and Individual Differences*, 57, 114-122. <u>https://doi.org/10.1016/j.lindif.2017.03.010</u>
- Colbert, D., Malone, A., Barrett, S., & Roche, B. (2020). The Relational Abilities Index+: Initial validation of a functionally understood proxy measure for intelligence. *Perspectives on Behavior Science*, *43*(1), 189-213. <u>https://doi.org/10.1007/s40614-019-00197-z</u>
- Colbert, D., Tyndall, I., Roche, B., & Cassidy, S. (2018). Can SMART training really increase intelligence? A replication study. *Journal of Behavioral Education*, *27*(4), 509-531. https://doi.org/10.1007/s10864-018-9302-2
- Colombo, J., & Richman, W. A. (2002). Infant timekeeping: Attention and temporal estimation in 4month-olds. *Psychological Science*, *13*(5), 475-479. <u>https://doi.org/10.1111/1467-9280.00484</u>
- Cooper, J. O., Heron, T. E., & Heward, W. L. (2020). *Applied behavior analysis* (3rd ed.). Pearson Education.
- Coull, J., & Nobre, A. (2008). Dissociating explicit timing from temporal expectation with fMRI. *Current Opinion in Neurobiology*, *18*(2), 137-144. <u>https://doi.org/10.1016/j.conb.2008.07.011</u>
- Coull, J. T., Vidal, F., Nazarian, B., & Macar, F. (2004). Functional anatomy of the attentional modulation of time estimation. *Science*, 303(5663), 1506-1508. <u>https://doi.org/10.1126/science.1091573</u>

- Cromer, R. F. (1971). The development of the ability to decenter in time. *British Journal of Psychology*, 62(3), 353-365. https://doi.org/10.1111/j.2044-8295.1971.tb02046.x
- Dahiya, A. V., McDonnell, C., DeLucia, E., & Scarpa, A. (2020). A systematic review of remote telehealth assessments for early signs of autism spectrum disorder: Video and mobile applications. *Practice Innovations*, 5(2), 150-164. <u>https://doi.org/10.1037/pri0000121</u>
- Daniel, T. O., Said, M., Stanton, C. M., & Epstein, L. H. (2015). Episodic future thinking reduces delay discounting and energy intake in children. *Eating Behaviors*, 18, 20-24. <u>https://doi.org/10.1016/j.eatbeh.2015.03.006</u>
- Daniel, T. O., Stanton, C. M., & Epstein, L. H. (2013). The future is now: Reducing impulsivity and energy intake using episodic future thinking. *Psychological Science*, 24(11), 2339-2342. <u>https://doi.org/10.1177/0956797613488780</u>
- Darcheville, J. C., Riviere, V., & Wearden, J. H. (1993). Fixed-interval performance and self-control in infants. *Journal of the Experimental Analysis of Behavior*, 60(2), 239-254. <u>https://doi.org/10.1901/jeab.1993.60-239</u>
- Davlin, N. L., Anne Rehfeldt, R., & Lovett, S. (2011). A relational frame theory approach to understanding perspective-taking using children's stories in typically developing children. *European Journal of Behavior Analysis*, 12(2), 403-430. https://doi.org/10.1080/15021149.2011.11434392
- Dawson, P., & Guare, R. (2009). Smart but scattered. The Guilford Press.
- Dawson, P., & Guare, R. (2010). *Executive skills in children and adolescents: A practical guide to assessment and intervention* (2nd ed.). The Guilford Press.
- de Rose, J. C., McIlvane, W. J., Dube, W. V., Galpin, V. C., & Stoddard, L. T. (1988). Emergent simple discrimination established by indirect relation to differential consequences. *Journal of the Experimental Analysis of Behavior*, 50(1), 1-20. <u>https://doi.org/10.1901/jeab.1988.50-1</u>
- Devany, J. M., Hayes, S. C., & Nelson, R. O. (1986). Equivalence class formation in language-able and language-disabled children. *Journal of the Experimental Analysis of Behavior*, 46(3), 243-257. <u>https://doi.org/10.1901/jeab.1986.46-243</u>
- Dickins, D. W., Singh, K. D., Roberts, N., Burns, P., Downes, J. J., Jimmieson, P., & Bentall, R. P. (2001). An fMRI study of stimulus equivalence. *Neuroreport*, 12(2), 405-411. <u>https://doi.org/10.1097/00001756-200102120-00043</u>
- Dilworth-Anderson, P., Burton, L. M., & Klein, D. M. (2005). Contemporary and emerging theories in studying families. In V. L. Bengtson, A. C. Acock, K. R. Allen, P. Dilworth-Anderson, & D. M. Klein (Eds.), *Sourcebook of family theory and research* (pp. 35-58). Sage Publications.
- DiPipi-Hoy, C., & Steere, D. (2016). *Teaching time management to learners with autism spectrum disorder*. AAPC Publishing.
- Dixon, M. R. (2015). *PEAK relational training system: Equivalence module*. Shawnee Scientific Press.

- Dixon, M. R. (2016). *PEAK relational training system: Transformation module*. Shawnee Scientific Press.
- Dixon, M. R., & Belisle, J. (2021). Selection by consequence: A response to Hayes and Fryling (2019). Journal of Contextual Behavioral Science, 19, 6-11. https://doi.org/10.1016/j.jcbs.2020.10.003
- Dixon, M. R., Belisle, J., & Stanley, C. R. (2018). Derived relational responding and intelligence: Assessing the relationship between the PEAK-E pre-assessment and IQ with individuals with autism and related disabilities. *The Psychological Record*, 68(4), 419-430. <u>https://doi.org/10.1007/s40732-018-0284-1</u>
- Dixon, M. R., & Holcomb, S. (2000). Teaching self-control to small groups of dually diagnosed adults. *Journal of Applied Behavior Analysis*, 33(4), 611-614. <u>https://doi.org/10.1901/jaba.2000.33-611</u>
- Dixon, M. R., & Holton, B. (2009). Altering the magnitude of delay discounting by pathological gamblers. *Journal of Applied Behavior Analysis*, 42(2), 269-275. <u>https://doi.org/10.1901/jaba.2009.42-269</u>
- Dixon, M. R., & Stanley, C. R. (2020). Relational frame theory: Basic relational operants. In M. Fryling, R. A. Rehfeldt, J. Tarbox, & L. J. Hayes (Eds.), *Applied behavior analysis of language and cognition: Core concepts and principles for practitioners* (pp. 174-197). Context Press.
- Dixon, M. R., & Tibbetts, P. A. (2009). The effects of choice on self-control. *Journal of Applied Behavior Analysis*, 42(2), 243-252. <u>https://doi.org/10.1901/jaba.2009.42-243</u>
- Dixon, M. R., Yi, Z., Sutton, A., & Pikula, A. (2023). Randomized-controlled trial examining differing content of ABA therapy delivered to autistic children. *Journal of Contextual Behavioral Science*, 28, 169-179. https://doi.org/10.1016/j.jcbs.2023.03.016
- Dougher, M. J., Augustson, E., Markham, M. R., Greenway, D. E., & Wulfert, E. (1994). The transfer of respondent eliciting and extinction functions through stimulus equivalence classes. *Journal* of the Experimental Analysis of Behavior, 62(3), 331-351. https://doi.org/10.1901/jeab.1994.62-331
- Dragoi, V., Staddon, J. E., Palmer, R. G., & Buhusi, C. V. (2003). Interval timing as an emergent learning property. *Psychological Review*, 110(1), 126-144. <u>https://doi.org/10.1037/0033-295x.110.1.126</u>
- Droit, S., Pouthas, V., & Jacquet, A.-Y. (1990). Temporal learning in 4 1/2- and 6-year-old children: Role of instructions and prior knowledge. *Journal of Experimental Child Psychology*, 50(2), 305-321. <u>https://doi.org/10.1016/0022-0965(90)90044-9</u>
- Droit, S., Pouthas, V., & Jacquet, A. Y. (1991). Apprentissage temporel chez des enfants âgés de 4 1/2 et 6 ans : rôle d'une horloge externe. *L'année psychologique*, *91*(3), 347-364. <u>https://doi.org/10.3406/psy.1991.29471</u>

- Droit-Volet, S. (1998). Time estimation in young children: An initial force rule governing time production. *Journal of Experimental Child Psychology*, *68*(3), 236-249.
- Droit-Volet, S. (2011). Child and time. In *Multidisciplinary aspects of time and time perception* (pp. 151-172). <u>https://doi.org/10.1007/978-3-642-21478-3\_13</u>
- Droit-Volet, S. (2013). Time perception in children: A neurodevelopmental approach. *Neuropsychologia*, *51*(2), 220-234. <u>https://doi.org/10.1016/j.neuropsychologia.2012.09.023</u>
- Droit-Volet, S., Clement, A., & Wearden, J. (2001). Temporal generalization in 3- to 8-year-old children. *Journal of Experimental Child Psychology*, 80(3), 271-288. <u>https://doi.org/10.1006/jecp.2001.2629</u>
- Droit-Volet, S., & Coull, J. T. (2016). Distinct developmental trajectories for explicit and implicit timing. *Journal of Experimental Child Psychology*, *150*, 141-154. https://doi.org/10.1016/j.jecp.2016.05.010
- Droit-Volet, S., Fanget, M., & Dambrun, M. (2015). Mindfulness meditation and relaxation training increases time sensitivity. *Consciousness and Cognition*, 31, 86-97. <u>https://doi.org/10.1016/j.concog.2014.10.007</u>
- Droit-Volet, S., & Gil, S. (2009). The time-emotion paradox. *Philosophical Transactions of The Royal Society B*, 364(1525), 1943-1953. <u>https://doi.org/10.1098/rstb.2009.0013</u>
- Droit-Volet, S., Lamotte, M., & Izaute, M. (2015). The conscious awareness of time distortions regulates the effect of emotion on the perception of time. *Consciousness and Cognition*, 38, 155-164. <u>https://doi.org/10.1016/j.concog.2015.02.021</u>
- Droit-Volet, S., & Meck, W. H. (2007). How emotions colour our perception of time. *Trends in Cognitive Sciences*, 11(12), 504-513. https://doi.org/10.1016/j.tics.2007.09.008
- Droit-Volet, S., & Wearden, J. H. (2001). Temporal bisection in children. *Journal of Experimental Child Psychology*, 80(2), 142-159. <u>https://doi.org/10.1006/jecp.2001.2631</u>
- Dugdale, N., & Lowe, C. F. (2000). Testing for symmetry in the conditional discriminations of language-trained chimpanzees. *Journal of the Experimental Analysis of Behavior*, 73(1), 5-22. <u>https://doi.org/10.1901/jeab.2000.73-5</u>
- Dunne, S., Foody, M., Barnes-Holmes, Y., Barnes-Holmes, D., & Murphy, C. (2014). Facilitating repertoires of coordination, opposition distinction, and comparison in young children with autism. *Behavioral Development Bulletin*, 19(2), 37-47. <u>https://doi.org/10.1037/h0100576</u>
- Dymond, S. (2014). Meaning is more than associations: Relational operants and the search for derived relations in nonhumans. *Journal of the Experimental Analysis of Behavior*, *101*(1), 152-155. https://doi.org/10.1002/jeab.57
- Dymond, S., May, R. J., Munnelly, A., & Hoon, A. E. (2010). Evaluating the evidence base for relational frame theory: A citation analysis. *The Behavior Analyst*, 33(1), 97-117. <u>https://doi.org/10.1007/bf03392206</u>

- Dymond, S., & Rehfeldt, R. A. (2000). Understanding complex behavior: The transformation of stimulus functions. *The Behavior Analyst*, 23(2), 239-254. https://doi.org/10.1007/bf03392013
- Eckard, M. L., & Lattal, K. A. (2019). The internal clock: A manifestation of a misguided mechanistic view of causation? *Perspectives on Behavior Science*, 43(1), 5-19. https://doi.org/10.1007/s40614-018-00189-5
- Ehri, L. C., & Galanis, A. H. (1980). Teaching children to comprehend propositions conjoined by "before" and "after". *Journal of Experimental Child Psychology*, 30(2), 308-324. <u>https://doi.org/10.1016/0022-0965(80)90065-x</u>
- Einstein, A. (1905). Zur Elektrodynamik bewegter Körper. Annalen der Physik, 322(10), 891-921. https://doi.org/10.1002/andp.19053221004
- Espinosa-Fernandez, L., de la Torre Vacas, L., Garcia-Viedma Mdel, R., Garcia-Gutierrez, A., & Torres Colmenero, C. J. (2004). Temporal performance in 4-8 year old children. The effect of chronometric information in task execution. *Acta Psychologica (Amst)*, *117*(3), 295-312. <u>https://doi.org/10.1016/j.actpsy.2004.08.001</u>
- Evans, V. (2013). Language and time: A cognitive linguistics approach. Cambridge University Press.
- Fayolle, S., Gil, S., & Droit-Volet, S. (2015). Fear and time: Fear speeds up the internal clock. Behavoural Processes, 120, 135-140. <u>https://doi.org/10.1016/j.beproc.2015.09.014</u>
- Ferguson, J., Craig, E. A., & Dounavi, K. (2019). Telehealth as a model for providing behaviour analytic interventions to individuals with autism spectrum disorder: A systematic review. *Journal of Autism and Developmental Disorders*, 49(2), 582-616. https://doi.org/10.1007/s10803-018-3724-5
- Ferster, C. B., & Skinner, B. F. (1957). *Schedules of reinforcement*. Appleton-Century-Crofts. https://doi.org/10.1037/10627-000
- Fields, L., Arntzen, E., Nartey, R. K., & Eilifsen, C. (2012). Effects of a meaningful, a discriminative, and a meaningless stimulus on equivalence class formation. *Journal of the Experimental Analysis of Behavior*, 97(2), 163-181. https://doi.org/10.1901/jeab.2012.97-163
- Finn, M., & De Houwer, J. (2021). The selective action of Cfunc control. *Journal of the Experimental Analysis of Behavior*, 116(3), 314-331. <u>https://doi.org/10.1002/jeab.717</u>
- Fivush, R., & Mandler, J. M. (1985). Developmental changes in the understanding of temporal sequence. *Child Development*, 56(6). <u>https://doi.org/10.2307/1130463</u>
- Flemming, T. M., Beran, M. J., & Washburn, D. A. (2007). Disconnect in concept learning by rhesus monkeys (Macaca mulatta): Judgment of relations and relations-between-relations. *Journal of Experimental Psychology. Animal Behavior Processes*, 33(1), 55-63. https://doi.org/10.1037/0097-7403.33.1.55

- Foody, M., Barnes-Holmes, Y., Barnes-Holmes, D., Törneke, N., Luciano, C., Stewart, I., & McEnteggart, C. (2014). RFT for clinical use: The example of metaphor. *Journal of Contextual Behavioral Science*, 3(4), 305-313. <u>https://doi.org/10.1016/j.jcbs.2014.08.001</u>
- French, L. A., & Nelson, K. (1985). Young children's knowledge of relational terms: Some ifs, ors, and buts. Springer-Verlag. <u>https://doi.org/10.1007/978-1-4613-8581-3</u>
- Friedman, W. J. (1977). The development of children's understanding of cyclic aspects of time. *Child Development*, 48(4). <u>https://doi.org/10.2307/1128523</u>
- Friedman, W. J. (1982). Conventional time concepts and children's structuring of time. In W. J.Friedman (Ed.), *The developmental psychology of time* (pp. 171-208). Academic Press.
- Friedman, W. J. (1986). The development of children's knowledge of temporal structure. *Child* Development, 57(6). <u>https://doi.org/10.2307/1130418</u>
- Friedman, W. J. (1990). Children's representations of the pattern of daily activities. *Child Development*, 61(5). <u>https://doi.org/10.2307/1130751</u>
- Friedman, W. J. (1991). The development of children's memory for the time of past events. *Child Development*, 62(1), 139-155. <u>https://doi.org/10.2307/1130710</u>
- Friedman, W. J. (2000). The development of children's knowledge of the times of future events. *Child Development*, 71(4), 913-932. <u>https://doi.org/10.1111/1467-8624.00199</u>
- Friedman, W. J. (2002). Children's knowledge of the future distances of daily activities and annual events. *Journal of Cognition and Development*, 3(3), 333-356. https://doi.org/10.1207/s15327647jcd0303\_4
- Friedman, W. J., Gardner, A. C., & Zubin, N. R. E. (1995). Children's comparisons of the recency of two events from the past year. *Child Development*, 66(4), 970-983. <u>https://doi.org/10.2307/1131792</u>
- Friedman, W. J., & Kemp, S. (1998). The effects of elapsed time and retrieval on young children's judgments of the temporal distances of past events. *Cognitive Development*, 13(3), 335-367. https://doi.org/10.1016/s0885-2014(98)90015-6
- Friedman, W. J., & Laycock, F. (1989). Children's analog and digital clock knowledge. *Child Development*, 60(2). <u>https://doi.org/10.2307/1130982</u>
- Fuhrman, O., & Boroditsky, L. (2010). Cross-cultural differences in mental representations of time: Evidence from an implicit nonlinguistic task. *Cognitive Science*, 34(8), 1430-1451. <u>https://doi.org/10.1111/j.1551-6709.2010.01105.x</u>
- Gable, P. A., & Poole, B. D. (2012). Time flies when you're having approach-motivated fun: Effects of motivational intensity on time perception. *Psychological Science*, 23(8), 879-886. <u>https://doi.org/10.1177/0956797611435817</u>
- Gale, R. (2016). The philosophy of time: A collection of essays. Springer.

- Galizio, M., & Bruce, K. E. (2018). Abstraction, multiple exemplar training and the search for derived stimulus relations in animals. *Perspectives on Behavior Science*, 41(1), 45-67. https://doi.org/10.1007/s40614-017-0112-y
- Gauvain, M., & Rogoff, B. (1989). Collaborative problem solving and children's planning skills. *Developmental Psychology*, 25(1), 139-151. <u>https://doi.org/10.1037/0012-1649.25.1.139</u>
- Ghirlanda, S., Lind, J., & Enquist, M. (2017). Memory for stimulus sequences: A divide between humans and other animals? *Royal Society Open Science*, 4(6), 161011. <u>https://doi.org/10.1098/rsos.161011</u>
- Gibbon, J. (1977). Scalar expectancy theory and Weber's law in animal timing. *Psychological Review*, 84(3), 279-325. <u>https://doi.org/10.1037/0033-295x.84.3.279</u>
- Gibson, J. L., Pennington, R. C., Stenhoff, D. M., & Hopper, J. S. (2010). Using desktop videoconferencing to deliver interventions to a preschool student with autism. *Topics in Early Childhood Special Education*, 29(4), 214-225. <u>https://doi.org/10.1177/0271121409352873</u>
- Gifford, E. V., & Hayes, S. C. (1999). Functional contextualism: A pragmatic philosophy for behavior science In W. O'Donohue & R. Kitchener (Eds.), *Handbook of behaviorism* (pp. 285-327). Academic Press.
- Gil, S., Niedenthal, P. M., & Droit-Volet, S. (2007). Anger and time perception in children. *Emotion*, 7(1), 219-225. <u>https://doi.org/10.1037/1528-3542.7.1.219</u>
- Gilroy, S. P., Lorah, E. R., Dodge, J., & Fiorello, C. (2015). Establishing deictic repertoires in autism. *Research in Autism Spectrum Disorders*, 19, 82-92. https://doi.org/10.1016/j.rasd.2015.04.004
- Giurfa, M., Zhang, S., Jenett, A., Menzel, R., & Srinivasan, M. V. (2001). The concepts of 'sameness' and 'difference' in an insect. *Nature*, *410*(6831), 930-933. <u>https://doi.org/10.1038/35073582</u>
- Gore, N. J., Barnes-Holmes, Y., & Murphy, G. (2010). The relationship between intellectual functioning and relational perspective-taking. *International Journal of Psychology and Psychological Therapy*, 10(1), 1-17.
- Greer, R. D., & Ross, D. E. (2008). *Verbal behavior analysis: Inducing and expanding new verbal capabilities in children with language delays.* Pearson Education, Inc.
- Grey, I., Healy, O., Leader, G., & Hayes, D. (2009). Using a Time Timer to increase appropriate waiting behavior in a child with developmental disabilities. *Research in Developmental Disabilities*, 30(2), 359-366. <u>https://doi.org/10.1016/j.ridd.2008.07.001</u>
- Grondin, S. (2008). Psychology of time (1st ed.). Emerald.
- Grondin, S. (2010). Timing and time perception: A review of recent behavioral and neuroscience findings and theoretical directions. *Attention, Perception, & Psychophysics*, 72(3), 561-582. <u>https://doi.org/10.3758/APP.72.3.561</u>
- Grünbaum, A. (1974). Philosophical problems of space and time. D. Reidel Publishing Company.

- Hanley, G. P., Heal, N. A., Tiger, J. H., & Ingvarsson, E. T. (2007). Evaluation of a class wide teaching program for developing preschool life skills. *Journal of Applied Behavior Analysis*, 40(2), 277-300. https://doi.org/10.1901/jaba.2007.57-06
- Harmon, K., Strong, R., & Pasnak, R. (1982). Relational responses in tests of transposition with Rhesus monkeys. *Learning and Motivation*, 13(4), 495-504. <u>https://doi.org/10.1016/0023-9690(82)90006-6</u>
- Harte, C., & Barnes-Holmes, D. (2021). A primer on relational frame theory. In *The Oxford handbook* of acceptance and commitment therapy. https://doi.org/10.1093/oxfordhb/9780197550076.013.4
- Harte, C., Barnes-Holmes, D., Barnes-Holmes, Y., & Kissi, A. (2020). The study of rule-governed behavior and derived stimulus relations: Bridging the gap. *Perspectives on Behavior Science*, 43(2), 361-385. https://doi.org/10.1007/s40614-020-00256-w
- Hayes, J., & Stewart, I. (2016). Comparing the effects of derived relational training and computer coding on intellectual potential in school-age children. *British Journal of Educational Psychology*, 86(3), 397-411. <u>https://doi.org/10.1111/bjep.12114</u>
- Hayes, S. C. (1989). Nonhumans have not yet shown stimulus equivalence. *Journal of the Experimental Analysis of Behavior*, 51(3), 385-392. <u>https://doi.org/10.1901/jeab.1989.51-385</u>
- Hayes, S. C. (1992). Verbal relations, time, and suicide. In *Understanding verbal relations* (pp. 109-118). Context Press.
- Hayes, S. C. (1993). Analytic goals and the varieties of scientific contextualism. In S. C. Hayes, L. J.
  Hayes, H. W. Reese, & T. R. Sarbin (Eds.), *Varieties of scientific contextualism* (pp. 1–27).
  Context Press.
- Hayes, S. C. (1994). Relational frame theory: A functional approach to verbal events. In S. C. Hayes, M. Sato, & K. Ono (Eds.), *Behavior analysis of language and cognition* (pp. 9-30). Context Press.
- Hayes, S. C., Barnes-Holmes, D., & Roche, B. (2001). *Relational frame theory: A post-Skinnerian* account of human language and cognition. Kluwer Academic/Plenum.
- Hayes, S. C., & Brownstein, A. J. (1986). Mentalism, behavior-behavior relations, and a behavioranalytic view of the purposes of science. *The Behavior Analyst*, 9(2), 175-190. <u>https://doi.org/10.1007/bf03391944</u>
- Hayes, S. C., Brownstein, A. J., Haas, J. R., & Greenway, D. E. (1986). Instructions, multiple schedules, and extinction: Distinguishing rule-governed from schedule-controlled behavior. *Journal of the Experimental Analysis of Behavior*, 46(2), 137-147.
   <a href="https://doi.org/10.1901/jeab.1986.46-137">https://doi.org/10.1901/jeab.1986.46-137</a>
- Hayes, S. C., Fox, E., Gifford, E. V., Wilson, K. G., Barnes-Holmes, D., & Healy, O. (2001). Derived relational responding as learned behavior. In S. C. Hayes, D. Barnes-Holmes, & B. Roche

(Eds.), *Relational frame theory: A post- Skinnerian account of human language and cognition* (pp. 21-49). Kluwer Academic/Plenum.

- Hayes, S. C., Gifford, E. V., & Ruckstuhl Jr., L. E. (1996). Relational frame theory and executive function: A behavioral approach. In G. R. Lyon & N. A. Krasnegor (Eds.), *Attention, memory, and executive function* (pp. 297-305). Paul Brookes Publishing Co.
- Hayes, S. C., Gifford, E. V., Townsend, R. C. J., & Barnes-Holmes, D. (2001). Thinking, problem solving, and pragmatic verbal analysis. In S. C. Hayes, D. Barnes-Holmes, & B. Roche (Eds.), *Relational frame theory: A post- Skinnerian account of human language and cognition* (pp. 87-101). Kluwer Academic/Plenum.
- Hayes, S. C., & Hayes, L. J. (1989). The verbal action of the listener as a basis for rule-governance. In S. C. Hayes (Ed.), *Rule-governed behavior: Cognition, contingencies, and instructional control* (pp. 153-190). Plenum.
- Hayes, S. C., Hayes, L. J., & Reese, H. W. (1988). Finding the philosophical core: A review of Stephen C. Pepper's world hypotheses: A study in evidence. *Journal of the Experimental Analysis of Behavior*, 50(1), 97-111. <u>https://doi.org/10.1901/jeab.1988.50-97</u>
- Hayes, S. C., Masuda, A., Shenk, C., Yadavaia, J. E., Boulanger, J., Vilardaga, R., Plumb, J., Fletcher, L., Bunting, K., Levin, M. E., Waltz, T., & Hildebrandt, M. J. (2007). Applied extensions of behavior principles: Applied behavioral concepts and behavioral theories. In D. W. Woods & J. W. Kanter (Eds.), *Understanding behavior disorders: A contemporary behavioral perspective* (pp. 47-80). New Harbinger Publications, Inc.
- Hayes, S. C., Wilson, K. G., & Gifford, E. V. (1999). Consciousness and Private Events. In B. A. Thyer (Ed.), *The Philosophical Legacy of Behaviorism* (pp. 153-187). Kluwer Academic Publishers.
- Healy, O., Barnes-Holmes, D., & Smeets, P. M. (2000). Derived relational responding as generalized operant behavior. *Journal of the Experimental Analysis of Behavior*, 74(2), 207-227. <u>https://doi.org/10.1901/jeab.2000.74-207</u>
- Hendricks, R. K., & Boroditsky, L. (2017). New space-time metaphors foster new nonlinguistic representations. *Topics in Cognitive Science*, 9(3), 800-818. <u>https://doi.org/10.1111/tops.12279</u>
- Hershfield, H. E., Goldstein, D. G., Sharpe, W. F., Fox, J., Yeykelis, L., Carstensen, L. L., & Bailenson, J. N. (2011). Increasing saving behavior through age-progressed renderings of the future self. *Journal of Marketing Research*, 48, S23-S37. https://doi.org/10.1509/jmkr.48.SPL.S23
- Horner, R. D., & Baer, D. M. (1978). Multiple-probe technique: A variation on the multiple baseline. Journal of Applied Behavior Analysis, 11(1), 189-196. <u>https://doi.org/10.1901/jaba.1978.11-189</u>

- Hughes, S., & Barnes-Holmes, D. (2013). A functional approach to the study of implicit cognition:
  The IRAP and the REC model. In S. Dymond & B. Roche (Eds.), *Advances in relational frame theory* (pp. 96-125). Context Press/New Harbinger Publications.
- Hughes, S., & Barnes-Holmes, D. (2014). Associative concept learning, stimulus equivalence, and relational frame theory: working out the similarities and differences between human and nonhuman behavior. *Journal of the Experimental Analysis of Behavior*, 101(1), 156-160. https://doi.org/10.1002/jeab.60
- Hughes, S., & Barnes-Holmes, D. (2016). Relational frame theory: The basic account. In R. Zettle, S. C. Hayes, D. Barnes-Holmes, & A. Biglan (Eds.), *The wiley handbook of contextual behavioral science* (pp. 129-179). Wiley Blackwell.
- Hyland, J. M., O'Hora, D. P., Leslie, J. C., & Smyth, S. (2012). Sequential responding in accordance with temporal relational cues: A comparison of before and after. *The Psychological Record*, 62(3), 463-484. https://doi.org/10.1007/bf03395814
- Hyland, J. M., Smyth, S., O'Hora, D. P., & Leslie, J. C. (2014). The effect of before and after instructions on the speed of sequential responding. *The Psychological Record*, 64(2), 311-319. <u>https://doi.org/10.1007/s40732-014-0026-y</u>
- Kaller, C. P., Rahm, B., Spreer, J., Mader, I., & Unterrainer, J. M. (2008). Thinking around the corner: The development of planning abilities. *Brain and Cognition*, 67(3), 360-370. <u>https://doi.org/10.1016/j.bandc.2008.02.003</u>
- Keller-Cohen, D. (1987). Context and strategy in acquiring temporal connectives. Journal of Psycholinguistic Research, 16(2), 165-183. <u>https://doi.org/10.1007/bf01072000</u>
- Kelly, A., & Barnes-Holmes, D. (2013). Implicit attitudes towards children with autism versus normally developing children as predictors of professional burnout and psychopathology. *Res Dev Disabil*, 34(1), 17-28. https://doi.org/10.1016/j.ridd.2012.07.018
- Kenny, N., Barnes-Holmes, D., & Stewart, I. (2014). Competing arbitrary and non-arbitrary relational responding in normally developing children and children diagnosed with autism. *The Psychological Record*, 64(4), 755-768. <u>https://doi.org/10.1007/s40732-014-0042-y</u>
- Kenny, N., Devlin, S., Barnes-Holmes, D., Barnes-Holmes, Y., & Stewart, I. (2014). Competing arbitrary and nonarbitrary stimulus relations: The effect of exemplar training in adult participants. *The Psychological Record*, 64(1), 53-61. <u>https://doi.org/10.1007/s40732-014-0004-4</u>
- Kent, G., Galvin, E., Barnes-Holmes, Y., Murphy, C., & Barnes-Holmes, D. (2017). Relational responding: Testing, training, and sequencing effects among children with autism and typically developing children. *Behavioral Development Bulletin*, 22(1), 94-110. https://doi.org/10.1037/bdb0000041
- Killeen, P. R., & Fetterman, J. G. (1988). A behavioral theory of timing. *Psychological Review*, 95(2), 274-295. <u>https://doi.org/10.1037/0033-295x.95.2.274</u>

- Kirsten, E. B., & Stewart, I. (2022). Assessing the development of relational framing in young children. *The Psychological Record*, 72, 221-246. <u>https://doi.org/10.1007/s40732-021-00457-</u> y
- Kirsten, E. B., Stewart, I., & McElwee, J. (2022a). Testing and training analogical relational responding in children with and without autism. *The Psychological Record*, 72, 561-583. <u>https://doi.org/10.1007/s40732-021-00493-8</u>
- Kirsten, E. B., Stewart, I., & McElwee, J. (2022b). Testing and training analogical responding in young children using a relational evaluation procedure. *The Psychological Record*, 72, 353-369. <u>https://doi.org/10.1007/s40732-021-00468-9</u>
- Kramer, R. S., Weger, U. W., & Sharma, D. (2013). The effect of mindfulness meditation on time perception. *Consciousness and Cognition*, 22(3), 846-852. <u>https://doi.org/10.1016/j.concog.2013.05.008</u>
- Kuo, H. C., Lee, C. C., & Chiou, W. B. (2016). The power of the virtual ideal self in weight control: Weight-reduced avatars can enhance the tendency to delay gratification and regulate dietary practices. *Cyberpsychology, Behavior, and Social Networking*, 19(2), 80-85.
   <u>https://doi.org/10.1089/cyber.2015.0203</u>
- Lamotte, M., Izaute, M., & Droit-Volet, S. (2012). Awareness of time distortions and its relation with time judgment: A metacognitive approach. *Consciousness and Cognition*, 21(2), 835-842. <u>https://doi.org/10.1016/j.concog.2012.02.012</u>
- Leander, J. D., Lippman, L. G., & Meyer, M. E. (1968). Fixed interval performance as related to subjects' verbalizations of the reinforcement contingency. *The Psychological Record*, 18(3), 469-474. <u>https://doi.org/10.1007/bf03393795</u>
- Ledford, J. R., & Gast., D. L. (2018). Single case research methodology: Applications in special education and behavioral sciences (3rd ed.). Routledge.
- Lejeune, H., Pouthas, V., & el Ahmadi, A. (1992). Temporal differentiation of response duration in 4½- and 6-year-olds: Role of verbal instructions versus demonstration. *Cahiers de Psychologie Cognitive/Current Psychology of Cognition*, 12(4), 315-332.
- Lejeune, H., Richelle, M., & Wearden, J. H. (2006). About Skinner and time: Behavior-analytic contributions to research on animal timing. *Journal of the Experimental Analysis of Behavior*, 85(1), 125-142. <u>https://doi.org/10.1901/jeab.2006.85.04</u>
- Levin, I., & Wilkening, F. (1989). Measurng time via counting: The development of children's conceptions of time as a quantifiable dimension. In I. Levin & D. Zakay (Eds.), *Time and human cognition: A lifespan perspective* (pp. 119-144). <u>https://doi.org/10.1016/S0166-4115(08)61040-7</u>
- Levin, I., & Zakay, D. (1989). *Time and human cognition: A life-span perspective*. Elsevier Science Publishers.

- Levin, M. E., Twohig, M. P., & Smith, B. M. (2016). Contextual Behavioral Science: An overview. In R. Zettle, S. C. Hayes, D. Barnes-Holmes, & A. Biglan (Eds.), *The wiley handbook of contextual behavioral science* (pp. 17-36). Wiley Blackwell.
- Lindgren, S., Wacker, D., Suess, A., Schieltz, K., Pelzel, K., Kopelman, T., Lee, J., Romani, P., & Waldron, D. (2016). Telehealth and autism: Treating challenging behavior at lower cost. *Pediatrics*, 137(2), S167-175. https://doi.org/10.1542/peds.2015-28510
- Lionello-DeNolf, K. M. (2009). The search for symmetry: 25 years in review. *Learning & Behavior*, 37(2), 188-203. https://doi.org/10.3758/LB.37.2.188
- Lipkens, R., Hayes, S. C., & Hayes, L. J. (1993). Longitudinal study of the development of derived relations in an infant. *Journal of Experimental Child Psychology*, 56(2), 201-239. <u>https://doi.org/10.1006/jecp.1993.1032</u>
- Lippman, L. G., & Meyer, M. E. (1967). Fixed interval performance as related to instructions and to subjects' verbalizations of the contingency. *Psychonomic Science*, 8(4), 135-136. <u>https://doi.org/10.3758/bf03331586</u>
- Liu, L., Feng, T., Chen, J., & Li, H. (2013). The value of emotion: How does episodic prospection modulate delay discounting? *PLoS One*, 8(11), e81717. https://doi.org/10.1371/journal.pone.0081717
- Lowe, C. F., Beasty, A., & Bentall, R. P. (1983). The role of verbal behavior in human learning: Infant performance on fixed-interval schedules. *Journal of the Experimental Analysis of Behavior*, 39(1), 157-164. <u>https://doi.org/10.1901/jeab.1983.39-157</u>
- Lowe, C. F., Harzem, P., & Bagshaw, M. (1978). Species differences in temporal control of behavior II: Human performance. *Journal of the Experimental Analysis of Behavior*, 29(3), 351-361. <u>https://doi.org/10.1901/jeab.1978.29-351</u>
- Luciana, M., & Nelson, C. A. (1998). The functional emergence of prefrontally-guided working memory systems in four- to eight-year-old children. *Neuropsychologia*, 36(3), 273-293. <u>https://doi.org/10.1016/s0028-3932(97)00109-7</u>
- Luciano, C., Rodríguez, M., Mañas, I., Ruiz, F., Berens, N. M., & Valdivia-Salas, S. (2009).
   Acquiring the earliest relational operants: Coordination, distinction, opposition, comparison, and hierarchy. In R. A. Rehfeldt & Y. Barnes-Holmes (Eds.), *Derived relational responding applications for learners with autism and other developmental disabilities* (pp. 149-172).
   New Harbinger Publications, Inc.
- Machado, A. (1997). Learning the temporal dynamics of behavior. *Psychological Review*, *104*(2), 241-265. https://doi.org/10.1037/0033-295x.104.2.241
- Machado, A., Malheiro, M. T., & Erlhagen, W. (2009). Learning to time: A perspective. Journal of the Experimental Analysis of Behavior, 92(3), 423-458. <u>https://doi.org/10.1901/jeab.2009.92-423</u>

- Madden, G. J., & Bickel, W. K. (2010). *Impulsivity: The behavioral and neurological science of discounting*. American Psychological Association.
- Mahy, C. E. V., Moses, L. J., O'Brien, B., Castro, A. W., Kopp, L., & Atance, C. M. (2020). The roles of perspective and language in children's ability to delay gratification. *Journal of Experimental Child Psychology*, 192, 104767. <u>https://doi.org/10.1016/j.jecp.2019.104767</u>
- Marano-Frezza, K. E., Vladescu, J. C., Reeve, K. F., Sidener, T. M., & Cox, D. J. (2022). Training staff to create equivalence-based instruction materials in Qualtrics. *The Psychological Record*, 72(4), 647-663. <u>https://doi.org/10.1007/s40732-021-00497-4</u>
- Matthews, B. A., Shimoff, E., Catania, A. C., & Sagvolden, T. (1977). Uninstructed human responding: Sensitivity to ratio and interval contingencies. *Journal of the Experimental Analysis of Behavior*, 27(3), 453-467. <u>https://doi.org/10.1901/jeab.1977.27-453</u>
- Mazachowsky, T. R., & Mahy, C. E. V. (2020). Constructing the children's future thinking questionnaire: A reliable and valid measure of children's future-oriented cognition. *Developmental Psychology*, 56(4), 756-772. <u>https://doi.org/10.1037/dev0000885</u>
- McColgan, K. L., & McCormack, T. (2008). Searching and planning: Young children's reasoning about past and future event sequences. *Child Development*, 79(5), 1477-1497. https://doi.org/10.1111/j.1467-8624.2008.01200.x
- McCormack, T. (2015). The development of temporal cognition. In *Handbook of child psychology* and developmental science (pp. 1-47). <u>https://doi.org/10.1002/9781118963418.childpsy215</u>
- McCormack, T., & Atance, C. M. (2011). Planning in young children: A review and synthesis. *Developmental Review*, 31(1), 1-31. <u>https://doi.org/10.1016/j.dr.2011.02.002</u>
- McCormack, T., Brown, G. D. A., Maylor, E. A., Darby, R. J., & Green, D. (1999). Developmental changes in time estimation: Comparing childhood and old age. *Developmental Psychology*, 35(4), 1143-1155. https://doi.org/10.1037/0012-1649.35.4.1143
- McCormack, T., & Hanley, M. (2011). Children's reasoning about the temporal order of past and future events. *Cognitive Development*, 26(4), 299-314. https://doi.org/10.1016/j.cogdev.2011.10.001
- McCormack, T., & Hoerl, C. (1999). Memory and temporal perspective: The role of temporal frameworks in memory development. *Developmental Review*, *19*(1), 154-182. <u>https://doi.org/10.1006/drev.1998.0476</u>
- McCormack, T., & Hoerl, C. (2017). The development of temporal concepts: Learning to locate events in time. *Timing & Time Perception*, 5(3-4), 297-327. <u>https://doi.org/10.1163/22134468-00002094</u>
- McCormack, T., & Hoerl, C. (2020). Children's future-oriented cognition. In *Advances in Child* Development and Behavior (2020/03/15 ed., Vol. 58, pp. 215-253). https://doi.org/10.1016/bs.acdb.2020.01.008

- McGreal, C., Hyland, J., O' Hora, D., & Hogan, M. (2016). Mutual entailment of temporal relations in younger and older adults: Reversing order judgments. *The Psychological Record*, 66(3), 419-428. https://doi.org/10.1007/s40732-016-0182-3
- McHugh, L., Barnes-Holmes, Y., & Barnes-Holmes, D. (2004). Perspective-taking as relational responding: A developmental profile. *The Psychological Record*, 54(1), 115-144. https://doi.org/10.1007/bf03395465
- McHugh, L., Barnes-Holmes, Y., & Barnes-Holmes, D. (2009). Understanding and training perspective taking as relational responding. In R. A. Rehfeldt & Y. Barnes-Holmes (Eds.), *Derived relational responding applications for learners with autism and other developmental disabilities: A progressive guide to change* (pp. 281-300). Context Press/New Harbinger Publications.
- McHugh, L., & Stewart, I. (2012). *The self and perspective taking: Contributions and applications from modern behavioral science*. New Harbinger Publications.
- McIlvane, W. J. (2014). "Associative concept learning in animals" by Zentall, Wasserman, and Urcuioli: A commentary. *Journal of the Experimental Analysis of Behavior*, *101*(1), 161-164. <u>https://doi.org/10.1002/jeab.59</u>
- McKeel, A. N., & Dixon, M. R. (2014). Furthering a behavior analytic account of self-control using relational frame theory. *Behavioral Development Bulletin*, 19(2), 111-118. <u>https://doi.org/10.1037/h0100581</u>
- McLay, L. K., Sutherland, D., Church, J., & Tyler-Merrick, G. (2013). The formation of equivalence classes in individuals with autism spectrum disorder: A review of the literature. *Research in Autism Spectrum Disorders*, 7(2), 418-431. <u>https://doi.org/10.1016/j.rasd.2012.11.002</u>
- McLoughlin, S., Tyndall, I., & Pereira, A. (2020). Convergence of multiple fields on a relational reasoning approach to cognition. *Intelligence*, 83. <u>https://doi.org/10.1016/j.intell.2020.101491</u>
- McLoughlin, S., Tyndall, I., Pereira, A., & Mulhern, T. (2020). Non-verbal IQ gains from relational operant training explain variance in educational attainment: An active-controlled feasibility study. *Journal of Cognitive Enhancement*, 5(1), 35-50. <u>https://doi.org/10.1007/s41465-020-00187-z</u>
- McTaggart, J. E. (1908). The unreality of time. *Mind*, *XVII*(4), 457-474. https://doi.org/10.1093/mind/XVII.4.457
- Miller, D. T., Weinstein, S. M., & Karniol, R. (1978). Effects of age and self-verbalization on children's ability to delay gratification. *Developmental Psychology*, 14(5), 569-570. <u>https://doi.org/10.1037/0012-1649.14.5.569</u>
- Ming, S., Mulhern, T., Stewart, I., Moran, L., & Bynum, K. (2018). Training class inclusion responding in typically developing children and individuals with autism. *Journal of Applied Behavior Analysis*, 51(1), 53-60. <u>https://doi.org/10.1002/jaba.429</u>

- Ming, S., & Stewart, I. (2017). When things are not the same: A review of research into relations of difference. *Journal of Applied Behavior Analysis*, 50(2), 429-455. https://doi.org/10.1002/jaba.367
- Ming, S., Stewart, I., McElwee, J., & Bynum, K. (2015). Contextual control over derived relational responding in a teenager with autism. *Research in Autism Spectrum Disorders*, 19, 7-17. <u>https://doi.org/10.1016/j.rasd.2015.03.003</u>
- Mischel, H. N., & Mischel, W. (1983). The development of children's knowledge of self-control strategies. *Child Development*, 54(3). <u>https://doi.org/10.2307/1130047</u>
- Mischel, W., Shoda, Y., & Rodriguez, M. I. (1989). Delay of gratification in children. *Science*, 244(4907), 933-938. <u>https://doi.org/10.1126/science.2658056</u>
- Montoya-Rodríguez, M. M., Molina, F. J., & McHugh, L. (2017). A review of relational frame theory research into deictic relational responding. *The Psychological Record*, 67(4), 569-579. https://doi.org/10.1007/s40732-016-0216-x
- Moore, B. D., Brooks, P. J., & Rabin, L. A. (2014). Comparison of diachronic thinking and event ordering in 5- to 10-year-old children. *International Journal of Behavioral Development*, 38(3), 282-292. <u>https://doi.org/10.1177/0165025414520806</u>
- Moore, C., Barresi, J., & Thompson, C. (1998). The cognitive basis of future-oriented prosocial behavior. *Social Development*, 7(2), 198-218. <u>https://doi.org/10.1111/1467-9507.00062</u>
- Moran, L., Stewart, I., McElwee, J., & Ming, S. (2014). Relational ability and language performance in children with autism spectrum disorders and typically developing children: A further test of the TARPA protocol. *The Psychological Record*, 64(2), 233-251. https://doi.org/10.1007/s40732-014-0032-0
- Moran, L., Walsh, L., Stewart, I., McElwee, J., & Ming, S. (2015). Correlating derived relational responding with linguistic and cognitive ability in children with autism spectrum disorders. *Research in Autism Spectrum Disorders*, 19, 32-43. https://doi.org/10.1016/j.rasd.2014.12.015
- Mulhern, T. (2023). The relational frame of temporality. In T. Mulhern (Ed.), *Relational frame theory made simple* (pp. 131-152). Spring Nature Switzerland AG.
- Mulhern, T., Stewart, I., & Elwee, J. M. (2017). Investigating relational framing of categorization in young children. *The Psychological Record*, 67(4), 519-536. <u>https://doi.org/10.1007/s40732-017-0255-y</u>
- Mulhern, T., Stewart, I., & McElwee, J. (2018). Facilitating relational framing of classification in young children. *Journal of Contextual Behavioral Science*, 8, 55-68. <u>https://doi.org/10.1016/j.jcbs.2018.04.001</u>
- Murphy, C., Lyons, K., Kelly, M., Barnes-Holmes, Y., & Barnes-Holmes, D. (2019). Using the Teacher IRAP (T-IRAP) interactive computerized programme to teach complex flexible

relational responding with children with diagnosed autism spectrum disorder. *Behavior Analysis in Practice*, *12*(1), 52-65. https://doi.org/10.1007/s40617-018-00302-9

- Najdowski, A. C. (2017). Flexible and focused: Teaching executive function skills to individuals with autism and attention disorders. Elsevier Inc.
- Neely, L., Rispoli, M., Gerow, S., Hong, E. R., & Hagan-Burke, S. (2017). Fidelity outcomes for autism-focused interventionists coached via telepractice: A systematic literature review. *Journal of Developmental and Physical Disabilities*, 29(6), 849-874. https://doi.org/10.1007/s10882-017-9550-4
- Nelson, K. (1996a). Emergence of the temporal mind. In *Language in cognitive development: The emergence of the mediated mind* (pp. 259-291). Cambridge University Press.
- Nelson, K. (1996b). *Language in cognitive development: Emergence of the mediated mind*. https://doi.org/10.1017/CBO9781139174619
- Nelson, K. (2007a). Developing past and future selves for time travel narratives. *Behavioral and Brain Sciences*, *30*(3), 327-328. <u>https://doi.org/10.1017/s0140525x07002130</u>
- Nelson, K. (2007b). Finding oneself in time. In Young minds in social worlds: Experience, meaning, and memory (pp. 179-208). Harvard University Press. <u>https://doi.org/10.4159/9780674041400</u>
- Neufeld, J., & Stewart, I. (2023). A behavioral approach to the human understanding of time: Relational frame theory and temporal relational framing. *The Psychological Record*, 73(2), 301-332. <u>https://doi.org/10.1007/s40732-022-00529-7</u>
- Neufeld, J., Stewart, I., & McElwee, J. (2023). Assessing temporal relational responding in young children. *The Psychological Record*, 73(2), 163-182. <u>https://doi.org/10.1007/s40732-023-00534-4</u>
- Newquist, M. H., Dozier, C. L., & Neidert, P. L. (2012). A comparison of the effects of brief rules, a timer, and preferred toys on self-control. *Journal of Applied Behavior Analysis*, 45(3), 497-509. <u>https://doi.org/10.1901/jaba.2012.45-497</u>
- Newsome, K. B., Berens, K. N., Ghezzi, P. M., Aninao, T., & Newsome, W. D. (2014). Training relational language to improve reading comprehension. *European Journal of Behavior Analysis*, 15(2), 165-197. https://doi.org/10.1080/15021149.2014.11434512
- O'Hora, D., & Maglieri, K. A. (2006). Goal statements and goal-directed behavior. *Journal of Organizational Behavior Management*, 26(1-2), 131-170.
   <a href="https://doi.org/10.1300/J075v26n01\_06">https://doi.org/10.1300/J075v26n01\_06</a>
- O'Connor, J., Barnes-Holmes, Y., & Barnes-Holmes, D. (2011). Establishing contextual control over symmetry and asymmetry performances in typically developing children and children with autism. *The Psychological Record*, 61(2), 287-312. <u>https://doi.org/10.1007/bf03395761</u>
- O'Connor, J., Rafferty, A., Barnes-Holmes, D., & Barnes-Holmes, Y. (2017). The role of verbal behavior, stimulus nameability, and familiarity on the equivalence performances of autistic

and normally developing children. *The Psychological Record*, *59*(1), 53-74. https://doi.org/10.1007/bf03395649

- O'Connor, M., Farrell, L., Munnelly, A., & McHugh, L. (2017). Citation analysis of relational frame theory: 2009–2016. *Journal of Contextual Behavioral Science*, 6(2), 152-158. <u>https://doi.org/10.1016/j.jcbs.2017.04.009</u>
- O'Hora, D., Barnes-Holmes, D., Roche, B., & Smeets, P. (2004). Derived relational networks and control by novel instructions: A possible model of generative verbal responding. *The Psychological Record*, 54(3), 437-460. <u>https://doi.org/10.1007/bf03395484</u>
- O'Hora, D., Barnes-Holmes, D., & Stewart, I. (2014). Antecedent and consequential control of derived instruction-following. *Journal of the Experimental Analysis of Behavior*, *102*(1), 66-85. <u>https://doi.org/10.1002/jeab.95</u>
- O'Hora, D., Pelaez, M., & Barnes-Holmes, D. (2005). Derived relational responding and performance on verbal subtests of the WAIS-III. *The Psychological Record*, 55(1), 155-175. <u>https://doi.org/10.1007/bf03395504</u>
- O'Hora, D., Peláez, M., Barnes-Holmes, D., Rae, G., Robinson, K., & Chaudhary, T. (2008). Temporal relations and intelligence: Correlating relational performance with performance on the WAIS-III. *The Psychological Record*, *58*(4), 569-584. <u>https://doi.org/10.1007/bf03395638</u>
- O'Toole, C., & Barnes-Holmes, D. (2009). Three chronometric indices of relational responding as predictors of performance on a brief intelligence test: The importance of relational flexibility. *The Psychological Record*, *59*(1), 119-132. https://doi.org/10.1007/bf03395652
- Overweg, J., Hartman, C. A., & Hendriks, P. (2018). Temporarily out of order: Temporal perspective taking in language in children with autism spectrum disorder. *Frontiers in Psychology*, 9, 1663. <u>https://doi.org/10.3389/fpsyg.2018.01663</u>
- Pepper, S. C. (1942). World hypotheses: A study in evidence. University of California Press.
- Perez, W. F., de Almeida, J. H., & de Rose, J. C. (2015). Transformation of meaning through relations of sameness and opposition. *The Psychological Record*, 65(4), 679-689. <u>https://doi.org/10.1007/s40732-015-0138-z</u>
- Perner, J. (2001). Episodic memory: Essential distinctions and developmental implications. In C.
   Moore & K. Lemmons (Eds.), *The self in time: Developmental perspectives* (pp. 181-202).
   Lawrence Erlbaum Associates Publishers.
- Peters, J., & Buchel, C. (2010). Episodic future thinking reduces reward delay discounting through an enhancement of prefrontal-mediotemporal interactions. *Neuron*, 66(1), 138-148. <u>https://doi.org/10.1016/j.neuron.2010.03.026</u>
- Piaget, J. (1971). The child's conception of time. Ballantine Books.
- Plumb, J. C., Stewart, I., Dahl, J., & Lundgren, T. (2009). In search of meaning: Values in modern clinical behavior analysis. *The Behavior Analyst*, 32(1), 85-103. <u>https://doi.org/10.1007/BF03392177</u>

- Pouthas, V. (1985). Timing behavior in young children: A developmental approach to conditioned spaced responding. In J. A. Michon & J. L. Jackson (Eds.), *Time, mind, and behavior* (pp. 112-130). Springer-Verlag.
- Pouthas, V., Droit, S., & Jacquet, A.-Y. (1993). Temporal experiences and time knowledge in infancy and early childhood. *Time & Society*, 2(2), 199-218. https://doi.org/10.1177/0961463x93002002004
- Pouthas, V., Droit, S., Jacquet, A. Y., & Wearden, J. H. (1990). Temporal differentiation of response duration in children of different ages: Developmental changes in relations between verbal and nonverbal behavior. *Journal of the Experimental Analysis of Behavior*, 53(1), 21-31. <u>https://doi.org/10.1901/jeab.1990.53-21</u>
- Pouthas, V., & Jacquet, A.-Y. (1987). A developmental study of timing behavior in 4 1/2- and 7-yearold children. *Journal of Experimental Child Psychology*, 43(2), 282-299. <u>https://doi.org/10.1016/0022-0965(87)90064-6</u>
- Prencipe, A., & Zelazo, P. D. (2005). Development of affective decision making for self and other: Evidence for the integration of first- and third-person perspectives. *Psychological Science*, *16*(7), 501-505. <u>https://doi.org/10.1111/j.0956-7976.2005.01564.x</u>
- Pyykkönen, P., & Järvikivi, J. (2012). Children and situation models of multiple events. Developmental Psychology, 48(2), 521-529. <u>https://doi.org/10.1037/a0025526</u>
- Rehfeldt, R. A., Dillen, J. E., Ziomek, M. M., & Kowalchuk, R. K. (2007). Assessing relational learning deficits in perspective-taking in children with high-functioning autism spectrum disorder. *The Psychological Record*, 57(1), 23-47. <u>https://doi.org/10.1007/bf03395563</u>
- Rehfeldt, R. A., & Root, S. L. (2005). Establishing derived requesting skills in adults with severe developmental disabilities. *Journal of Applied Behavior Analysis*, 38(1), 101-105. https://doi.org/10.1901/jaba.2005.106-03
- Rosenfarb, I. S., Newland, M. C., Brannon, S. E., & Howey, D. S. (1992). Effects of self-generated rules on the development of schedule-controlled behavior. *Journal of the Experimental Analysis of Behavior*, 58(1), 107-121. https://doi.org/10.1901/jeab.1992.58-107
- Scholten, H., Scheres, A., de Water, E., Graf, U., Granic, I., & Luijten, M. (2019). Behavioral trainings and manipulations to reduce delay discounting: A systematic review. *Psychonomic Bulletin & Review*, 26(6), 1803-1849. <u>https://doi.org/10.3758/s13423-019-01629-2</u>
- Schusterman, R. J., & Kastak, D. (1993). A California sea lion (zalophus californianus) is capable of forming equivalence relations. *The Psychological Record*, 43(4), 823-839. <u>https://doi.org/10.1007/bf03395915</u>
- Schweitzer, J. B., & Sulzer-Azaroff, B. (1988). Self-control: Teaching tolerance for delay in impulsive children. *Journal of the Experimental Analysis of Behavior*, 50(2), 173-186. <u>https://doi.org/10.1901/jeab.1988.50-173</u>

- Shallice, T. (1982). Specific impairments of planning. *Philosophical Transactions of the Royal* Society of London B, 298(1089), 199-209. https://doi.org/10.1098/rstb.1982.0082
- Sidman, M. (1971). Reading and auditory-visual equivalences. *Journal of Speech and Hearing Research 14*(1), 5-13. <u>https://doi.org/10.1044/jshr.1401.05</u>
- Sidman, M., Rauzin, R., Lazar, R., Cunningham, S., Tailby, W., & Carrigan, P. (1982). A search for symmetry in the conditional discriminations of rhesus monkeys, baboons, and children. *Journal of the Experimental Analysis of Behavior*, 37(1), 23-44. <u>https://doi.org/10.1901/jeab.1982.37-23</u>
- Skinner, B. F. (1938). Temporal discrimination of the stimulus. In *The behavior of organisms: An experimental analysis* (pp. 263-307). Appleton-Cetury-Crofts.
- Skinner, B. F. (1953). Operant discrimination. In *Science and human behavior* (pp. 107-128). MacMillan.
- Skinner, B. F. (1957a). A functional analysis of verbal behavior. In *Verbal behavior* (pp. 1-12). Appleton-Century-Crofts.
- Skinner, B. F. (1957b). The tact. In Verbal behavior (pp. 81-146). Appleton-Century-Crofts.
- Skinner, B. F. (1969). *Contingencies of reinforcement: A theoretical analysis*. Appleton-Century-Crofts.
- Staddon, J. E., & Higa, J. J. (1999). Time and memory: Towards a pacemaker-free theory of interval timing. *Journal of the Experimental Analysis of Behavior*, 71(2), 215-251. https://doi.org/10.1901/jeab.1999.71-215
- Stamps, L. E. (1977). Temporal conditioning of heart rate responses in newborn infants. Developmental Psychology, 13(6), 624-629. <u>https://doi.org/10.1037/0012-1649.13.6.624</u>
- Steele, D., & Hayes, S. C. (1991). Stimulus equivalence and arbitrarily applicable relational responding. *Journal of the Experimental Analysis of Behavior*, 56(3), 519-555. <u>https://doi.org/10.1901/jeab.1991.56-519</u>
- Stewart, I. (2016). The fruits of a functional approach for psychological science. *International Journal* of Psychology, 51(1), 15-27. https://doi.org/10.1002/ijop.12184
- Stewart, I., Barnes-Holmes, D., & Roche, B. (2004). A functional-analytic model of analogy using the Relational Evaluation Procedure. *The Psychological Record*, 54(4), 531-552. <u>https://doi.org/10.1007/bf03395491</u>
- Stewart, I., Barnes-Holmes, D., Roche, B., & Smeets, P. M. (2002). Stimulus equivalence and nonarbitrary relations. *The Psychological Record*, 52(1), 77-88. <u>https://doi.org/10.1007/bf03395415</u>
- Stewart, I., Barrett, K., McHugh, L., Barnes-Holmes, D., & O'Hora, D. (2013). Multiple contextual control over non-arbitrary relational responding and a preliminary model of pragmatic verbal analysis. *Journal of the Experimental Analysis of Behavior*, 100(2), 174-186. <u>https://doi.org/10.1002/jeab.39</u>

- Stewart, I., & McElwee, J. (2009). Relational responding and conditional discrimination procedures: An apparent inconsistency and clarification. *The Behavior Analyst*, 32(2), 309-317. https://doi.org/10.1007/bf03392194
- Stewart, I., McElwee, J., & Ming, S. (2013). Language generativity, response generalization, and derived relational responding. *The Analysis of Verbal Behavior*, 29(1), 137-155. https://doi.org/10.1007/bf03393131
- Stewart, I., & Roche, B. (2013). Relational frame theory: An overview. In S. Dymond & B. Roche (Eds.), Advances in relational frame theory: Research and application (pp. 51-71). Context Press/New Harbinger Publications.
- Stewart, I., Tarbox, J., Roche, B., & O'Hora, D. (2013). Education, intellectual development, and relational frame theory. In S. Dymond & B. Roche (Eds.), *Advances in relational frame theory: Research and application* (pp. 177-198). Context Press/New Harbinger Publications.
- Stewart, I., Villate, J., & McHugh, L. (2012). Approaches to the self. In *The self and perspective taking: Contributions and applicationsfrom modern behavioral science* (pp. 3-36). New Harbinger Publications, Inc.
- Stromer, R., McComas, J. J., & Rehfeldt, R. A. (2000). Designing interventions that include delayed reinforcement: Implications of recent laboratory research. *Journal of Applied Behavior Analysis*, 33(3), 359-371. <u>https://doi.org/10.1901/jaba.2000.33-359</u>
- Sundberg, M. L. (2008). *The Verbal Behavior Milestones Assessment and Placement Program: The VB-MAPP* (2nd. ed.). AVB Press.
- Tarbox, J., Campbell, V., & Pio, S. (2020). Rule-governed behavior and verbal regulation. In M. Fryling, R. A. Rehfeldt, J. Tarbox, & L. J. Hayes (Eds.), *Applied behavior analysis of language and cognition*. Context Press/New Harbinger Publications.
- Tarbox, J., Szabo, T. G., & Aclan, M. (2020). Acceptance and commitment training within the scope of practice of applied behavior analysis. *Behavior Analysis in Practice*, 15(1), 11-32. <u>https://doi.org/10.1007/s40617-020-00466-3</u>
- Tarbox, J., Zuckerman, C. K., Bishop, M. R., Olive, M. L., & O'Hora, D. P. (2011). Rule-governed behavior: Teaching a preliminary repertoire of rule-following to children with autism. *The Analysis of Verbal Behavior*, 27(1), 125-139. <u>https://doi.org/10.1007/bf03393096</u>
- Tartas, V. (2001). The development of systems of conventional time: A study of the appropriation of temporal locations by four-to-ten-year old children. *European Journal of Psychology of Education*, 16(2), 197-208. <u>https://doi.org/10.1007/bf03173025</u>
- Taylor, I., & O'Reilly, M. F. (2000). Generalization of Supermarket Shopping Skills for Individuals with Mild Intellectual Disabilities Using Stimulus Equivalence Training. *The Psychological Record*, 50(1), 49-62. <u>https://doi.org/10.1007/bf03395342</u>
- Thompson, E. P. (1967). Time, work-discipline, and industrial capitalism. *Past and Present*, 38(1), 56-97. <u>https://doi.org/10.1093/past/38.1.56</u>

- Thompson, R. K. R., & Oden, D. L. (2010). Categorical perception and conceptual judgments by nonhuman primates: The paleological monkey and the analogical ape. *Cognitive Science*, 24(3), 363-396. https://doi.org/10.1207/s15516709cog2403\_2
- Thornton, S. J., & Vukelich, R. (1988). Effects of children's understanding of time concepts on historical understanding. *Theory & Research in Social Education*, 16(1), 69-82. https://doi.org/10.1080/00933104.1988.10505556
- Tillman, K. A., & Barner, D. (2015). Learning the language of time: Children's acquisition of duration words. *Cognitive Psychology*, 78, 57-77. <u>https://doi.org/10.1016/j.cogpsych.2015.03.001</u>
- Tillman, K. A., Fukuda, E., & Barner, D. (2022). Children gradually construct spatial representations of temporal events. *Child Development*, 93(5), 1380-1397. <u>https://doi.org/10.1111/cdev.13780</u>
- Tillman, K. A., Marghetis, T., Barner, D., & Srinivasan, M. (2017). Today is tomorrow's yesterday: Children's acquisition of deictic time words. *Cognitive Psychology*, 92, 87-100. <u>https://doi.org/10.1016/j.cogpsych.2016.10.003</u>
- Tillman, K. A., Tulagan, N., Fukuda, E., & Barner, D. (2018). The mental timeline is gradually constructed in childhood. *Developmental Science*, 21(6), e12679. <u>https://doi.org/10.1111/desc.12679</u>
- Toner, I. J., & Smith, R. A. (1977). Age and overt verbalization in delay-maintenance behavior in children. *Journal of Experimental Child Psychology*, 24(1), 123-128. https://doi.org/10.1016/0022-0965(77)90025-x
- Toussaint, K. A., Scheithauer, M. C., Tiger, J. H., & Saunders, K. J. (2017). Teaching identity matching of braille characters to beginning braille readers. *Journal of Applied Behavior Analysis*, 50(2), 278-289. <u>https://doi.org/10.1002/jaba.382</u>
- Trepal, H., Haberstroh, S., Duffey, T., & Evans, M. (2007). Considerations and strategies for teaching online counseling skills: Establishing relationships in cyberspace. *Counselor Education and Supervision*, 46(4), 266-279. https://doi.org/10.1002/j.1556-6978.2007.tb00031.x
- Trosborg, A. (1982). Children's comprehension of 'before' and 'after' reinvestigated. *Journal of Child Language*, 9(2), 381-402. <u>https://doi.org/10.1017/s0305000900004773</u>
- Truppa, V., Piano Mortari, E., Garofoli, D., Privitera, S., & Visalberghi, E. (2011). Same/different concept learning by capuchin monkeys in matching-to-sample tasks. *PLoS One*, 6(8), e23809. <u>https://doi.org/10.1371/journal.pone.0023809</u>
- Twohig, M. P., Hayes, S. C., & Berens, N. M. (2007). Implications of verbal processes for childhood disorders: Tourette's disorder, attention deficit hyperactivity disorder, and autism. In D. W. Woods & J. W. Kanter (Eds.), *Understanding behavioral disorders: A contemporary behavioral perspective* (pp. 417-443). New Harbingers Publications, Inc.

- Tyrberg, M. J., Parling, T., & Lundgren, T. (2021). Patterns of relational framing in executive function: An investigation of the Wisconsin Card Sorting Test. *The Psychological Record*. https://doi.org/10.1007/s40732-021-00459-w
- Urcuioli, P. J. (2008). Associative symmetry, antisymmetry, and a theory of pigeons' equivalenceclass formation. *Journal of the Experimental Analysis of Behavior*, 90(3), 257-282. <u>https://doi.org/10.1901/jeab.2008.90-257</u>
- vanMarle, K., & Wynn, K. (2006). Six-month-old infants use analog magnitudes to represent duration. *Developmental Science*, 9(5), F41-49. <u>https://doi.org/10.1111/j.1467-</u> <u>7687.2006.00508.x</u>
- Vaughan, M. E. (1985). Repeated acquisition in the analysis of rule-governed behavior. Journal of the Experimental Analysis of Behaviors, 44(2), 175-184. <u>https://doi.org/10.1901/jeab.1985.44-175</u>
- Vismara, L. A., McCormick, C., Young, G. S., Nadhan, A., & Monlux, K. (2013). Preliminary findings of a telehealth approach to parent training in autism. *Journal of Autism and Developmental Disorders*, 43(12), 2953-2969. <u>https://doi.org/10.1007/s10803-013-1841-8</u>
- Vollmer, T. R., Borrero, J. C., Lalli, J. S., & Daniel, D. (1999). Evaluating self-control and impulsivity in children with severe behavior disorders. *Journal of Applied Behavior Analysis*, 32(4), 451-466. <u>https://doi.org/10.1901/jaba.1999.32-451</u>
- Wackermann, J., & Ehm, W. (2006). The dual klepsydra model of internal time representation and time reproduction. *Journal of Theoretical Biology*, 239(4), 482-493. <u>https://doi.org/10.1016/j.jtbi.2005.08.024</u>
- Wainer, A. L., & Ingersoll, B. R. (2011). The use of innovative computer technology for teaching social communication to individuals with autism spectrum disorders. *Research in Autism Spectrum Disorders*, 5(1), 96-107. <u>https://doi.org/10.1016/j.rasd.2010.08.002</u>
- Watson, P. J., & Workman, E. A. (1981). The non-concurrent multiple baseline across-individuals design: An extension of the traditional multiple baseline design. *Journal of Behavior Therapy* and Experimental Psychiatry, 12(3), 257-259. https://doi.org/10.1016/0005-7916(81)90055-0
- Wearden, J. H., & Lejeune, H. (2008). Scalar properties in human timing: Conformity and violations. Quarterly Journal of Experimental Psychology, 61(4), 569-587. https://doi.org/10.1080/17470210701282576
- Weil, T. M., Hayes, S. C., & Capurro, P. (2011). Establishing a deictic relational repertoire in young children. *The Psychological Record*, 61(3), 371-390. <u>https://doi.org/10.1007/bf03395767</u>
- Weiner, H. (1964). Conditioning history and human fixed-interval performance. *Journal of the Experimental Analysis of Behavior*, 7, 383-385. <u>https://doi.org/10.1901/jeab.1964.7-383</u>
- Weist, R. M. (1989). Time concepts in language and thought: Filling the Piagetian void from two to five years. In I. Levin & D. Zakay (Eds.), *Time and human cognition: A life-span perspective* (pp. 63-118). Elsevier. <u>https://doi.org/10.1016/s0166-4115(08)61039-0</u>

- Whelan, R., & Schlund, M. W. (2013). Reframing relational frame theory research: Gaining a new perspective through the application of novel behavioral and neurophysiological methods. In S. Dymond & B. Roche (Eds.), *Advances in relational frame theory: Research and application* (pp. 73-96). New Harbinger Publications, Inc.
- Whipp, R., Adam, B., & Sabelis, I. (2002). *Making time: Time and management in modern organizations*. Oxford University Press.
- Whiting, S. W., & Dixon, M. R. (2015). Delayed reinforcement and self-control: Increasing tolerance for delay with children and adults with autism. In F. D. DiGennaro Reed & D. D. Reed (Eds.), *Autism service delivery: Bridging the gap between science and practice* (pp. 407-435).
  Springer Science + Business Media.
- Wilkening, F., Levin, I., & Druyan, S. (1987). Children's counting strategies for time quantification and integration. *Developmental Psychology*, 23(6), 823-831. <u>https://doi.org/10.1037/0012-1649.23.6.823</u>
- Wittmann, M., & Schmidt, S. (2014). Mindfulness meditation and the experience of time. In S. Schmidt & H. Walach (Eds.), *Meditation – Neuroscientific approaches and philosophical implications* (pp. 199-209). Springer International Publishing. <u>https://doi.org/10.1007/978-3-</u> <u>319-01634-4\_11</u>
- Wymer, S. C., Tarbox, J., Beavers, G. A., & Tullis, C. A. (2016). Teaching children with autism to follow rules specifying a behavior and consequence. *The Analysis of Verbal Behavior*, 32(2), 265-274. <u>https://doi.org/10.1007/s40616-016-0059-1</u>
- Ye, Z., Kutas, M., St George, M., Sereno, M. I., Ling, F., & Munte, T. F. (2012). Rearranging the world: Neural network supporting the processing of temporal connectives. *Neuroimage*, 59(4), 3662-3667. <u>https://doi.org/10.1016/j.neuroimage.2011.11.039</u>
- Zakay, D., & Block, R. A. (1996). The role of attention in time estimation processes. In M. A. Pastor
  & J. Artieda (Eds.), *Time, internal clocks and movement* (pp. 143-164). North-Holland/Elsevier Science Publishers. https://doi.org/10.1016/S0166-4115(96)80057-4
- Zeiler, M. D., & Kelley, C. A. (1969). Fixed-ratio and fixed-interval schedules of cartoon presentation. *Journal of Experimental Child Psychology*, 8(2), 306-313. <u>https://doi.org/10.1016/0022-0965(69)90105-2</u>
- Zelaznik, H. N., Spencer, R. M. C., & Ivry, R. B. (2002). Dissociation of explicit and implicit timing in repetitive tapping and drawing movements. *Journal of Experimental Psychology: Human Perception and Performance*, 28(3), 575-588. <u>https://doi.org/10.1037/0096-1523.28.3.575</u>
- Zentall, T. R., Wasserman, E. A., & Urcuioli, P. J. (2014). Associative concept learning in animals. Journal of the Experimental Analysis of Behavior, 101(1), 130-151. <u>https://doi.org/10.1002/jeab.55</u>
- Zhang, M., & Hudson, J. A. (2018). The development of temporal concepts: Linguistic factors and cognitive processes. *Frontiers in Psychology*, 9, 2451. <u>https://doi.org/10.3389/fpsyg.2018.02451</u>
- Zygmont, D. M., Lazar, R. M., Dube, W. V., & McIlvane, W. J. (1992). Teaching arbitrary matching via sample stimulus-control shaping to young children and mentally retarded individuals: A methodological note. *Journal of the Experimental Analysis of Behavior*, 57(1), 109-117. <u>https://doi.org/10.1901/jeab.1992.57-109</u>

Appendices

# Appendix A: Temporal Relations Assessment Trial List

#### Nonarbitrary Trial Types

Observed Sequence	Question	Correct Response
Bike then fish	was bike before fish?	Yes
Fish then shoe	Was shoe before fish?	No
Shoe then cookie	Was shoe after cookie?	No
Cookie then plane	Was plane after cookie?	Yes
Plane then cat	Was plane before cat?	Yes
Cat then ball	Was hall before cat?	No
Pall then apple	Was ball after apple?	No
		NO
Apple then blke	was blke after apple?	res
Bike then shoe then cat	Was bike before cat?	Yes
Shoe then cat then apple	Was apple before shoe?	No
Cat then apple then plane	Was cat after plane?	No
Apple then plane then ball	Was ball after apple?	Yes
Plane then ball then fish	Was plane before fish?	Ves
Dall than figh than applie	Was plane before hall?	No
Ball then fish then cookie	was cookie before ball?	NO
Fish then cookie then bike	Was fish after bike?	No
Cookie then bike then shoe	Was shoe after cookie?	Yes
Arbitrary Trial Types		
	Mutual Entailment	
Relational Statement	Ouestion	Correct Response
Red before blue	Is red before blue?	Vec
		I CS
Blue before yellow	is yellow before blue?	NO
Yellow before green	Is yellow after green?	No
Green before red	Is red after green?	Yes
Red after blue	Is red before blue?	No
Blue after vellow	Is vellow before blue?	Ves
Vallaw often groom	Is yellow often arean?	Vea
r enow after green	is yenow after green?	i es
Green after red	Is red after green?	No
Blue before red	Is red before blue?	No
Yellow before blue	Is vellow before blue?	Yes
Green before vellow	Is vellow after green?	Ves
Red hafera green	Is red after groon?	No
Red before green		NO
Blue after red	Is red before blue?	Yes
Yellow after blue	Is yellow before blue?	No
Green after yellow	Is yellow after green?	No
Red after green	Is red after green?	Yes
8	Combinatorial Entailment	
Palational Statement	Question	Correct Decrease
Relational Statement	Question	
Red before blue, blue before yellow	Is red before yellow?	Yes
Blue before yellow, yellow before green	Is green before blue?	No
Yellow before green, green before red	Is yellow after red?	No
Green before red, red before blue	Is blue after green?	Ves
Bad after blue, blue after vallery	Is and before yellow?	I es
Red aller blue, blue aller yellow		NO
Blue after yellow, yellow after green	Is green before blue?	Yes
Yellow after green, green after red	Is yellow after red?	Yes
Green after red, red after blue	Is blue after green?	No
Yellow before blue, blue before red	Is red before vellow?	No
Green before vellow, vellow before blue	Is green before blue?	Vec
Bad hafara graan graan hafara vallavy	Is yellow often red?	Vec
Neu belore green, green belore yenow		100
Blue before red, red before green	Is blue after green?	No
Yellow after blue, blue after red	Is red before yellow?	Yes
Green after yellow, yellow after blue	Is green before blue?	No
Red after green green after yellow	Is vellow after red?	No
Blue after red red after green	Is blue after green?	Vec
Blue alter fed, fed alter green		103
	Transformation of Function Trial Types	
Assigned function (for all ToF trials)	Relational statement	Correct response
$\text{Red} \rightarrow \text{clap}$	Red before blue	Claps then raises hand
Blue $\rightarrow$ raise hand	Blue before vellow	Raises hand then touches nose
Vallew , touch nose	Vallow before green	Touchos nose then wewes
$1 \text{ crow} \rightarrow \text{ could noise}$		NU 1 1
Green $\rightarrow$ wave hand	Green before red	waves then claps
	Red after blue	Raises hand then claps
	Blue after yellow	Touches nose then raises hand
	Yellow after green	Waves hand then touches nose
	Green after red	Clans then waves
	Pad bafora blua blua bafora vallow	Clong than raises hand then waves hand
	Neu belore blue, blue belore yellow	Chaps then raises hand then waves hand
	Blue before yellow, yellow before green	Raises hand then touches nose then waves
	Yellow before green, green before red	Touches nose then waves then claps
	Green before red, red before blue	Waves then claps then raises hand
	Red after blue, blue after vellow	Touches nose then raises hand, then claps
	Blue after vellow vellow after green	Wayes then touches nose then raises hand
	Vallow often group on the start	Clang than ways they true list ways
	i chow after green, green after red	Ciaps then waves then touches nose

# **Appendix B: Training Protocol**

### Arbitrary Trials: Mutual Entailment (Yes/No)

**Target behavior:** When given a context of a relation between 2 stimuli, the participant will derive the mutually entailed relation evidenced by a "yes" vocal response when the presented question corresponds with the relation and a "no" vocal response when the presented question does not correspond with the relation.

- Correct responses include: If given the context A is before B, and asked, "Is A before B?" the participant will say, "yes." If given the context B is after A, and asked, "Is B after A?" the participant will say, "no."
- Incorrect responses include: Saying "no" when the correct answer is "yes" or saying "yes" when the correct answer is "no", saying both "yes" and "no" in succession, or providing no response after 10s.

Materials: pictures of mastered stimuli (e.g., colors, shapes, and common items).

Train Block (stimulus set 1: same set from assessment)

- 1. Implementation Steps:
  - a. Present 2 sample stimuli with a contextual cue that indicates the directly trained relation (e.g., red B blue).
  - b. Present one of the four trial types (i.e., "Is A before B?" "Is B before A?" "Is A after B?" "Is B after A?")
  - c. If participant emits the correct response, immediately provide specific verbal praise and provide token.
  - d. If participant emits an incorrect response, provide verbal feedback and instruct the participant to try again.
    - i. If participant responds correctly after receiving verbal feedback, provide specific verbal praise.
      - ii. If error persists, repeat instruction while simultaneously modeling the correct response. Do not provide verbal praise and move on to next trial.
  - e. Repeat sequence with the other 3 types of question formats.
- 2. Reinforcement plan: use tokens, back-up reinforcers, and specific verbal praise
  - a. Provide token for each independent response. Once all tokens have been earned, they can be exchanged for a back-up reinforcer at the following time: \_\_\_\_\_
  - b. Provide specific social praise for each independent response and for correct responses following verbal feedback.
  - c. Session reward can be earned by beating the number of correct responses from the previous session. (Gilroy 2015; Mulhern, 2018)
- 3. Data collection:
  - a. Score (+) for each trial in which the question was answered correctly and independently.
  - b. Score (-) for each trial in which the question required a prompt to be answered correctly.
- 4. Mastery criteria:  $\geq 85\%$  accuracy across 2 consecutive sessions.
  - a. Once the participant reaches mastery criteria with the training set, proceed to next to test for mutual entailment with novel stimulus set

**Test Block** (novel stimulus set)

- 1. Complete a test block with a novel stimulus set once mastery is obtained with the trained stimulus set.
- 2. All test trials are identical to training trials with exception of consequences for correct responses, incorrect responses, and the use of a novel stimulus set
  - a. For correct responses, <u>do not</u> provide verbal raise or tokens
  - b. For incorrect responses, <u>do not</u> provide prompting/error-correction
- 3. Mastery criteria:  $\geq 85\%$  for one session.
  - a. If participant fails to meet mastery criteria on the test set, immediately begin MET with this stimulus set.
  - b. Once training criteria is reached with the current stimulus set, re-test the participant on another novel set.
    i. If the participant fails this test, repeat this training sequence until sequence until the test is passed.
  - c. Once the mastery criteria for the test block is reached, proceed to next phase of intervention.

- 1. All maintenance checks will be conducted with same stimuli that were used in training blocks.
- 2. Stimuli used in test blocks will be used for generalization probes.

# Arbitrary Combinatorial Entailment (Yes/No)

**Target behavior:** When given a context of a relation between 2 stimuli, the participant will derive the mutually entailed relation evidenced by a "yes" vocal response when the presented question corresponds with the relation and a "no" vocal response when the presented question does not correspond with the relation.

- Correct responses include: If given the context A is before B, and asked, "Is A before B?" the participant will say, "yes." If given the context B is after A, and asked, "Is B after A?" the participant will say, "no."
- Incorrect responses include: Saying "no" when the correct answer is "yes" or saying "yes" when the correct answer is "no", saying both "yes" and "no" in succession, or providing no response after 10s.

#### Materials: pictures of colors, shapes, and common items.

#### Train Block (stimulus set 1: same set from assessment)

- 1. Implementation Steps:
  - a. Present 2 sample stimuli with a contextual cue that indicates the directly trained relation (e.g., red B blue).
  - b. Present one of the four trial types (i.e., "Is A before B?" "Is B before A?" "Is A after B?" "Is B after A?")
  - c. If participant emits the correct response, immediately provide specific verbal praise and provide token.
  - d. If participant emits an incorrect response, provide verbal feedback and instruct the participant to try again.
    - i. If participant responds correctly after receiving verbal feedback, provide specific verbal praise.
    - ii. If error persists, repeat instruction while simultaneously modeling the correct response. Do not provide verbal praise and move on to next trial.
  - e. Repeat sequence with the other 3 types of question formats.
  - Reinforcement plan: use tokens, back-up reinforcers, and specific verbal praise
    - a. Provide token for each independent response. Once all tokens have been earned, they can be exchanged for a back-up reinforcer at the following time: \_\_\_\_\_\_
    - b. Provide specific social praise for each independent response and for correct responses following verbal feedback.
    - c. Session reward can be earned by beating the number of correct responses from the previous session. (Gilroy 2015; Mulhern, 2018)
- 3. Data collection:

2.

- a. Score (+) for each trial in which the question was answered correctly and independently.
- b. Score (-) for each trial in which the question required a prompt to be answered correctly.
- 4. Mastery criteria:  $\geq 85\%$  accuracy across 2 consecutive sessions.
  - a. Once the participant reaches mastery criteria with the training set, proceed to next to test for mutual entailment with novel stimulus set

### Test Block (novel stimulus set)

- 1. Complete a test block with a novel stimulus set once mastery is obtained with the trained stimulus set.
- 2. All test trials are identical to training trials with exception of consequences for correct responses, incorrect responses, and the use of a novel stimulus set
  - a. For correct responses, <u>do not</u> provide verbal raise or tokens
  - b. For incorrect responses, do not provide prompting/error-correction
- 3. Mastery criteria:  $\geq 85\%$  for one session.
  - a. If participant fails to meet mastery criteria on the test set, immediately begin MET with this stimulus set.
  - b. Once training criteria is reached with the current stimulus set, re-test the participant on another novel set.
    - i. If the participant fails this test, repeat this training sequence until sequence until the test is passed.
  - c. Once the mastery criteria for the test block is reached, proceed to next phase of intervention.

- 1. All maintenance checks will be conducted with same stimuli that were used in training blocks.
- 2. Stimuli used in test blocks will be used for generalization probes.

## Transformation of Function—Mutual Entailment

**Target behavior:** The participant will do the sequence of motor actions in the correct order as indicated by the relational statement.

- Correct responses include: given an image of "red *before* blue" and trained that red = clap, and blue = raise hand; the participant first claps, and then raises their hand. Given an image of red *after* blue and trained red = clap and blue = raise hand; the participant first raises their hand, and then claps.
- Incorrect responses include: acting out the target actions in the incorrect order, acting out the actions simultaneously, or acting out other actions.

Materials: pictures of colors, shapes, and common items; visual of color-motor action correspondence.

**Pre-training** (complete at the beginning of each training session):

- 1. Present stimulus from the current set.
- 2. Vocally state and model the motor action that corresponds with the stimulus.
- 3. Allow student opportunity to imitate the action and repeat with all stimuli in the set.
- 4. Re-present stimuli in random order to test for accuracy.
  - a. If participant accurately demonstrates all actions without prompting, proceed to training phase.
  - b. If less than 100%, repeat steps 1-4 until accurate with all stimuli in the set.

#### **Training:**

- 1. Implementation Steps:
  - a. Present 2 sample stimuli with a contextual cue that indicates the directly trained relation (e.g., red B blue).
  - b. Also present the color-motor actions visual for learner.
  - c. Next, read the relation out-loud (e.g. "e.g., red is before blue.")
  - d. Deliver instruction for learner to perform the actions in the correct order (e.g., "Do these actions in order).
  - e. If participant emits the correct response, immediately provide specific verbal praise and provide token.
  - f. If participant emits an incorrect response, provide verbal feedback and instruct the participant to try again.
    - i. If participant responds correctly after receiving verbal feedback, provide specific verbal praise.
      - ii. If error persists, repeat instruction while simultaneously modeling the correct response. Do not provide verbal praise and move on to next trial.
    - Repeat sequence with the remaining trials in the trial block.
- 2. Reinforcement plan: use tokens, back-up reinforcers, and specific verbal praise
  - a. Provide token for each independent response. Once all tokens have been earned, they can be exchanged for a back-up reinforcer at the following time: \_\_\_\_\_
  - b. Provide specific social praise for each independent response and for correct responses following verbal feedback.
  - c. Session reward can be earned by beating the number of correct responses from the previous session. (Gilroy 2015; Mulhern, 2018)
- 3. Data collection:

b.

g.

- a. Score (+) for each trial in which both actions were acted out in the correct order.
- b. Score (-) for each trial in which an incorrect response was made (see above).
- Mastery criteria:  $\geq 85\%$  accuracy across 2 consecutive sessions.
  - a. Once the participant reaches mastery criteria with the training set, proceed to next to test for transformation of function via mutual entailment with novel stimulus set.

### **Testing:**

4.

- 1. Complete a test block with a novel stimulus set once mastery is obtained with the trained stimulus set.
- 2. All test trials are identical to training trials with exception of consequences for correct responses, incorrect responses, and the use of a novel stimulus set.
  - a. For correct responses, <u>do not</u> provide verbal raise or tokens
  - b. For incorrect responses, do not provide prompting/error-correction
- 3. Mastery criteria:  $\geq 85\%$  for one session.
  - a. If participant fails to meet mastery criteria on the test set, immediately begin MET with this stimulus set.
    - Once training criteria is reached with the current stimulus set, re-test the participant on another novel set.
      - i. If the participant fails this test, repeat this training sequence until sequence until the test is passed.
  - c. Once the mastery criteria for the test block is reached, proceed to next phase of intervention.

- 1. All maintenance checks will be conducted with same stimuli that were used in training blocks.
- 2. Stimuli used in test blocks will be used for generalization probes.

## Transformation of Function—Combinatorial Entailment

**Target behavior:** The participant will do the sequence of motor actions in the correct order as indicated by the relational statement.

- Correct responses include: given an image of "red *before* blue *before* yellow" and trained that red = clap, blue = raise hand, and yellow = tap table; the participant first claps, then raises their hand, then taps the table. Given an image of red *after* blue *after* yellow and trained red = clap, blue = raise hand, and yellow = tap table; the participant first taps the table, then raises their hand, and then claps.
- Incorrect responses include acting out the target actions in the incorrect order, acting out the actions simultaneously, or acting out other actions.

Materials: pictures of colors, shapes, and common items; visual of color-motor action correspondence.

Pre-training (complete at the beginning of each training session):

- 1. Present stimulus from the current set.
- 2. Vocally state and model the motor action that corresponds with the stimulus.
- 3. Allow student opportunity to imitate the action and repeat with all stimuli in the set.
- 4. Re-present stimuli in random order to test for accuracy.
  - a. If participant accurately demonstrates all actions without prompting, proceed to training phase.
  - b. If less than 100%, repeat steps 1-4 until accurate with all stimuli in the set.

#### **Training:**

- 1. Implementation Steps:
  - a. Present 2 sample stimuli with a contextual cue that indicates the directly trained relation (e.g., red B blue).
  - b. Also present the color-motor actions visual for learner.
  - c. Next, read the relation out-loud (e.g. "e.g., red is before blue.")
  - d. Deliver instruction for learner to perform the actions in the correct order (e.g., "Do these actions in order).
  - e. If participant emits the correct response, immediately provide specific verbal praise and provide token.
  - f. If participant emits an incorrect response, provide verbal feedback and instruct the participant to try again.
    - i. If participant responds correctly after receiving verbal feedback, provide specific verbal praise.
      - ii. If error persists, repeat instruction while simultaneously modeling the correct response. Do not provide verbal praise and move on to next trial.
  - g. Repeat sequence with the remaining trials in the trial block.
- 2. Reinforcement plan: use tokens, back-up reinforcers, and specific verbal praise
  - a. Provide token for each independent response. Once all tokens have been earned, they can be exchanged for a back-up reinforcer at the following time: \_\_\_\_\_
  - b. Provide specific social praise for each independent response and for correct responses following verbal feedback.
  - c. Session reward can be earned by beating the number of correct responses from the previous session. (Gilroy 2015; Mulhern, 2018)
- 3. Data collection:
  - a. Score (+) for each trial in which all three actions were acted out in the correct order.
  - b. Score (-) for each trial in which an incorrect response was made (see above).
- 4. Mastery criteria:  $\geq 85\%$  accuracy across 2 consecutive sessions.
  - a. Once the participant reaches mastery criteria with the training set, proceed to next to test for transformation of function via mutual entailment with novel stimulus set.

### **Testing:**

- 1. Complete a test block with a novel stimulus set once mastery is obtained with the trained stimulus set.
- 2. All test trials are identical to training trials with exception of consequences for correct responses, incorrect responses, and the use of a novel stimulus set.
  - a. For correct responses, <u>do not</u> provide verbal raise or tokens
  - b. For incorrect responses, do not provide prompting/error-correction
- 3. Mastery criteria:  $\geq 85\%$  for one session.
  - a. If participant fails to meet mastery criteria on the test set, immediately begin MET with this stimulus set.
  - b. Once training criteria is reached with the current stimulus set, re-test the participant on another novel set.
    - i. If the participant fails this test, repeat this training sequence until sequence until the test is passed.
  - c. Once the mastery criteria for the test block is reached, proceed to next phase of intervention.

- 1. All maintenance checks will be conducted with same stimuli that were used in training blocks.
- 2. Stimuli used in test blocks will be used for generalization probes.

## **Appendix C: Example Data Sheets**

Learner ID:

## **Arbitrary Mutual Entailment**

Stimulus Notes Stimulus **Stimulus Set** Phase A2 blue A1 red Set #: Stimulus B2 yellow B1 blue Baseline 1 red Train 2 A3 A4 blue yellow green Test yellow 3 B3 green B4 red Maint. 4 green Gen.

1. Present trials in random order.

2. For each trial, present the instruction listed in the corresponding row. Follow protocol according to current phase of data collection.

3. On the first trial of each trial type, record the participant's response.

4. For first trial probes on trial types answered incorrectly, run additional training trials.

#	Trial Type	Stimulus Presentation	Instruction	Answer	1st Trial Probe	Additional Training Trials
1	A1 b B1	red b blue	red before blue?	Yes		
2	A1 a B1	red a blue	red before blue?	No		
3	A2 b B2	blue b yellow	yellow before blue?	No		
4	A2 a B2	blue a yellow	yellow before blue?	Yes		
5	A3 b B3	yellow b green	yellow after green?	No		
6	A3 a B3	yellow a green	yellow after green?	Yes		
7	A4 b B4	green b red	red after green?	Yes		
8	A4 a B4	green a red	red after green?	No		
	First trial probes correct/8			correct/8	/8	
	% first trial probes correct				%	

#### Arbitrary Transformation of Function - Mutual Entailment

Appendix C

Date: \_\_\_\_\_ Initials: \_\_\_\_\_

Ы		Stimulus Set			Stimulus	Action		Stimulus	Action	Notes
Phase	Set #:		A1	red	clap	A2	blue	raise hand		
Baseline		Stimulus	Motor Action	B1	blue	raise hand	B2	yellow	tap table	
Train	1	red	clap							
Test	2	blue	raise hand	A3	yellow	tap table	A4	green	wave	
Maint.	3	yellow	tap table	B3	green	wave	B4	red	clap	
Gen.	4	green	wave							

1. Present trials in random order.

2. For each trial, present the instruction, "Do these in order." Follow protocol according to current phase of data collection.

3. On the first trial of each trial type, record (+) if independent and (—) if prompt required.

4. For first trial probes on trial	types answered incorrectly, run additional training trials.
	· / F · · · · · · · · · · · · · · · · ·

#	Trial Type	Stimulus Presentation	Answer	1st Trial Probe	Additional Training Trials
1	A1 b B1	red b blue	clap—raise hand		
2	A1 a B1	red a blue	raise hand—clap		
3	A2 b B2	blue b yellow	raise hand—tap table		
4	A2 a B2	blue a yellow	tap table—raise hand		
5	A3 b B3	yellow b green	tap table—wave		
6	A3 a B3	yellow a green	wave—tap table		
7	A4 b B4	green b red	wave—clap		
8	A4 b B4	green a red	clap—wave		
		F	irst trial probes correct/8	/8	

% first trial probes correct %

# Learner ID: \_\_\_\_\_

#### **Arbitrary Combinatorial Entailment**

Appendix C

Date: \_\_\_\_\_ Initials: \_\_\_\_\_

Phase	Stimulus Set	Stimulus	Stimulus	Notes
1 hase	Set #:	A1 red	A2 blue	
Baseline	Stimulus	B1 blue	B2 yellow	
Train	1 red	C1 yellow	C2 green	
Test	2 blue	A3 yellow	A4 green	
Maint.	3 yellow	B3 green	B4 red	
Gen.	4 green	C3 red	C4 blue	

1. Present trials in random order.

2. For each trial, present the instruction kisted in the corresponding row. Follow protocol according to current phase of data collection.

3. On the first trial of each trial type, record the participant's response.

4. For first trial probes on trial types answered incorrectly, run additional training trials.

#	Trial Type	Stimulus Presentation	Instruction	Answer	1st Trial Probe	Additional Training Trials
1	A1 b B1 b C1	red b blue b yellow	red before yellow?	Yes		
2	A1 a B1 a C1	red a blue a yellow	red before yellow?	No		
3	A2 b B2 b C2	blue b yellow b green	green before blue?	No		
4	A2 a B2 a C2	blue a yellow a green	green before blue?	Yes		
5	A3 b B3 b C3	yellow b green b red	yellow after red?	No		
6	A3 a B3 a C3	yellow a green a red	yellow after red?	Yes		
7	A4 b B4 b C4	green b red b blue	blue after green?	Yes		
8	A4 a B4 a C4	green a red a blue	blue after green?	No		
			$\mathbf{T}^{1}$ ( ( 1 1	1/0	/0	

First trial probes correct/8 /8

% first trial probes correct %

Appendix C

Learner ID:

#### Arbitrary Transformation of Function - Combinatorial Entailment

Date: \_\_\_\_\_ Initials: \_\_\_\_\_

Dham		Stimulus Set			Stimulus	Action		Stimulus	Action	Notes
Phase		Set #:		A1	red	clap	A2	blue	raise hand	
Baseline		Stimulus	Motor Action	B1	blue	raise hand	B2	yellow	tap table	
Train	1	red	clap	C1	yellow	tap table	C2	green	wave	
Test	2	blue	raise hand	A3	yellow	tap table	A4	green	wave	
Maint.	3	yellow	tap table	В3	green	wave	B4	red	clap	
Gen.	4	green	wave	C3	red	clap	C4	blue	raise hand	

1. Present trials in random order.

2. For each trial, present the instruction, "Do these in order." Follow protocol according to current phase of data collection.

3. On the first trial of each trial type, record (+) if independent and (--) if prompt required.

A For first trial m	robes on trial types s	newered incorrectly run	additional training trials
T. I'OI III St ullai pi	toocs on that types a	mowercu meenceny, run	auditional training trials.
1	- 1		0

#	Trial Type	Stimulus Presentation	Answer	1st Trial Probe	Additional Training Trials
1	A1 b B1 b C1	red b blue b yellow	clap—raise hand—tap table		
2	A1 a B1 a C1	red a blue a yellow	tap table—raise hand—clap		
3	A2 b B2 b C2	blue b yellow b green	raise hand-tap table-wave		
4	A2 a B2 a C2	blue a yellow a green	wave-tap table-raise hand		
5	A3 b B3 b C3	yellow b green b red	tap table—wave—clap		
6	A3 a B3 a C3	yellow a green a red	clap—wave—tap table		
7	A4 b B4 b C4	green b red b blue	wave—clap—raise hand		
8	A4 a B4 a C4	green a red a blue	raise hand—clap—wave		
			First trial probes correct/8	/8	
			% first trial probes correct	%	

% first trial probes correct